## Some lectures in basic analysis

Stephen Semmes Rice University

# Preface

These informal notes are largely concerned with metric spaces, sequences and series, and continuous mappings. In particular, compactness, completeness, connectedness, and related notions will play central roles.

There are many textbooks in analysis, in which the reader can find much more information. These notes are intended to complement such textbooks, some of which can be found in the bibliography.

The abstract notion of a metric space is one of the main topics discussed in these notes, and is defined in Section 1.1. Basic examples include the real line, the complex plane, and n-dimensional Euclidean spaces, with their standard Euclidean metrics.

Some important properties of the real numbers that will be needed here are discussed in Sections 1.2 and 1.3. Some readers may be familiar with properties like these already, and otherwise more information can be found in many texts, including ones in the bibliography.

Many readers probably have at least some familiarity with sequences and series of real numbers. Here we shall consider convergence of sequences in arbitrary metric spaces, as well as some topics related to infinite series of real and complex numbers. As usual, we shall focus on some of the more theoretical aspects, rather than some of the typical examples that many readers may have seen previously.

One of the motivations for the study of continuity and convergence is the theory behind calculus. Some readers may have some familiarity with this already, and otherwise some of the results discussed here can be used to make precise well-known arguments related to differentiation and integration. Some related matters will be discussed beginning in Chapter 5.

Some additional topics related to those discussed here will be mentioned in Section 4.15. Some related aspects of history may be found in [7, 8, 12, 13, 27, 32, 33, 34, 35, 59, 60, 63, 76, 83, 84, 85, 87, 89, 90, 91, 92, 93, 95, 108, 110, 125, 144, 188], for instance. Some related songs may be found in [148, 149, 150]. Some remarks concerning the clarity of explanations in mathematics may be found in [94]. The reader may also be interested in [25].

A question came up about the axiom of choice, and we shall not deal with this much here. The axiom of choice is implicitly used in some proofs, particularly with a sequence of choices. Sometimes the axiom of choice is implicitly used, and a slightly different argument would not require it. A very nice reference for some topics like these is [122]. Some more information may be found in [104, 165], for instance.

Countability and uncountability of sets are discussed at the beginning of Chapter 2, without getting into cardinalities of sets too much. This is discussed further in [122] too. The reader may also be interested in Gödel's paper [77] on the continuum hypothesis. This is mentioned in [122] as well.

I would like to dedicate these notes to the memory of my undergraduate advisor, Dr. Richard M. Summerville.

# Contents

1	Me	tric spaces 1			
	1.1	The definition, and some examples			
		1.1.1 The standard Euclidean metric on the real line 1			
		1.1.2 The standard Euclidean metric on $\mathbf{R}^n$			
		1.1.3 Some more metrics on $\mathbb{R}^n$			
		1.1.4 The discrete metric			
		1.1.5 Subsets of metric spaces			
	1.2	Supremum and infimum			
		1.2.1 The least upper bound property of $\mathbf{R}$			
		1.2.2 Lower bounds and the infimum			
	1.3	Additional properties of $\mathbf{R}$			
		1.3.1 Existence of $n$ th roots $\ldots \ldots \ldots$			
	1.4	Open balls and open sets 5			
		1.4.1 Open balls are open sets			
		1.4.2 Some more open sets $\ldots \ldots \ldots$			
		1.4.3 Open intervals in $\mathbf{R}$			
	1.5	Limit points and closed sets			
		1.5.1 A property of limit points			
		1.5.2 Closed sets in metric spaces			
		1.5.3 Closed intervals in $\mathbf{R}$			
		1.5.4 Complements of subsets of $M$			
	1.6	Unions, intersections, and closures			
		1.6.1 Finite intersections of open sets			
		1.6.2 Finite unions of closed sets			
		1.6.3 Arbitrary unions of open sets			
		1.6.4 Arbitrary intersections of closed sets			
		1.6.5 Closures of subsets of $M$			
	1.7	Compactness			
		1.7.1 An example of a compact set in the real line 10			
		1.7.2 An example of a noncompact subset of $\mathbf{R}$			
		1.7.3 The limit point property $\ldots \ldots \ldots$			
		1.7.4 Compact sets have the limit point property 11			
	1.8	Bounded sets			
		1.8.1 Compact sets are bounded			

		1.8.2 Sets with the limit point property are bounded	13
	1.9	Compactness and closed sets	13
		1.9.1 Compact sets are closed	14
		1.9.2 Closed sets contained in compact sets are compact	14
	1.10	Relatively open sets	15
		1.10.1 A characterization of relatively open subsets of $Y$	15
		1.10.2 Compactness and relatively open sets	16
	1.11	Sequences of closed intervals	17
		1.11.1 Some more properties of the intersection of the $I_i$ 's	18
	1.12	Compactness of closed intervals	19
		1.12.1 A sequence of closed intervals	19
		1.12.2 Using the nonemptiness of the intersection of the $I_i$ 's	20
	1.13	Cells in $\mathbf{R}^n$	20
		1.13.1 Nested sequences of cells	21
		1.13.2 Some unions of smaller cells	21
	1.14	Compactness of cells	22
		1.14.1 Using the nonemptiness of the intersection of the $C_l{\rm 's}$	22
		1.14.2 Closed and bounded subsets of $\mathbf{R}^n$ are compact	23
	1.15	Totally bounded sets	23
		1.15.1 Compact sets are totally bounded	23
		1.15.2 The limit point property implies total boundedness	23
·	• • • • • • • •		25
2		ric spaces, continued	25
			() [
	2.1	Countable sets	25 25
	2.1	2.1.1 More on countability and sequences	25
	2.1	<ul><li>2.1.1 More on countability and sequences</li></ul>	$\frac{25}{26}$
	2.1	<ul> <li>2.1.1 More on countability and sequences</li></ul>	25 26 26
		<ul> <li>2.1.1 More on countability and sequences</li></ul>	25 26 26 27
	2.1 2.2	2.1.1More on countability and sequences	25 26 26 27 27
	2.2	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable setsUncountable sets	25 26 26 27 27 28
		2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable setsUncountable sets	25 26 26 27 27 28 28
	2.2	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.1.4Cartesian products of countable sets2.1.4Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.2.1Dense subsets of metric spaces	25 26 27 27 28 28 28 28
	2.2	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.1.4Cartesian products of countable sets2.1.4Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.2.1Dense subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples	25 26 27 27 28 28 28 28 28
	2.2	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.1.4Cartesian products of countable sets2.1.4Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.2.1Dense subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability	25 26 27 27 28 28 28 28 28 28 28
	2.2 2.3	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability	25 26 27 27 28 28 28 28 28 28 28 28 28 29
	2.2	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.1.4Cartesian products of countable setsUncountable sets	25 26 27 27 28 28 28 28 28 28 28 29 29
	2.2 2.3	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.1.4Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separabilityBases	25 26 27 27 28 28 28 28 28 28 28 29 29 30
	2.2 2.3	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases	25 26 27 27 28 28 28 28 28 28 28 29 29 30 30
	2.2 2.3 2.4	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separability	25 26 27 27 28 28 28 28 28 28 29 29 30 30 30 31
	2.2 2.3	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separability2.4.3Necessity of separability	25 26 26 27 27 28 28 28 28 28 28 29 29 30 30 31 31
	<ul><li>2.2</li><li>2.3</li><li>2.4</li><li>2.5</li></ul>	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separability2.4.3Necessity of separability2.5.1The proof of Lindelöf's theorem	25 26 27 27 28 28 28 28 28 28 29 29 30 30 31 31 31
	2.2 2.3 2.4	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separabilityLindelöf's theorem2.5.1The proof of Lindelöf's theoremCountable open coverings	25 26 27 27 28 28 28 28 28 29 29 30 30 30 31 31 31 31 32
	<ul><li>2.2</li><li>2.3</li><li>2.4</li><li>2.5</li></ul>	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separabilityLindelöf's theorem2.5.1The proof of Lindelöf's theorem2.6.1Checking that $L$ has infinitely many elements	25 26 27 27 28 28 28 28 28 28 29 29 30 30 31 31 31 32 33
	<ul><li>2.2</li><li>2.3</li><li>2.4</li><li>2.5</li></ul>	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.3.4Some examples of bases2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separability2.5.1The proof of Lindelöf's theorem2.5.1The proof of Lindelöf's theorem2.6.1Checking that $L$ has infinitely many elements2.6.2Using the limit point property	25 26 27 27 28 28 28 28 28 28 29 29 30 30 31 31 31 31 32 33 33
	<ul> <li>2.2</li> <li>2.3</li> <li>2.4</li> <li>2.5</li> <li>2.6</li> </ul>	2.1.1More on countability and sequences2.1.2A lemma about unions of sequences of finite sets2.1.3Sequences of countable sets2.1.4Cartesian products of countable sets2.2.1Perfect subsets of metric spaces2.2.1Perfect subsets of metric spaces2.3.1Dense subsets of metric spaces2.3.2The definition of separability, and some examples2.3.3A characterization of separability2.3.4Total boundedness implies separability2.3.4Total boundedness implies separability2.4.1Some examples of bases2.4.2Separability and countable bases2.4.3Necessity of separabilityLindelöf's theorem2.5.1The proof of Lindelöf's theorem2.6.1Checking that $L$ has infinitely many elements	25 26 27 27 28 28 28 28 28 28 29 29 30 30 31 31 31 32 33

	2.7.2	Subsets of $M$	34
2.8	Conne	ectedness	34
	2.8.1	Separated subsets of $M$	34
	2.8.2	Connected sets	35
	2.8.3	Connected subsets of $\mathbf{R}$	35
	2.8.4	Suprema, infima, and closures	35
2.9	The C	Cantor set	36
	2.9.1	Some elements of $E$	36
	2.9.2	Every element of $E$ is a limit point of $A$	37
Sea	1100000	s and series	38
3.1		lex numbers	38
0.1	3.1.1	Complex conjugates and absolute values	38
	3.1.1 3.1.2	Addition and multiplication	$\frac{38}{38}$
	3.1.2 3.1.3	The triangle inequality	$\frac{38}{39}$
3.2		·	$\frac{39}{39}$
3.2	3.2.1	University of the limit	39 39
	3.2.1 3.2.2	Uniqueness of the limit	$\frac{39}{40}$
		Bounded sequences	40 40
	3.2.3	Convergent sequences are bounded	
<u></u>	3.2.4	Monotonic sequences of real numbers	40
3.3		and products	41
	3.3.1	Sums of convergent sequences	41
	3.3.2	A lemma about products	42
	3.3.3	A lemma about constant multiples	42
	3.3.4	Products of convergent sequences	43
<b>.</b>	3.3.5	Sequences of reciprocals	43
3.4		quences and sequential compactness	44
	3.4.1	Sequential compactness	44
	3.4.2	The limit point property implies sequential compactness .	44
	3.4.3	Sequential compactness implies the limit point property .	45
3.5		ny sequences and completeness	45
	3.5.1	Cauchy sequences	45
	3.5.2	Convergent sequences are Cauchy sequences	46
	3.5.3	Cauchy sequences are bounded	46
	3.5.4	Cauchy sequences with convergent subsequences	46
	3.5.5	Complete metric spaces	47
3.6	Some	particular sequences	47
	3.6.1	$\{1/j^p\}_{j=1}^{\infty}$	47
	3.6.2	$\{a^j\}_{j=1}^{\infty}$	48
	3.6.3	$\{j^{\alpha}a^{j}\}_{j=1}^{\infty}$	48
	3.6.4	$\{p^{1/j}\}_{j=1}^{\infty}$	48
	3.6.5	$\{j^{1/j}\}_{j=1}^{\infty}$	49
3.7	Infinit	ge series	49
	3.7.1	Sums and constant multiples	50
	3.7.2	The Cauchy criterion	50

vi

3

	3.7.3	Absolute convergence	51
	3.7.4	Absolute convergence implies convergence	51
	3.7.5	Infinite series with nonnegative terms	51
	3.7.6	The comparison test	51
	3.7.7	Geometric series	52
3.8	Power	series	52
	3.8.1	An absolute convergence property	52
	3.8.2	Another absolute convergence property	53
	3.8.3	The radius of convergence	53
	3.8.4	A characterization of the radius of convergence $\ldots$ .	53
3.9	Cauch	y's condensation test	54
	3.9.1	The "if" part	54
	3.9.2	The "only if" part	54
	3.9.3	Some examples of series of this type	55
3.10	Anoth	er criterion for convergence	55
	3.10.1	A related series	56
	3.10.2	Leibniz' alternating series test	56
	3.10.3	Some more complicated examples	56
3.11	Extend	ded real numbers	57
	3.11.1	Suprema and infima of sets of extended real numbers	57
	3.11.2	Sums and products of extended real numbers	57
	3.11.3	Sequences in <b>R</b> tending to $+\infty$ or $-\infty$	58
	3.11.4	Some properties of sequences of real numbers	58
	3.11.5	Subsequences of sequences in $\mathbf{R}$	59
3.12	Limits	superior and inferior $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	60
	3.12.1	A characterization of the upper limit of a sequence	60
	3.12.2	A characterization of the lower limit of a sequence $\ . \ . \ .$	61
	3.12.3	Equality of the upper and lower limits	61
	3.12.4	Some variants of these definitions and results $\ldots \ldots$	61
		Upper limits of sums of sequences of real numbers	62
3.13	Applic	eations for infinite series	62
	3.13.1	The proof of the root test, and some additional remarks $% \left( {{{\bf{n}}_{\rm{c}}}} \right)$ .	63
	3.13.2	The root test and the radius of convergence of a power	
		series	63
		The ratio test	64
3.14		angements of infinite series	65
		Rearrangements with nonnegative terms	65
		Rearrangements of absolutely convergent series	65
		Convergence without absolute convergence	66
3.15		y products of infinite series	66
		Cauchy products of series with nonnegative terms	67
	3.15.2	Cauchy products and absolute convergence $\ldots$	67

n	777	TE	NTO	$\Gamma C$
υU	JIN	IĽ.	1 N 1	D.

4	Con	tinuou	s mappings	69
	4.1 Continuity at a point			69
		4.1.1	Characterizing continuity in terms of convergent sequences	69
		4.1.2	Continuous complex-valued functions	70
		4.1.3	Some examples of continuous functions	70
	4.2		ositions and inverse images	71
		4.2.1	Continuity and open sets	71
		4.2.2	Continuity and closed sets	71
		4.2.3	Compositions of continuous mappings	72
		4.2.4	Another proof using convergent sequences	72
		4.2.5	A proof using open sets	72
	4.3		s of compact sets	73
	4.0	4.3.1	Continuity and sequential compactness	73
		4.3.2	The extreme value theorem	73 74
	4.4		m continuity	74 74
	4.4	4.4.1		$74 \\ 75$
		4.4.1	Uniform continuity and compactness	75 75
	4.5			75
	4.0		s of connected sets	
	1.0	4.5.1	The intermedate value theorem	76
	4.6		m convergence	77
		4.6.1	Pointwise and uniform convergence	77
		4.6.2	Uniform convergence and continuity	77
		4.6.3	Weierstrass' criterion for uniform convergence	78
	4.7		uity of power series	78
		4.7.1	Continuity on a closed disk	79
		4.7.2	Continuity on an open disk, or the whole plane	79
	4.8	The su	premum metric	80
		4.8.1	Bounded functions	80
		4.8.2	Defining the supremum metric	80
		4.8.3	Convergence with respect to the supremum metric	80
	4.9	Compl	eteness of $C_b(M,N)$	81
		4.9.1	Pointwise convergence of Cauchy sequences	81
		4.9.2	Checking uniform convergence	82
		4.9.3	Boundedness of the limit	82
	4.10	The lir	nit of a function $\ldots$	82
		4.10.1	Characterizing limits in terms of convergent sequences	83
		4.10.2	Limits of complex-valued functions	83
		4.10.3	Limits and continuity	83
	4.11	One-si	ded limits	83
		4.11.1	Monotonically increasing functions	84
			Continuity of monotone functions	85
	4.12		ibing jump discontinuities	85
			Some infinite series of functions	86
	4.13		imits of limits	87
			When $\{q_j\}_{j=1}^{\infty}$ converges in $N$	87
	4.14		onnectedness	87

viii

		4.14.1 Continuous paths and connectedness	87
		4.14.2 Path connected sets	88
		4.14.3 Path connected sets are connected	
		4.14.4 Convex subsets of $\mathbf{R}^n$	
		4.14.5 More on path connectedness	
	4.15	Integral metrics and other topics	89
		4.15.1 The $L^1$ metric on $C([a, b])$	89
		4.15.2 Comparison with the supremum metric	
		4.15.3 Isometric mappings and completions	89
		4.15.4 The Baire category theorem	
		4.15.5 The contraction mapping theorem	
		4.15.6 Brouwer's fixed-point theorem	90
		4.15.7 Weierstrass' approximation theorem	90
5	Diff	erentiating functions of one variable	91
	5.1	Functions on subsets of $\mathbf{R}$	91
		5.1.1 The derivative of a real-valued function	91
		5.1.2 Differentiability implies continuity	91
		5.1.3 Differentiability of sums and products	
		5.1.4 Differentiability of reciprocals	92
	5.2	The chain rule	93
		5.2.1 The statement and some indications of the proof $\ldots$ .	93
		5.2.2 Differentiating inverse functions	
		5.2.3 Functions on open intervals	
		5.2.4 More on the differentiability of the inverse function	
	5.3	One-sided derivatives	
	5.4	More on one-sided derivatives	
		5.4.1 The generalized mean value theorem	
	5.5	The Darboux property	
		5.5.1 Another property of the derivative	
	5.6	Some examples	
	5.7	Lipschitz mappings	
		5.7.1 Real-valued functions on subsets of the real line	99
6		grating functions of one variable	101
	6.1	Suprema and infima of functions	
		6.1.1 Multiplication by a real number	
		6.1.2 Sums of bounded functions	
	6.2	More on suprema and infima of functions	
		6.2.1 Products of nonnegative functions	
		6.2.2 Some monotonicity properties	
		6.2.3 Some more monotonicity properties	
	6.3	Darboux sums	
		6.3.1 Upper and lower Darboux sums	
	. ·	6.3.2 Riemann sums	
	6.4	Darboux–Stieltjes sums	104

		6.4.1 Riemann–Stieltjes sums	. 105
		6.4.2 A basic sum using $\alpha$	. 105
	6.5	More on Darboux sums	. 105
		6.5.1 Some basic estimates for upper and lower sums	. 105
		6.5.2 Refinements of partitions	. 106
	6.6	Upper and lower integrals	
		6.6.1 Some basic estimates for upper and lower integrals $\ldots$	. 107
		6.6.2 Upper and lower Riemann–Stieltjes integrals of $f$	. 107
	6.7	Common refinements	
		6.7.1 Using common refinements	. 108
		6.7.2 Some remarks about upper and lower sums	
		6.7.3 A property of refinements	
	6.8	Riemann–Stieltjes integrals	
		6.8.1 Riemann–Stieltjes integrability	
		6.8.2 A characterization of integrability	
	6.9	Continuity and integrability	
		6.9.1 Some more criteria for integrability	
		6.9.2 A related example	
	6.10	More on Riemann–Stieltjes integrals	
		6.10.1 A remark about approximating integrals by sums	
	6.11	Multiplying functions by constants	
		6.11.1 Multiplication by $-1$	
		6.11.2 Multiplying $\alpha$ by $r > 0$	
	6.12	Adding two functions	
		6.12.1 Sums of integrable functions	
		6.12.2 Some related results	
	6.13	Another monotonically increasing function	
		6.13.1 More on these upper and lower integrals	
		6.13.2 Integrability with respect to $\alpha + \beta$	. 117
7	Mar	a an differentiation and intermation	118
1	7.1	re on differentiation and integration Some inequalities with integrals	
	1.1	7.1.1 A simple estimate for the absolute value of the integral	
	7.2	Subintervals of $[a, b]$	
	1.2	7.2.1 Starting with an arbitrary partition of $[a, b]$	
		7.2.1 Starting with an arbitrary partition of $[a, b]$	
		7.2.2 Integrability on $[a, b]$	
	7.3	Derivatives of integrals $\dots \dots \dots$	
	1.0	7.3.1 Using the Riemann integral	
	7.4	Integrals of derivatives	
	1.1	7.4.1 The case where $f$ is continuous on $[a, b]$	
		7.4.2 When the derivative is Riemann integrable on $[a, b]$	
	7.5	More on Riemann–Stieltjes integrability	
		7.5.1 Integrability of $\phi \circ f$	
		7.5.2 Integrability of products $\dots \dots \dots$	
		7.5.3 Integrability of $ f $	
		1.0.0 mooproprint of [1]	. 120

х

7.6	Differe	ntiability of $\alpha$	126
7.7	Chang	ing variables	126
	7.7.1	Compositions with $\phi$	127
	7.7.2	Partitions of $[a_0, b_0]$	127
	7.7.3	Partitions of [a, b]	127
	7.7.4	Integrability on $[a_0, b_0]$	128
	7.7.5	Some additional remarks	128
7.8	Some s	sums of step functions	128
	7.8.1	Riemann–Stieltjes integrals with respect to $\alpha$	129
	7.8.2	Some related sums	129
7.9	Limits	of integrals	130
Bibliog	graphy		132
Index			145

## xi

xii

## Chapter 1

# Metric spaces

## 1.1 The definition, and some examples

**Definition 1.1.1** A metric space is a set M together with a nonnegative realvalued function d(x, y) defined for  $x, y \in M$  that satisfies the following three conditions. First, for each  $x, y \in M$ , we have that

(1.1.2) 
$$d(x,y) = 0 \quad if and only if \quad x = y.$$

Second, d(x, y) should be symmetric in x and y, so that

(1.1.3) 
$$d(x,y) = d(y,x) \text{ for every } x, y \in M.$$

The third condition is that

$$(1.1.4) d(x,z) \le d(x,y) + d(y,z) for every x, y, z \in M,$$

which is called the triangle inequality. Under these conditions,  $d(\cdot, \cdot)$  is said to define a metric on M.

## 1.1.1 The standard Euclidean metric on the real line

As a basic example, the real line **R** is a metric space with respect to the standard *Euclidean metric*. If x is a real number, then the *absolute value* |x| of x is defined as usual by

(1.1.5) 
$$|x| = x \quad \text{when } x \ge 0$$
$$= -x \quad \text{when } x \le 0.$$

It is well known and not difficult to see that

$$(1.1.6) |x+y| \le |x|+|y|$$

for every  $x, y \in \mathbf{R}$ . More precisely, we have equality in (1.1.6) when x and y have the same sign, and otherwise there is some cancellation in the left side.

The standard Euclidean metric on  $\mathbf{R}$  is defined by

(1.1.7) 
$$d(x,y) = |x-y|.$$

One can check directly that this satisfies the requirements of a metric on  $\mathbf{R}$ . In particular, (1.1.4) follows from (1.1.6) in this situation.

## 1.1.2 The standard Euclidean metric on $\mathbb{R}^n$

Let n be a positive integer, and let  $\mathbf{R}^n$  be the usual space of n-tuples of real numbers. Thus an element of  $\mathbf{R}^n$  can be given as

(1.1.8) 
$$x = (x_1, \dots, x_n),$$

where  $x_1, \ldots, x_n \in \mathbf{R}$ . The standard Euclidean metric on  $\mathbf{R}^n$  is defined by

(1.1.9) 
$$d(x,y) = \left(\sum_{j=1}^{n} (x_j - y_j)^2\right)^{1/2},$$

using the nonnegative square root on the right side. It is easy to see that this satisfies the first two requirements (1.1.2), (1.1.3) of a metric. It is well known that (1.1.9) satisfies the triangle inequality (1.1.4), but this is more complicated when  $n \geq 2$ , and we shall not give a proof here.

## 1.1.3 Some more metrics on $\mathbb{R}^n$

One can check that

(1.1.10) 
$$d'(x,y) = \sum_{j=1}^{n} |x_j - y_j|$$

and

(1.1.11) 
$$d''(x,y) = \max_{1 \le j \le n} |x_j - y_j|$$

also define metrics on  $\mathbb{R}^n$ . In particular, the triangle inequality for (1.1.10) and (1.1.11) can be obtained from (1.1.6). If n = 1, then (1.1.9), (1.1.10), and (1.1.11) are all the same as (1.1.7).

## 1.1.4 The discrete metric

Let M be any set. If  $x, y \in M$ , then put

(1.1.12) 
$$d(x,y) = 0 \quad \text{when } x = y$$
$$= 1 \quad \text{when } x \neq y.$$

One can check that this defines a metric on M. This is called the *discrete metric* on M.

#### 1.1.5 Subsets of metric spaces

Now let (M, d(x, y)) be an arbitrary metric space, and let E be a subset of M. It is easy to see that the restriction of d(x, y) to  $x, y \in E$  defines a metric on E. Thus E may be considered as a metric space as well, using this metric.

## **1.2** Supremum and infimum

**Definition 1.2.1** Let A be a subset of the real line. A real number b is said to be an upper bound for A if for every  $a \in A$  we have that  $a \leq b$ . A real number c is said to be the least upper bound or supremum of A in **R** if it satisfies the following two conditions. First,

(1.2.2) c is an upper bound for A.

Second,

(1.2.3) if  $b \in \mathbf{R}$  is an upper bound for A, then  $c \leq b$ .

It is easy to see that such a c is unique when it exists. More precisely, if  $c', c'' \in \mathbf{R}$  both satisfy these two conditions, then one can check that  $c' \leq c''$  and  $c'' \leq c'$ . This implies that c' = c'', as desired.

## 1.2.1 The least upper bound property of R

An important feature of the real numbers is the *least upper bound property*. This says that if A is a nonempty subset of  $\mathbf{R}$  with an upper bound in  $\mathbf{R}$ , then A has a least upper bound in  $\mathbf{R}$ . As before, we shall also refer to this as the supremum of A in  $\mathbf{R}$ , which may be denoted sup A.

The least upper bound property is an important difference between the real line and the set  $\mathbf{Q}$  of rational numbers. If A is a nonempty subset of  $\mathbf{Q}$  with an upper bound in  $\mathbf{Q}$ , then the supremum of A in  $\mathbf{R}$  may not be an element of  $\mathbf{Q}$ .

There are well-known ways in which  $\mathbf{R}$  can be obtained from  $\mathbf{Q}$ , but we shall not pursue this here.

## 1.2.2 Lower bounds and the infimum

**Definition 1.2.4** Let A be a subset of **R** again. A real number y is said to be a lower bound for A if for every  $x \in A$  we have that  $y \leq x$ . A real number z is said to be the greatest lower bound or infimum of A in **R** if it satisfies the following two conditions. First,

Second,

(1.2.6) if  $y \in \mathbf{R}$  is a lower bound for A, then  $y \leq z$ .

One can check that such a z is unique when it exists, as before. In this case, z may be denoted inf A.

If A is not the empty set  $\emptyset$  and A has a lower bound in **R**, then A has a greatest lower bound in **R**. To see this, put

$$(1.2.7) B = \{ y \in \mathbf{R} : y \text{ is a lower bound for } A \}.$$

Thus  $B \neq \emptyset$ , by hypothesis, and every element of A is an upper bound for B. The least upper bound property for **R** implies that B has a least upper bound in **R**. One can verify that sup B also satisfies the requirements of the greatest lower bound for A, as desired.

Alternatively, let -A be the set of real numbers of the form -a, with  $a \in A$ . Of course,  $-A \neq \emptyset$ , because  $A \neq \emptyset$ . If  $y \in \mathbf{R}$  is a lower bound for A, then -y is an upper bound for -A. Hence -A has a supremum in  $\mathbf{R}$ . One can check that  $-\sup(-A)$  satisfies the requirements of the infimum of A.

## 1.3 Additional properties of R

The archimedean property of the real numbers is the following:

(1.3.1) if x and y are positive real numbers, then there is a positive integer n such that y < n x.

Here is another important property of the real line:

(1.3.2) if  $a, b \in \mathbf{R}$  and a < b, then there is an  $r \in \mathbf{Q}$  such that a < r < b.

It is not too difficult to see that each of these two properties implies that other. There is a well-known argument by which the archimedean property can be obtained using the least upper bound property of  $\mathbf{R}$ . Alternatively, (1.3.2) can be obtained from standard constructions of  $\mathbf{R}$  from  $\mathbf{Q}$ .

#### **1.3.1** Existence of *n*th roots

Let x be a positive real number, and let n be a positive integer. It is well known that there is a unique positive real number y such that

 $(1.3.3) y^n = x.$ 

The uniqueness of y can be verified directly, but the existence of y is more complicated. Let A be the set of nonnegative real numbers t such that

$$(1.3.4) t^n < x.$$

Note that  $0 \in A$ , so that  $A \neq \emptyset$ . If  $x \leq 1$ , then it is easy to see that 1 is an upper bound for A. Otherwise, if x > 1, then one can check that x is an upper bound for A. In both cases, A has an upper bound in **R**, which implies that A

has a supremum in **R**. It is well known that  $y = \sup A$  satisfies (1.3.3), but we shall not prove this here.

If m is another positive integer, then one can define  $x^{m/n}$  to be the positive nth root of  $x^m > 0$ . Using this, one can define  $x^r$  when  $r \in \mathbf{Q}$  and r > 0. More precisely, one can verify that this does not depend on the particular representation for r as a ratio of positive integers. Note that  $x^r \ge 1$  when  $x \ge 1$ , and  $x^r \le 1$  when  $x \le 1$ .

If t is a positive real number, then one can define  $x^t$  as a positive real number too. More precisely, if  $x \ge 1$ , then one can take  $x^t$  to be the supremum of the set of positive real numbers of the form  $x^r$ , where  $r \in \mathbf{Q}$  and  $0 < r \le t$ . Otherwise, if  $0 < x \le 1$ , then one can reduce to the previous case, by considering 1/x.

## 1.4 Open balls and open sets

Let (M, d(x, y)) be a metric space.

**Definition 1.4.1** If x is an element of M and r is a positive real number, then the open ball in M centered at x with radius r is defined to be the set of  $y \in M$ such that d(x, y) < r, i.e.,

(1.4.2) 
$$B(x,r) = \{ y \in M : d(x,y) < r \}.$$

A subset U of M is said to be an open set in M with respect to  $d(\cdot, \cdot)$  if for every  $x \in U$  there is a positive real number r such that

$$(1.4.3) B(x,r) \subseteq U.$$

Note that M is automatically an open subset of itself. The empty set is an open set in M as well.

#### 1.4.1 Open balls are open sets

**Proposition 1.4.4** If  $z \in M$  and t is a positive real number, then B(z,t) is an open set in M.

To see this, let an element x of B(z,t) be given. Thus d(x,z) < t, so that r = t - d(x,z) > 0. It suffices to show that

$$(1.4.5) B(x,r) \subseteq B(z,t).$$

To do this, let y be an arbitrary point in B(x, r), so that d(x, y) < r. Using the triangle inequality, we get that

(1.4.6) 
$$d(y,z) \le d(y,x) + d(x,z) < r + d(x,z) = t.$$

This means that d(y, z) < t, and hence  $y \in B(z, t)$ . This shows that (1.4.5) holds, as desired.

## 1.4.2 Some more open sets

**Proposition 1.4.7** If  $z \in M$  and t is a nonnegative real number, then

(1.4.8) 
$$V(z,t) = \{ w \in M : d(z,w) > t \}$$

is an open set in M.

To prove this, let  $x \in V(z,t)$  be given, so that r = d(x,z) - t > 0. Under these conditions, one can use the triangle inequality to show that

$$(1.4.9) B(x,r) \subseteq V(z,t).$$

The details are left as an exercise.

## 1.4.3 Open intervals in R

If a and b are real numbers with a < b, then

$$(1.4.10) (a,b) = \{x \in \mathbf{R} : a < x < b\}$$

is the *open interval* in **R** from *a* to *b*. It is easy to see that this is an open set in **R**, with respect to the standard Euclidean metric. This is the same as the open ball in **R** centered at the midpoint (a + b)/2 of the interval with radius (b - a)/2.

## **1.5** Limit points and closed sets

Let (M, d(x, y)) be a metric space, and let E be a subset of M.

**Definition 1.5.1** A point  $p \in M$  is said to be a limit point of E in M if for every positive real number r there is a point  $q \in E$  such that d(p,q) < r and  $p \neq q$ .

Note that a limit point of E in M may or may not be an element of E.

## 1.5.1 A property of limit points

**Proposition 1.5.2** If  $p \in M$  is a limit point of E, then for each r > 0 there are infinitely many elements of E in B(p, r).

Let r > 0 be given, and suppose for the sake of a contradiction that there are only finitely many elements of E in B(p, r). Let  $q_1, \ldots, q_n$  be a list of the elements of E in B(p, r) that are not equal to p. Put

(1.5.3) 
$$t = \min_{1 \le j \le n} d(p, q_j),$$

which is a positive real number. By construction, there are no elements of E in B(p,t), except perhaps p, contradicting the hypothesis that p be a limit point of E in M.

#### 1.5.2 Closed sets in metric spaces

**Definition 1.5.4** A subset E of M is said to be a closed set with respect to  $d(\cdot, \cdot)$  if for every  $p \in M$  such that p is a limit point of E, we have that  $p \in E$ .

If  $E \subseteq M$  has only finitely many elements, then the previous proposition implies that E has no limit points in M, and hence that E is a closed set in M. Observe that M is automatically a closed set in itself.

## 1.5.3 Closed intervals in R

If  $a, b \in \mathbf{R}$  and  $a \leq b$ , then the closed interval in  $\mathbf{R}$  from a to b is defined by

$$(1.5.5) [a,b] = \{x \in \mathbf{R} : a \le x \le b\}.$$

It is easy to see that this is a closed set in  $\mathbf{R}$ , with respect to the standard Euclidean metric.

## **1.5.4** Complements of subsets of M

If A and B are sets, then let  $A \setminus B$  be the set of points in A and not in B, i.e.,

$$(1.5.6) A \setminus B = \{ x \in A : x \notin B \}.$$

If M is a set and E is a subset of M, then  $M \setminus E$  is called the *complement* of E in M. In this case, we have that

(1.5.7) 
$$M \setminus (M \setminus E) = E.$$

Let (M, d(x, y)) be a metric space again.

**Proposition 1.5.8** A subset E of M is a closed set if and only if  $M \setminus E$  is an open set in M.

Suppose first that E is a closed set, and let us check that  $M \setminus E$  is an open set. Let  $p \in M \setminus E$  be given. Note that p is not a limit point of E, because  $p \notin E$  and E is a closed set. This implies that there is an r > 0 such that B(p,r) does not contain any elements of E, except perhaps p. It follows that B(p,r) does not contain any elements of E, because  $p \notin E$ . Equivalently, this means that

$$.5.9) B(p,r) \subseteq M \setminus E,$$

as desired.

(1)

Conversely, suppose that  $M \setminus E$  is an open set in M, and let us show that E is a closed set. If  $p \in M \setminus E$ , then there is an r > 0 such that (1.5.9) holds, because  $M \setminus E$  is an open set. This is the same as saying that

$$(1.5.10) B(p,r) \cap E = \emptyset$$

In particular, this means that p is not a limit point of E in M. It follows that every limit point of E in M is contained in E, as desired.

Using the proposition, we get that a subset U of M is an open set if and only if  $M \setminus U$  is a closed set. More precisely, this follows from the proposition with  $E = M \setminus U$ .

## 1.6 Unions, intersections, and closures

Let (M, d(x, y)) be a metric space.

#### **1.6.1** Finite intersections of open sets

**Proposition 1.6.1** If  $U_1, \ldots, U_n$  are finitely many open subsets of M, then their intersection  $\bigcap_{j=1}^n U_j$  is an open set in M as well.

Let x be an arbitrary element of  $\bigcap_{j=1}^{n} U_j$ , so that  $x \in U_j$  for each  $j = 1, \ldots, n$ . Thus, for each  $j = 1, \ldots, n$ , there is a positive real number  $r_j$  such that

$$(1.6.2) B(x,r_j) \subseteq U_j,$$

because  $U_j$  is an open set in M. Put

(1.6.3) 
$$r = \min(r_1, \dots, r_n)$$

and note that r > 0. Clearly

$$(1.6.4) B(x,r) \subseteq B(x,r_j) \subseteq U_j$$

for each  $j = 1, \ldots, n$ , so that

(1.6.5) 
$$B(x,r) \subseteq \bigcap_{j=1}^{n} U_j,$$

as desired.

## 1.6.2 Finite unions of closed sets

**Corollary 1.6.6** If  $E_1, \ldots, E_n$  are finitely many closed subsets of M, then their union  $\bigcup_{j=1}^{n} E_j$  is a closed set too.

It suffices to show that the complement  $M \setminus \left(\bigcup_{j=1}^{n} E_{j}\right)$  of the union is an open set in M, by Proposition 1.5.8. It is well known and not difficult to check that

(1.6.7) 
$$M \setminus \left(\bigcup_{j=1}^{n} E_{j}\right) = \bigcap_{j=1}^{n} \left(M \setminus E_{j}\right).$$

We also have that  $M \setminus E_j$  is an open set in M for each j = 1, ..., n, because  $E_j$  is a closed set by hypothesis. Proposition 1.6.1 implies that the right side of (1.6.7) is an open set in M, as desired.

#### **1.6.3** Arbitrary unions of open sets

**Proposition 1.6.8** Let A be a nonempty set, and suppose that  $U_{\alpha}$  is an open subset of M for each  $\alpha \in A$ . Under these conditions,  $\bigcup_{\alpha \in A} U_{\alpha}$  is an open set in M.

Let  $x \in \bigcup_{\alpha \in A} U_{\alpha}$  be given, so that there is an index  $\alpha_0 \in A$  such that  $x \in U_{\alpha_0}$ . Because  $U_{\alpha_0}$  is an open set in M, by hypothesis, there is a positive real number r such that

$$(1.6.9) B(x,r) \subseteq U_{\alpha_0}$$

This implies that

(1.6.10) 
$$B(x,r) \subseteq \bigcup_{\alpha \in A} U_{\alpha}$$

as desired.

#### **1.6.4** Arbitrary intersections of closed sets

**Corollary 1.6.11** If A is a nonempty set, and  $E_{\alpha} \subseteq M$  is a closed set for each  $\alpha \in A$ , then  $\bigcap_{\alpha \in A} E_{\alpha}$  is a closed set in M.

One can verify that

(1.6.12) 
$$M \setminus \left(\bigcap_{\alpha \in A} E_{\alpha}\right) = \bigcup_{\alpha \in A} (M \setminus E_{\alpha}),$$

as before. Note that  $M \setminus E_{\alpha}$  is an open set for every  $\alpha \in A$ , because  $E_{\alpha}$  is a closed set, by hypothesis. Thus the right side of (1.6.12) is an open set too, by Proposition 1.6.8. This implies that  $\bigcap_{\alpha \in A} E_{\alpha}$  is a closed set, by Proposition 1.5.8.

Alternatively, suppose that  $p \in M$  is a limit point of  $\bigcap_{\alpha \in A} E_{\alpha}$ . If  $\beta$  is any element of A, then one can check that p is a limit point of  $E_{\beta}$ , because  $\bigcap_{\alpha \in A} E_{\alpha} \subseteq E_{\beta}$ . This implies that  $p \in E_{\beta}$ , because  $E_{\beta}$  is a closed set in M. It follows that  $p \in \bigcap_{\alpha \in A} E_{\alpha}$ , because the previous statement holds for every  $\beta \in A$ . This shows that  $\bigcap_{\alpha \in A} E_{\alpha}$  contains all of its limit points in M, as desired.

## **1.6.5** Closures of subsets of M

**Definition 1.6.13** Let E be a subset of M. The closure of E in M is the set  $\overline{E}$  consisting of all  $p \in M$  such that  $p \in E$ , p is a limit point of E, or both.

If  $E \subseteq M$  is a closed set, then  $\overline{E} = E$ , because E contains its limit points in M. Conversely, if  $\overline{E} = E$ , then E contains all of its limit points in M, and hence E is a closed set.

**Proposition 1.6.14** If E is any subset of M, then  $\overline{E}$  is a closed set in M.

It is enough to show that  $M \setminus \overline{E}$  is an open set in M, as in Proposition 1.5.8. Let  $p \in M \setminus \overline{E}$  be given, so that  $p \notin E$  and p is not a limit point of E. The latter implies that there is a positive real number r such that B(p, r) does not contain any elements of E, other than p. This means that B(p, r) does not contain any elements of E, because  $p \notin E$ . Equivalently,  $B(p, r) \subseteq M \setminus E$ . Using this, one can verify that

$$(1.6.15) B(p,r) \subseteq M \setminus E,$$

as desired. Alternatively, one can show that every limit point of  $\overline{E}$  in M is a limit point of E, and hence an element of  $\overline{E}$ .

## 1.7 Compactness

Let (M, d(x, y)) be a metric space, and let K be a subset of M.

**Definition 1.7.1** A family  $\{U_{\alpha}\}_{\alpha \in A}$  of open subsets of M is said to be an open covering of K in M if

(1.7.2) 
$$K \subseteq \bigcup_{\alpha \in A} U_{\alpha}.$$

If every open covering of K in M can be reduced to a finite subcovering of K, then K is said to be compact in M. More precisely, K is compact in M if for every open covering  $\{U_{\alpha}\}_{\alpha \in A}$  of K in M, there are finitely many indices  $\alpha_1, \ldots, \alpha_n \in A$  such that

(1.7.3) 
$$K \subseteq \bigcup_{j=1}^{n} U_{\alpha_j}.$$

It is easy to see that finite subsets of M are compact.

## 1.7.1 An example of a compact set in the real line

Let us consider a couple of examples in the real line, equipped with the standard Euclidean metric. Consider

(1.7.4) 
$$K = \{0\} \cup \{1/j : j \in \mathbf{Z}_+\}$$

where  $\mathbf{Z}_+$  denotes the set of positive integers. We would like to show that this set is compact in  $\mathbf{R}$ . To do this, let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of K in  $\mathbf{R}$ . In particular, there is an  $\alpha_0 \in A$  such that  $0 \in U_{\alpha_0}$ , because  $0 \in K$ . This implies that there is a positive real number r such that

$$(1.7.5) \qquad (-r,r) \subseteq U_{\alpha_0},$$

because  $U_{\alpha_0}$  is an open set in **R**. Let *n* be a positive integer such that 1/r < n, as in the archimedean property for **R**. Equivalently, 1/n < r. If n = 1, then it

#### 1.7. COMPACTNESS

follows that  $K \subseteq U_{\alpha_0}$ . Otherwise, if n > 1, then for each  $j = 1, \ldots, n-1$ , let  $\alpha_j$  be an element of A such that  $1/j \in U_{\alpha_j}$ . Under these conditions, we get that

(1.7.6) 
$$K \subseteq \bigcup_{j=0}^{n-1} U_{\alpha_j}$$

as desired.

## 1.7.2 An example of a noncompact subset of R

Next, let us show that the open unit interval (0, 1) is not compact in **R**. Thus we would like to find an open covering of (0, 1) in **R** for which there is no finite subcovering. One way to do this is to use the family of open intervals of the form (1/j, 1), where j is a positive integer greater than or equal to two. Each of these open intervals is an open set in **R** with respect to the standard Euclidean metric, and it is easy to see that

(1.7.7) 
$$\bigcup_{j=2}^{\infty} (1/j, 1) = (0, 1)$$

However, one can check that (0, 1) is not contained in the union of finitely many of these intervals. More precisely,

$$(1.7.8) (1/j,1) \subseteq (1/(j+1),1)$$

for each  $j \ge 2$ . This implies that the union of finitely many intervals of the form (1/j, 1) is equal to a single such interval.

#### 1.7.3 The limit point property

Let (M, d(x, y)) be any metric space again.

**Definition 1.7.9** A subset E of M is said to have the limit point property if for every subset L of E such that L has infinitely many elements, there is a point  $p \in E$  that is a limit point of L in M.

If  $E \subseteq M$  has only finitely many elements, then E automatically has the limit point property.

#### 1.7.4 Compact sets have the limit point property

**Proposition 1.7.10** If  $K \subseteq M$  is compact, then K has the limit point property.

Let L be an infinite subset of K, and suppose for the sake of a contradiction that L has no limit point in K. Let  $p \in K$  be given, so that p is not a limit point of L. This means that there is a positive real number r(p) such that B(p, r(p))does not contain any elements of L, except perhaps p itself. The family of open balls B(p, r(p)) of this type, with  $p \in K$ , is an open covering of K, because  $p \in B(p, r(p))$  for each  $p \in K$ . If K is compact, then there are finitely many elements  $p_1, \ldots, p_n$  of K such that

(1.7.11) 
$$K \subseteq \bigcup_{j=1}^{n} B(p_j, r(p_j)).$$

In particular, it follows that

(1.7.12) 
$$L \subseteq \bigcup_{j=1}^{n} B(p_j, r(p_j)),$$

because  $L \subseteq K$ , by hypothesis. This implies that L has at most n elements, because  $p_i$  is the only element of  $B(p_i, r(p_i))$  that can be in L. This contradicts the hypothesis that L have infinitely many elements, as desired.

#### Bounded sets 1.8

Let (M, d(x, y)) be a metric space.

**Definition 1.8.1** A subset E of M is said to be bounded in M with respect to  $d(\cdot, \cdot)$  if there is a point  $p \in M$  and a positive real number r such that

$$(1.8.2) E \subseteq B(p,r)$$

The empty set is considered to be a bounded set in M, even when  $M = \emptyset$ .

#### Compact sets are bounded 1.8.1

**Proposition 1.8.3** If  $K \subseteq M$  is compact and  $p \in M$ , then there is a positive real number r such that  $K \subseteq B(p, r).$ (1.8.4)

Let  $p \in M$  be given, and observe that

(1.8.5) 
$$\bigcup_{j=1}^{\infty} B(p,j) = M.$$

It follows that the family of open balls  $B(p, j), j \in \mathbb{Z}_+$ , is an open covering of K in M. If K is compact, then there are finitely many positive integers  $j_1, \ldots, j_n$ such that

(1.8.6) 
$$K \subseteq \bigcup_{l=1}^{n} B(p, j_l).$$

This implies that (1.8.4) holds with

(1.8.7) 
$$r = \max(j_1, \dots, j_n),$$

as desired.

#### 1.8.2 Sets with the limit point property are bounded

**Proposition 1.8.8** If  $E \subseteq M$  has the limit point property and  $p \in M$ , then there is a positive real number r such that  $E \subseteq B(p, r)$ .

Suppose for the sake of a contradiction that for each positive integer j,

$$(1.8.9) E \not\subseteq B(p,j).$$

Thus, for each positive integer j, there is an element of E whose distance to p is at least j. Let us now choose, for each positive integer j, such a point  $x_j \in E$  with

$$(1.8.10) d(p, x_j) \ge j$$

Let

(1.8.11) 
$$L = \{x_j : j \in \mathbf{Z}_+\}$$

be the set of points in E that have been chosen in this way.

Suppose for the sake of a contradiction that L has only finitely many elements. This implies that there is an  $x \in E$  such that  $x = x_j$  for infinitely many j. In this case, we would have that

$$(1.8.12) d(p,x) \ge j$$

for infinitely many positive integers j, by (1.8.10). However, the archimedean property for **R** implies that there is a positive integer n such that  $d(p, x) \leq n$ , so that (1.8.12) holds for only finitely many  $j \in \mathbf{Z}_+$ .

Thus L has infinitely many elements. If E has the limit point property, then there is a point  $q \in E$  such that q is a limit point of L in M. In particular, this implies that B(q, 1) contains infinitely many elements of L, as in Proposition 1.5.2. It follows that

(1.8.13) 
$$d(q, x_i) < 1$$

for infinitely many  $j \in \mathbf{Z}_+$ , by the definition (1.8.11) of *L*. Using this and the triangle inequality, we get that

$$(1.8.14) d(p, x_j) \le d(p, q) + d(q, x_j) < d(p, q) + 1$$

for infinitely many  $j \in \mathbf{Z}_+$ . Combining this with (1.8.10), we obtain that

$$(1.8.15) j < d(p,q) + 1$$

for infinitely many  $j \in \mathbb{Z}_+$ . This is a contradiction, because (1.8.15) holds for only finitely many positive integers j, as before.

## **1.9** Compactness and closed sets

Let (M, d(x, y)) be a metric space.

#### 1.9.1 Compact sets are closed

**Proposition 1.9.1** If  $K \subseteq M$  is compact, then K is a closed set in M.

Let  $p \in M \setminus K$  be given, and let us show that p is not a limit point of K. If j is a positive integer, then V(p, 1/j) consists of the elements of M whose distance to p is larger than 1/j, as in (1.4.8). Remember that this is an open set in M, as in Proposition 1.4.7. It is easy to see that

(1.9.2) 
$$\bigcup_{j=1}^{\infty} V(p, 1/j) = M \setminus \{p\}.$$

This implies that the family of sets V(p, 1/j),  $j \in \mathbb{Z}_+$ , is an open covering of K in M, because  $p \in M \setminus K$ , by hypothesis.

If K is compact, then there are finitely many positive integers  $j_1, \ldots, j_n$  such that

(1.9.3) 
$$K \subseteq \bigcup_{l=1}^{n} V(p, 1/j_l).$$

This means that

$$(1.9.4) K \subseteq V(p,r),$$

where r is the positive real number defined by

(1.9.5) 
$$1/r = \max(j_1, \dots, j_n)$$

In particular, there are no elements of K in B(p, r), so that p is not a limit point of K, as desired.

#### **1.9.2** Closed sets contained in compact sets are compact

**Proposition 1.9.6** Suppose that  $K \subseteq M$  is compact, and that  $E \subseteq M$  is a closed set. If  $E \subseteq K$ , then E is compact in M.

To see this, let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of E in M. Note that  $M \setminus E$  is an open set, because E is a closed set, as in Proposition 1.5.8. We also have that

(1.9.7) 
$$K \subseteq M = E \cup (M \setminus E) \subseteq \left(\bigcup_{\alpha \in A} U_{\alpha}\right) \cup (M \setminus E).$$

This implies that the collection of  $U_{\alpha}$ ,  $\alpha \in A$ , together with  $M \setminus E$ , forms an open covering of K in M.

If K is compact, then there are finitely many indices  $\alpha_1, \ldots, \alpha_n \in A$  such that

(1.9.8) 
$$K \subseteq \left(\bigcup_{j=1}^{n} U_{\alpha_j}\right) \cup (M \setminus E).$$

In particular, this means that

(1.9.9) 
$$E \subseteq \left(\bigcup_{j=1}^{n} U_{\alpha_j}\right) \cup (M \setminus E),$$

because  $E \subseteq K$ , by hypothesis. Of course, this implies that

(1.9.10) 
$$E \subseteq \bigcup_{j=1}^{n} U_{\alpha_j},$$

as desired.

## 1.10 Relatively open sets

Let (M, d(x, y)) be a metric space, and let Y be a subset of M. Remember that the restriction of d(x, y) to  $x, y \in Y$  defines a metric on Y, so that Y may be considered as a metric space as well.

**Definition 1.10.1** A subset E of Y is said to be relatively open in Y if E is an open set in Y with respect to the restriction of d(x, y) to  $x, y \in Y$ .

Suppose for instance that  $M = \mathbb{R}^2$ , equipped with the standard Euclidean metric, and Y is a line in  $\mathbb{R}^2$ . One can have open line segments in Y that are relatively open in Y, and not open as subsets of M.

Let (M, d(x, y)) be any metric space again, and let Y be a subset of M. If  $x \in Y$  and r is a positive real number, then let  $B_Y(x, r)$  be the corresponding open ball in Y, with respect to the restriction of  $d(\cdot, \cdot)$  to Y. Thus

(1.10.2) 
$$B_Y(x,r) = \{y \in Y : d(x,y) < r\} = B(x,r) \cap Y,$$

where we continue to use B(x, r) for the open ball in M centered at x with radius r. By definition, a subset E of Y is relatively open in Y if and only if for every  $x \in E$  there is an r > 0 such that

$$(1.10.3) B_Y(x,r) \subseteq E.$$

## 1.10.1 A characterization of relatively open subsets of Y

**Proposition 1.10.4** A subset E of Y is relatively open in Y if and only if there is an open subset U of M such that

$$(1.10.5) E = U \cap Y.$$

To prove the "if" part, suppose that  $U \subseteq M$  is an open set, and let us check that  $E = U \cap Y$  is relatively open in Y. Let  $x \in E$  be given, so that  $x \in U$ 

in particular. Because U is an open set in M, there is an r > 0 such that  $B(x,r) \subseteq U$ . This implies that

$$(1.10.6) B_Y(x,r) = B(x,r) \cap Y \subseteq U \cap Y = E_y$$

as desired.

Conversely, suppose that E is a relatively open subset of Y. Thus, for each  $x \in E$ , there is a positive real number r(x) such that

$$(1.10.7) B_Y(x, r(x)) \subseteq E.$$

Put

(1.10.8) 
$$U = \bigcup_{x \in E} B(x, r(x)).$$

This may be interpreted as being the empty set when  $E = \emptyset$ . Observe that U is an open set in M, because open balls are open sets, and a union of open sets is an open set too, as in Propositions 1.4.4 and 1.6.8.

We would like to verify that (1.10.5) holds in this situation. It is easy to see that  $E \subseteq U$ , because  $x \in B(x, r(x))$  for every  $x \in E$ . This implies that

$$(1.10.9) E \subseteq U \cap Y,$$

because  $E \subseteq Y$ , by hypothesis. Using the definition (1.10.8) of U, we get that

(1.10.10) 
$$U \cap Y = \left(\bigcup_{x \in E} B(x, r(x))\right) \cap Y = \bigcup_{x \in E} (B(x, r(x)) \cap Y).$$

It follows that

(

1.10.11) 
$$U \cap Y = \bigcup_{x \in E} B_Y(x, r(x)) \subseteq E,$$

as desired, using (1.10.2) in the first step, and (1.10.7) in the second step.

## 1.10.2 Compactness and relatively open sets

**Proposition 1.10.12** Let K be a subset of Y. Under these conditions, K is compact as a subset of M if and only K is compact as a subset of Y, with respect to the restriction of  $d(\cdot, \cdot)$  to Y.

Suppose first that K is compact as a subset of Y, and let us check that K is compact as a subset of M. Let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of K in M. If  $\alpha \in A$ , then  $U_{\alpha}$  is an open subset of M, and hence  $U_{\alpha} \cap Y$  is relatively open in Y, by the previous proposition. We also have that

(1.10.13) 
$$K \subseteq \left(\bigcup_{\alpha \in A} U_{\alpha}\right) \cap Y = \bigcup_{\alpha \in A} (U_{\alpha} \cap Y),$$

#### 1.11. SEQUENCES OF CLOSED INTERVALS

because K is contained in each of  $\bigcup_{\alpha \in A} U_{\alpha}$  and Y, by hypothesis. This shows that  $\{U_{\alpha} \cap Y\}_{\alpha \in A}$  is an open covering of K in Y. If K is compact as a subset of Y, then there are finitely many indices  $\alpha_1, \ldots, \alpha_n \in A$  such that

(1.10.14) 
$$K \subseteq \bigcup_{j=1}^{n} (U_{\alpha_j} \cap Y).$$

This implies that  $K \subseteq \bigcup_{j=1}^{n} U_{\alpha_j}$ , as desired.

Conversely, suppose that K is compact as a subset of M, and let us show that K is compact as a subset of Y. To do this, let  $\{E_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of K in Y. This means that  $E_{\alpha}$  is a relatively open subset of Y for each  $\alpha \in A$ , and that

(1.10.15) 
$$K \subseteq \bigcup_{\alpha \in A} E_{\alpha}$$

If  $\alpha \in A$ , then there is an open subset  $U_{\alpha}$  of M such that

$$(1.10.16) E_{\alpha} = U_{\alpha} \cap Y$$

by Proposition 1.10.4. This leads to an open covering  $\{U_{\alpha}\}_{\alpha\in A}$  of K in M. If K is compact in M, then there are finitely many indices  $\alpha_1, \ldots, \alpha_n \in A$  such that  $K \subseteq \bigcup_{j=1}^{n} U_{\alpha_j}$ . It follows that

(1.10.17) 
$$K \subseteq \left(\bigcup_{j=1}^{n} U_{\alpha_j}\right) \cap Y = \bigcup_{j=1}^{n} (U_{\alpha_j} \cap Y) = \bigcup_{j=1}^{n} E_{\alpha_j},$$

as desired, because  $K \subseteq Y$ .

#### Sequences of closed intervals 1.11

**Proposition 1.11.1** Let  $I_1, I_2, I_3, \ldots$  be an infinite sequence of closed intervals in the real line. If  $I_{j+1} \subseteq I_j$ 

(1.11.2)

for every positive integer j, then

(1.11.3) 
$$\bigcap_{j=1}^{\infty} I_j \neq \emptyset.$$

If  $j \in \mathbf{Z}_+$ , then there are real numbers  $a_j$ ,  $b_j$  such that

$$(1.11.4) a_j \le b_j$$

and  $I_j = [a_j, b_j]$ , by the definition of a closed interval in Subsection 1.5.3. We also have that

 $a_j \leq a_{j+1}$  and  $b_{j+1} \leq b_j$ (1.11.5)

for every  $j \ge 1$ , because of (1.11.2). Put

$$(1.11.6) A = \{a_j : j \in \mathbf{Z}_+\}.$$

Note that  $A \neq \emptyset$ , because  $a_1 \in A$ . Let us check that (1.11.7)  $a_j \leq b_l$ 

for all positive integers j and l. If  $j \leq l$ , then

$$(1.11.8) a_j \le a_l \le b_l.$$

Similarly, if  $j \ge l$ , then (1.11.9)  $a_j \le b_j \le b_l$ .

Thus, for each positive integer l,  $b_l$  is an upper bound of A. This implies that the supremum of A exists in  $\mathbf{R}$ , and satisfies

$$(1.11.10) \qquad \qquad \sup A \le b_l$$

for every  $l \ge 1$ . Of course,  $a_j \le \sup A$  for every  $j \ge 1$ , by construction. It follows that (1.11.11)  $\sup A \in I_j$ 

for every  $j \ge 1$ , as desired.

# **1.11.1** Some more properties of the intersection of the $I_j$ 's Put

(1.11.12) 
$$B = \{b_l : l \in \mathbf{Z}_+\}$$

which is another nonempty set of real numbers. Using (1.11.7), we get that  $a_j$  is a lower bound for B for each  $j \ge 1$ . This means that the infimum of B exists in  $\mathbf{R}$ , and satisfies
(1.11.13)  $a_j \le \inf B$ 

for every  $j \ge 1$ . We also have that  $\inf B \le b_l$  for every  $l \ge 1$ , so that

$$(1.11.14) \qquad \qquad \inf B \in I_j$$

for every  $j \ge 1$ . Observe that (1.11.15)  $\sup A \le \inf B$ ,

by (1.11.10) or (1.11.13). One can check that

(1.11.16) 
$$\bigcap_{j=1}^{\infty} I_j = [\sup A, \inf B]$$

However, this is not needed for Proposition 1.11.1.

## 1.12 Compactness of closed intervals

**Theorem 1.12.1** Let a and b be real numbers, with  $a \leq b$ . Under these conditions, the closed interval I = [a, b] is a compact subset of the real line, with respect to the standard Euclidean metric.

Let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of I in  $\mathbf{R}$ , and suppose for the sake of a contradiction that I cannot be covered by finitely many  $U_{\alpha}$ 's. Put

(1.12.2) L = [a, (a+b)/2] and R = [(a+b)/2, b],

which are the closed intervals corresponding to the left and right halves of I. In particular,

and

(1.12.4)  $\operatorname{length}(L) = \operatorname{length}(R) = \operatorname{length}(I)/2,$ 

where length (I) = b - a is the usual length of an interval. It follows from (1.12.3) that at least one of L and R cannot be covered by finitely many  $U_{\alpha}$ 's, because of the analogous property for I. Let us now choose  $I_1$  to be either L or R, in such a way that  $I_1$  cannot be covered by finitely many  $U_{\alpha}$ 's.

## 1.12.1 A sequence of closed intervals

Continuing in this way, we can find an infinite sequence  $I_1, I_2, I_3, \ldots$  of closed intervals in the real line with the following properties. First,

(1.12.5) 
$$I_1 \subseteq I$$
 and  $I_{j+1} \subseteq I_j$  for every  $j \ge 1$ .

Second,

(1.12.6) 
$$\operatorname{length}(I_1) = \operatorname{length}(I)/2 \quad \text{and} \\ \operatorname{length}(I_{j+1}) = \operatorname{length}(I_j)/2 \text{ for every } j \ge 1.$$

Third,

(1.12.7) for each  $j \in \mathbf{Z}_+$ ,  $I_j$  cannot be covered by finitely many  $U_{\alpha}$ 's.

More precisely, suppose that  $I_j = [a_j, b_j]$  has been chosen in this way for some positive integer j. Put

(1.12.8) 
$$L_j = [a_j, (a_j + b_j)/2]$$
 and  $R_j = [(a_j + b_j)/2, b_j],$ 

so that

 $(1.12.9) I_j = L_j \cup R_j$ 

and

(1.12.10)  $\operatorname{length}(L_i) = \operatorname{length}(R_i) = \operatorname{length}(I_i)/2.$ 

Using (1.12.7) and (1.12.9), we get that at least one of  $L_j$  and  $R_j$  cannot be covered by finitely many  $U_{\alpha}$ 's. We take  $I_{j+1}$  to be either  $L_j$  or  $R_j$ , in such a way that  $I_{j+1}$  cannot be covered by finitely many  $U_{\alpha}$ 's.

# 1.12.2 Using the nonemptiness of the intersection of the $I_i$ 's

Note that  $\bigcap_{j=1}^{\infty} I_j \neq \emptyset$ , by (1.12.5) and Proposition 1.11.1. Let x be an element of  $\bigcap_{j=1}^{\infty} I_j$ , which is also an element of I, because  $I_1 \subseteq I$ . This implies that there is an  $\alpha_0 \in A$  such that  $x \in U_{\alpha_0}$ , because I is covered by the  $U_{\alpha}$ 's. It follows that

$$(1.12.11) \qquad (x-r,x+r) \subseteq U_{\alpha_0}$$

for some positive real number r, because  $U_{\alpha_0}$  is an open set in **R**. If  $j \in \mathbf{Z}_+$  is sufficiently large, then one can check that

$$(1.12.12) \qquad \qquad \operatorname{length}(I_j) = 2^{-j} \operatorname{length}(I) < r.$$

In this case, we have that

(1.12.13) 
$$I_j \subseteq (x - r, x + r) \subseteq U_{\alpha_0},$$

because  $x \in I_j$ . This contradicts (1.12.7), as desired.

## 1.13 Cells in $\mathbb{R}^n$

Let n be a positive integer. If  $A_1, \ldots, A_n$  are n sets, then their Cartesian product may be denoted

$$(1.13.1) A_1 \times \cdots \times A_n$$

or

$$(1.13.2) \qquad \qquad \prod_{j=1}^n A_j.$$

This is the set of *n*-tuples  $a = (a_1, \ldots, a_n)$  such that  $a_j \in A_j$  for each  $j = 1, \ldots, n$ .

Now let  $a_1, b_1, \ldots, a_n, b_n$  be real numbers with  $a_j \leq b_j$  for each  $j = 1, \ldots, n$ , and put  $I_j = [a_j, b_j]$  for every  $j = 1, \ldots, n$ . The Cartesian product

(1.13.3) 
$$C = \prod_{j=1}^{n} I_j$$

of  $I_1, \ldots, I_n$  is called a *cell* in  $\mathbb{R}^n$ . The *diameter* of C is defined by

(1.13.4) 
$$\operatorname{diam} C = \left(\sum_{j=1}^{n} (b_j - a_j)^2\right)^{1/2}.$$

This is the same as the maximum of the distances between elements of C, with respect to the standard Euclidean metric on  $\mathbb{R}^n$ .

## 1.13.1 Nested sequences of cells

**Proposition 1.13.5** If  $C_1, C_2, C_3, \ldots$  is an infinite sequence of cells in  $\mathbb{R}^n$  such that

for every positive integer l, then

(1.13.7) 
$$\bigcap_{l=1}^{\infty} C_l \neq \emptyset$$

This is the same as Proposition 1.11.1 when n = 1, and we can reduce to that case otherwise. If  $l \in \mathbb{Z}_+$ , then there are *n* closed intervals  $I_{1,l}, \ldots, I_{n,l}$  in **R** such that

(1.13.8) 
$$C_l = \prod_{j=1}^n I_{j,l}$$

as in (1.13.3). Using (1.13.6), we get that

$$(1.13.9) I_{j,l+1} \subseteq I_{j,l}$$

for every positive integer l and j = 1, ..., n. This implies that

(1.13.10) 
$$\bigcap_{l=1}^{\infty} I_{j,l} \neq \emptyset$$

for every j = 1, ..., n, by Proposition 1.11.1. If  $x = (x_1, ..., x_n)$  has the property that

$$(1.13.11) x_j \in \bigcap_{l=1} I_{j,l}$$

for each  $j = 1, \ldots, n$ , then

$$(1.13.12) x \in \bigcap_{l=1}^{\infty} C_l,$$

as desired.

## 1.13.2 Some unions of smaller cells

**Proposition 1.13.13** If C is any cell in  $\mathbb{R}^n$ , then C can be expressed as the union of  $2^n$  cells, each of which has one-half the diameter of C.

As before, C can be expressed as in (1.13.3), where  $I_j = [a_j, b_j]$  is a closed interval in **R** for each j = 1, ..., n. Let  $L_j = [a_j, (a_j + b_j)/2]$  and  $R_j = [(a_j + b_j)/2, b_j]$  be the left and right halves of  $I_j$  for each j = 1, ..., n, so that  $I_j = L_j \cup R_j$  for every j = 1, ..., n. Consider the cells in  $\mathbf{R}^n$  that can be obtained by taking the Cartesian product of n closed intervals, where for each j = 1, ..., n, the *j*th interval is either  $L_j$  or  $R_j$ . It is easy to see that there are  $2^n$  of these cells. One can check that the union of these  $2^n$  cells is equal to C. One can also verify that the diameter of each of these  $2^n$  cells is equal to diam C/2. This uses the fact that the lengths of  $L_j$  and  $R_j$  are equal to length $(I_j)/2$  for each  $j = 1, \ldots, n$ .

## 1.14 Compactness of cells

**Theorem 1.14.1** If C is a cell in  $\mathbb{R}^n$  for some positive integer n, then C is compact, with respect to the standard Euclidean metric on  $\mathbb{R}^n$ .

Of course, this is the same as Theorem 1.12.1 when n = 1, and we can use essentially the same argument here. Let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of C in  $\mathbb{R}^n$ , and suppose for the sake of a contradiction that C cannot be covered by finitely many  $U_{\alpha}$ 's. Under these conditions, we can find an infinite sequence of cells  $C_1, C_2, C_3, \ldots$  with the following properties. First,

(1.14.2) 
$$C_l \subseteq C_{l-1}$$
 for every  $l \ge 1$ ,

where we put  $C_0 = C$ , for convenience. Second,

(1.14.3) 
$$\operatorname{diam} C_l = \operatorname{diam} C_{l-1}/2 \text{ for every } l \ge 1.$$

Third,

(1.14.4) for each  $l \in \mathbf{Z}_+$ ,  $C_l$  cannot be covered by finitely many  $U_{\alpha}$ 's.

To see this, suppose that  $C_l$  has already been chosen in this way for some nonnegative integer l. Using Proposition 1.13.13, we can express  $C_l$  as the union of  $2^n$  cells, each of which has one-half the diameter of  $C_l$ . At least one of these  $2^n$  cells cannot be covered by finitely many  $U_{\alpha}$ 's, because otherwise  $C_l$ could be covered by finitely many  $U_{\alpha}$ 's. We take  $C_{l+1}$  to be one of these  $2^n$ cells, in such a way that  $C_{l+1}$  cannot be covered by finitely many  $U_{\alpha}$ 's.

# 1.14.1 Using the nonemptiness of the intersection of the $C_l$ 's

It follows that  $\bigcap_{l=1}^{\infty} C_l \neq \emptyset$ , by Proposition 1.13.5. Let x be an element of  $\bigcap_{l=1}^{\infty} C_l$ , which is an element of C in particular. Thus there is an  $\alpha_0 \in A$  such that  $x \in U_{\alpha_0}$ . Because  $U_{\alpha_0}$  is an open set in  $\mathbb{R}^n$ , there is a positive real number r such that

(1.14.5)

$$B(x,r) \subseteq U_{\alpha_0}.$$

Here B(x, r) is the open ball in  $\mathbb{R}^n$  centered at x with radius r with respect to the standard Euclidean metric. Let l be a positive integer that is large enough so that

 $(1.14.6) \qquad \qquad \operatorname{diam} C_l = 2^{-l} \operatorname{diam} C < r.$ 

It is easy to see that

(1.14.7)  $C_l \subseteq B(x,r) \subseteq U_{\alpha_0},$ 

because  $x \in C_l$ . This contradicts (1.14.4), as desired.

## 1.14.2 Closed and bounded subsets of $\mathbb{R}^n$ are compact

**Corollary 1.14.8** If  $E \subseteq \mathbf{R}^n$  is closed and bounded, with respect to the standard Euclidean metric, then E is compact.

Indeed, one can find a cell  $C \subseteq \mathbf{R}^n$  such that  $E \subseteq C$ , because E is bounded in  $\mathbf{R}^n$ . To get that E is compact, one can use the compactness of C and the hypothesis that E be a closed set, as in Proposition 1.9.6.

## 1.15 Totally bounded sets

Let (M, d(x, y)) be a metric space.

**Definition 1.15.1** A subset E of M is said to be totally bounded with respect to  $d(\cdot, \cdot)$  if for every positive real number r there are finitely many elements  $x_1, \ldots, x_n$  of M such that

(1.15.2) 
$$E \subseteq \bigcup_{j=1}^{n} B(x_j, r).$$

If  $E = \emptyset$ , then this condition is interpreted as holding with n = 0, even if  $M = \emptyset$ .

## 1.15.1 Compact sets are totally bounded

**Proposition 1.15.3** If  $K \subseteq M$  is compact, then K is totally bounded.

Let r > 0 be given. The collection of open balls B(x,r) centered at elements x of K and with radius r forms an open covering of K in M. If K is compact, then there are finitely many elements  $x_1, \ldots, x_n$  of K such that  $K \subseteq \bigcup_{j=1}^n B(x_j, r)$ , as desired.

#### 1.15.2 The limit point property implies total boundedness

**Proposition 1.15.4** If  $E \subseteq M$  has the limit point property, then E is totally bounded in M.

Let r > 0 be given again. We would like to show that E is contained in the union of finitely many open balls in M of radius r. Of course, this is trivial when  $E = \emptyset$ , and so we may suppose that there is an element  $x_1$  of E. If

$$(1.15.5) E \subseteq B(x_1, r),$$

then we can stop. Otherwise, we let  $x_2$  be an element of  $E \setminus B(x_1, r)$ .

Suppose that  $x_1, \ldots, x_n \in E$  have been chosen in this way for some positive integer n. If  $E \subseteq \bigcup_{j=1}^n B(x_j, r)$ , then we can stop. Otherwise, we let  $x_{n+1}$  be an element of

(1.15.6) 
$$E \setminus \left( \bigcup_{j=1}^{n} B(x_j, r) \right).$$

Suppose for the sake of a contradiction that this process does not stop after finitely many steps. In this case, we get an infinite sequence  $x_1, x_2, x_3, \ldots$  of elements of E such that

$$(1.15.7) d(x_j, x_l) \ge r$$

for all positive integers j, l with j < l. Let

(1.15.8) 
$$L = \{x_j : j \in \mathbf{Z}_+\}$$

be the set of points that have been chosen in this way. Note that L has infinitely many elements, because  $x_j \neq x_l$  when j < l. Of course,  $L \subseteq E$ , by construction.

If E has the limit point property, then there is a point  $p \in E$  such that p is a limit point of L in M. This implies that B(p, r/2) contains infinitely many elements of L, as in Proposition 1.5.2. In particular, there are positive integers j, l such that j < l and  $x_j, x_l \in B(p, r/2)$ . It follows that

(1.15.9) 
$$d(x_i, x_l) \le d(x_i, p) + d(p, x_l) < r/2 + r/2 = r,$$

using the triangle inequality in the first step. This contradicts (1.15.7), as desired.

# Chapter 2

# Metric spaces, continued

# 2.1 Countable sets

To say that a nonempty set A has finitely many elements means that there is a finite sequence  $x_1, \ldots, x_n$  of elements of A in which every element of A occurs exactly once. In this case, the number of elements of A is the positive integer n. The empty set corresponds to taking n = 0.

**Definition 2.1.1** A set A is said to be countably infinite if there is a sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of A in which every element of A occurs exactly once. More precisely, this means that  $x_j \in A$  for every  $j \in \mathbf{Z}_+$ , and for every  $a \in A$ , there is a unique positive integer j such that  $x_j = a$ .

Of course, the set  $\mathbf{Z}_+$  of positive integers is countably infinite, since we can take  $x_j = j$  for every  $j \ge 1$ . The set  $\mathbf{Z}$  of all integers is countably infinite as well. To see this, one can take

(2.1.2) 
$$x_j = j/2$$
 when  $j$  is even  
=  $-(j-1)/2$  when  $j$  is odd.

#### 2.1.1 More on countability and sequences

**Proposition 2.1.3** Let A and B be sets, and let  $\{x_j\}_{j=1}^{\infty}$  be a sequence of elements of B in which every element of A occurs at least once. More precisely, this means that  $x_j \in B$  for every  $j \ge 1$ , and for every  $a \in A$  there is a positive integer j such that  $x_j = a$ . Under these conditions, A has only finitely or countably many elements.

Of course, if  $A = \emptyset$ , then A has only finitely many elements. Otherwise, let  $j_1$  be the smallest positive integer such that  $x_{j_1} \in A$ . If this is the only element of A, then we can stop. Otherwise, let  $j_2$  be the smallest positive integer such that  $x_{j_2} \in A$  and  $x_{j_2} \neq x_{j_1}$ . Note that  $j_1 < j_2$ , by construction.

Suppose that positive integers  $j_1 < \cdots < j_n$  have been chosen in this way for some  $n \in \mathbf{Z}_+$ . If  $x_{j_1}, \ldots, x_{j_n}$  are all of the elements of A, then we can stop. Otherwise, let  $j_{n+1}$  be the smallest positive integer such that  $x_{j_{n+1}} \in A$  and  $x_{j_{n+1}} \neq x_{j_l}$  for each  $l = 1, \ldots, n$ . By construction,  $j_n < j_{n+1}$ , since otherwise  $j_{n+1}$  would have been chosen earlier in the process.

If this process stops after finitely many steps, then A has only finitely many elements. Otherwise, we get an infinite sequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of elements of A in which every element of A occurs exactly once, so that A is countably infinite.

**Corollary 2.1.4** If A is a subset of a countably infinite set B, then A has only finitely or countably many elements.

#### 2.1.2 A lemma about unions of sequences of finite sets

**Lemma 2.1.5** If  $A_1, A_2, A_3, \ldots$  is an infinite sequence of finite sets, then their union  $\bigcup_{j=1}^{\infty} A_j$  has only finitely or countably many elements.

In this situation, it is not difficult to find a sequence in which every element of  $\bigcup_{j=1}^{\infty} A_j$  occurs at least once, by listing the elements of  $A_j$  for each  $j \in \mathbb{Z}_+$ , one after the other. To get that  $\bigcup_{j=1}^{\infty} A_j$  has only finitely of countably many elements, one can use Proposition 2.1.3.

Let us use the lemma to show that the set  $\mathbf{Z}_{+}^{2} = \mathbf{Z}_{+} \times \mathbf{Z}_{+}$  of all ordered pairs of positive integers is countably infinite. Put

(2.1.6) 
$$A_n = \{(j,l) \in \mathbf{Z}^2_+ : j+l = n+1\}$$

for each positive integer n. It is easy to see that  $A_n$  has exactly n elements for every n, and that

(2.1.7) 
$$\bigcup_{n=1}^{\infty} A_n = \mathbf{Z}_+^2.$$

It follows that  $\mathbf{Z}_{+}^{2}$  is countably infinite, by Lemma 2.1.5, and because  $\mathbf{Z}_{+}^{2}$  obviously has infinitely many elements. One can also use the fact that the  $A_{n}$ 's are pairwise disjoint, to enumerate the elements of  $\mathbf{Z}_{+}^{2}$  more directly.

#### 2.1.3 Sequences of countable sets

**Proposition 2.1.8** Let  $A_1, A_2, A_3, \ldots$  be an infinite sequence of sets, each of which has only finitely or countably many elements. Under these conditions,  $\bigcup_{i=1}^{\infty} A_j$  has only finitely or countably many elements as well.

By hypothesis, for each positive integer j, there is an infinite sequence  $\{x_{j,l}\}_{l=1}^{\infty}$  in which every element of  $A_j$  occurs at least once. Let  $\{(j_n, l_n)\}_{n=1}^{\infty}$  be a sequence of elements of  $\mathbf{Z}_+^2$  in which every element of  $\mathbf{Z}_+^2$  occurs once. It follows that every element of  $\bigcup_{j=1}^{\infty} A_j$  occurs at least once in the sequence  $\{x_{j_n,l_n}\}_{n=1}^{\infty}$ . This implies that  $\bigcup_{j=1}^{\infty} A_j$  has only finitely or countably many elements, as in Proposition 2.1.3.

Let us now use the proposition to show that the set  $\mathbf{Q}$  of rational numbers is countably infinite. If j is a positive integer, then let  $A_j$  be the set of integer multiples of 1/j. It is easy to see that  $A_j$  is countably infinite for each  $j \ge 1$ , because the set  $\mathbf{Z}$  of all integers is countably infinite. It follows that  $\mathbf{Q} = \bigcup_{j=1}^{\infty} A_j$  is countably infinite, as desired.

#### 2.1.4 Cartesian products of countable sets

**Proposition 2.1.9** Let n be a positive integer, and let  $A_1, \ldots, A_n$  be n countably infinite sets. Under these conditions,  $\prod_{i=1}^{n} A_i$  is countably infinite too.

Of course, this is trivial when n = 1, and the n = 2 case corresponds to the countability of  $\mathbf{Z}_{+}^{2}$ . If  $n \geq 2$ , then there is a simple one-to-one correspondence between  $\prod_{j=1}^{n} A_{j}$  and  $\left(\prod_{j=1}^{n-1} A_{j}\right) \times A_{n}$ . More precisely, if  $a = (a_{1}, \ldots, a_{n})$  is an element of  $\prod_{j=1}^{n} A_{j}$ , then

$$(2.1.10) \qquad ((a_1, \dots, a_{n-1}), a_n)$$

defines an element of  $\left(\prod_{j=1}^{n-1} A_j\right) \times A_n$ , and every element of  $\left(\prod_{j=1}^{n-1} A_j\right) \times A_n$  corresponds to a unique element of  $\prod_{j=1}^n A_j$  in this way. Thus the countability of  $\prod_{j=1}^n A_j$  can be obtained from the countability of  $\prod_{j=1}^{n-1} A_j$  and the n = 2 case. This permits one to prove the proposition by induction.

# 2.2 Uncountable sets

**Definition 2.2.1** A set is said to be uncountable if it is neither finite nor countably infinite.

Let  $\mathcal{B}$  be the set of all infinite sequences  $x = \{x_j\}_{j=1}^{\infty}$  such that for each  $j \in \mathbb{Z}_+, x_j = 0$  or 1. It is well known that  $\mathcal{B}$  is uncountable. To see this, let  $x(1), x(2), x(3), \ldots$  be any infinite sequence of elements of  $\mathcal{B}$ . Thus, for each  $l \in \mathbb{Z}_+, x(l) = \{x_j(l)\}_{j=1}^{\infty}$  is an infinite sequence with terms in  $\{0, 1\}$ . Put

(2.2.2) 
$$y_j = 1 - x_j(j) \in \{0, 1\}$$

for each  $j \in \mathbf{Z}_+$ , so that  $y = \{y_j\}_{j=1}^\infty$  defines an element of  $\mathcal{B}$ . If l is any positive integer, then  $y \neq x(l)$ , because  $y_l \neq x_l(l)$ , by construction. This shows that the elements of  $\mathcal{B}$  cannot all be listed by a sequence, so that  $\mathcal{B}$  is uncountable. This is Cantor's *diagonalization argument*.

There is a standard way to associate to each  $x \in \mathcal{B}$  an element of the closed unit interval [0, 1]. It is well known that every element of [0, 1] corresponds to an element of  $\mathcal{B}$  in this way, but some elements of [0, 1] may correspond to more than one element of  $\mathcal{B}$ . More precisely, it is well known and not too difficult to show that at most two elements of  $\mathcal{B}$  can correspond to the same element of [0, 1]. In fact, this only happens for countably many elements of  $\mathcal{B}$ , and hence countably many elements of [0, 1]. One can use this to show that [0, 1] is uncountable, because  $\mathcal{B}$  is uncountable.

#### 2.2.1 Perfect subsets of metric spaces

**Definition 2.2.3** Let (M, d(x, y)) be a metric space. A subset E of M is said to be perfect in M if E is a closed set, and if every element of E is a limit point of E.

Closed intervals in the real line of positive length are perfect, for instance, with respect to the standard Euclidean metric on  $\mathbf{R}$ . If E is a nonempty perfect subset of  $\mathbf{R}^n$  for some positive integer n, with respect to the standard Euclidean metric, then it can be shown that E is uncountable. This can be extended to a broad class of metric spaces using the Baire category theorem. Another type of example of a perfect set in  $\mathbf{R}$  will be discussed in Section 2.9.

# 2.3 Separable metric spaces

Let (M, d(x, y)) be a metric space.

#### 2.3.1 Dense subsets of metric spaces

**Definition 2.3.1** A subset E of M is said to be dense in M if every element of M is either an element of E, a limit point of E, or both.

Equivalently,  $E \subseteq M$  is dense in M if the closure  $\overline{E}$  of E in M is equal to M.

#### 2.3.2 The definition of separability, and some examples

**Definition 2.3.2** If there is a dense set  $E \subseteq M$  such that E has only finitely or countably many elements, then M is said to be separable as a metric space.

It is easy to see that the set  $\mathbf{Q}$  of rational numbers is dense in the real line, with respect to the standard Euclidean metric. Thus  $\mathbf{R}$  is separable, with respect to the standard metric, because  $\mathbf{Q}$  is countably infinite, as in Subsection 2.1.8.

Similarly, if n is a positive integer, then let  $\mathbf{Q}^n$  be the set of  $x \in \mathbf{R}^n$  such that  $x_j \in \mathbf{Q}$  for each j = 1, ..., n. One can check that  $\mathbf{Q}^n$  is dense in  $\mathbf{R}^n$ , with respect to the standard Euclidean metric. It follows from Proposition 2.1.9 that  $\mathbf{Q}^n$  is countably infinite, so that  $\mathbf{R}^n$  is separable with respect to the standard metric.

#### 2.3.3 A characterization of separability

**Proposition 2.3.3** Separability of M is equivalent to the following condition: for every r > 0 there is a subset E(r) of M such that E(r) has only finitely or countably many elements and

(2.3.4) 
$$\bigcup_{x \in E(r)} B(x, r) = M.$$

#### 2.4. BASES

If M is separable, then there is a dense set  $E \subseteq M$  with only finitely or countably many elements. In this case, one can check that (2.3.4) holds with E(r) = E for every r > 0.

Conversely, suppose that M satisfies the condition described in the statement of the proposition. This implies that for each positive integer j there is a subset E(1/j) of M with only finitely or countably many elements such that (2.3.4) holds with r = 1/j. Put

(2.3.5) 
$$E = \bigcup_{j=1}^{\infty} E(1/j),$$

which has only finitely or countably many elements, by Proposition 2.1.8. One can check that E is dense in M, using (2.3.4) for r = 1/j,  $j \in \mathbb{Z}_+$ . This means that M is separable, as desired.

#### 2.3.4 Total boundedness implies separability

Corollary 2.3.6 If M is totally bounded, then M is separable.

Indeed, if M is totally bounded, then for each r > 0 there is a finite subset E(r) of M that satisfies (2.3.4).

## 2.4 Bases

The topics in this section are a bit abstract, and one may wish to skip this at first. We shall say more about this in the next section.

Let (M, d(x, y)) be a metric space.

**Definition 2.4.1** A collection  $\mathcal{B}$  of open subsets of M is said to be a base for the topology of M if for every  $x \in M$  and r > 0 there is a  $V \in \mathcal{B}$  such that  $x \in V$  and  $V \subseteq B(x, r)$ .

If  $\mathcal{B}$  is a base for the topology of M and  $W \subseteq M$  is an open set, then

$$(2.4.2) W = \bigcup \{ V : V \in \mathcal{B}, \ V \subseteq W \}$$

More precisely, the right side of (2.4.2) is automatically contained in W, by construction. In order to verify the opposite inclusion, let an arbitrary element x of W be given. Thus there is an r > 0 such that  $B(x,r) \subseteq W$ , because W is an open set in M. This implies that there is a  $V \in \mathcal{B}$  such that  $x \in V$  and  $V \subseteq B(x,r) \subseteq W$ , as in Definition 2.4.1. It follows that x is an element of the right side of (2.4.2). This means that W is contained in the right side of (2.4.2), as desired.

Conversely, let  $\mathcal{B}$  be a collection of open subsets of M, and suppose that every open subset of M can be expressed as the union of some elements of  $\mathcal{B}$ . Let  $x \in M$  and r > 0 be given, and remember that B(x,r) is an open set in M. By hypothesis, B(x,r) can be expressed as a union of elements of  $\mathcal{B}$ . In particular, there is a  $V \in \mathcal{B}$  such that  $x \in V$  and  $V \subseteq B(x,r)$ , because  $x \in B(x,r)$ . This shows that  $\mathcal{B}$  is a base for the topology of M.

#### 2.4.1 Some examples of bases

Of course, the collection of all open subsets of M is a base for the topology of M. The collection of all open balls in M is a base for the topology of M as well, because of the way that open sets in M are defined. The collection of open balls in M with radius of the form 1/j for some positive integer j is a base for the topology of M too, because of the archimedean property of the real numbers.

**Proposition 2.4.3** Let E be a subset of M, and let  $\mathcal{B}_E$  be the collection of open balls in M of the form B(x, 1/j), with  $x \in E$  and  $j \in \mathbb{Z}_+$ . If E is dense in M, then  $\mathcal{B}_E$  is a base for the topology of M.

It is easy to see that E has to be dense in M in order for  $\mathcal{B}_E$  to be a base for the topology of M.

Suppose that E is dense in M, and let us check that  $\mathcal{B}_E$  is a base for the topology of M. Let  $x \in M$  and r > 0 be given, as in Definition 2.4.1. The archimedean property for the real numbers implies that there is a positive integer j such that  $2/j \leq r$ . Because E is dense in M, there is a  $y \in E$  such that d(x, y) < 1/j. Thus  $B(y, 1/j) \in \mathcal{B}_E$  and  $x \in B(y, 1/j)$ . One can also verify that

$$(2.4.4) B(y,1/j) \subseteq B(x,r),$$

using the triangle inequality. More precisely, if  $z \in B(y, 1/j)$ , then

$$(2.4.5) d(x,z) \le d(x,y) + d(y,z) < 1/j + 1/j = 2/j \le r,$$

so that  $z \in B(x, r)$ , as desired.

#### 2.4.2 Separability and countable bases

**Corollary 2.4.6** If M is separable, then there is a base for the topology of M with only finitely or countably many elements.

Suppose that M is separable, and let E be a dense subset of M with only finitely or countably many elements. It suffices to check that the collection  $\mathcal{B}_E$ defined in Proposition 2.4.3 has only finitely or countably many elements. If jis a positive integer, then let  $\mathcal{B}_{E,j}$  be the collection of open balls of the form B(x, 1/j), with  $x \in E$ . Observe that

(2.4.7) 
$$\mathcal{B}_E = \bigcup_{j=1}^{\infty} \mathcal{B}_{E,j},$$

by the definition of  $\mathcal{B}_E$ . It is easy to see that  $\mathcal{B}_{E,j}$  has only finitely or countably many elements for each  $j \in \mathbb{Z}_+$ , because E has only finitely or countably many elements, by hypothesis. More precisely, one can list the elements of E with a finite or infinite sequence, and use that to list the elements of  $\mathcal{B}_{E,j}$  with a finite or infinite sequence. This implies that (2.4.7) has only finitely or countably many elements, by Proposition 2.1.8.

#### 2.4.3 Necessity of separability

**Proposition 2.4.8** If there is a base  $\mathcal{B}$  for the topology of M such that  $\mathcal{B}$  has only finitely or countably many elements, then M is separable.

Let  $\mathcal{B}$  be a base for the topology of M. If  $V \in \mathcal{B}$  and  $V \neq \emptyset$ , then let us pick a point  $x_V$  in V. Let (2.4.9)  $E = \{x_V : V \in \mathcal{B}, V \neq \emptyset\}$ 

be the set of points in M that have been chosen in this way. One can check that E is dense in M, because  $\mathcal{B}$  is a base for the topology of M. If  $\mathcal{B}$  has only finitely or countably many elements, then E has only finitely or countably many elements. More precisely, if one can list the elements of  $\mathcal{B}$  with a finite or infinite sequence, then one can list the elements of E with a finite or infinite sequence too. In this case, we get that M is separable, as desired.

# 2.5 Lindelöf's theorem

The topics in this section are also a bit abstract, and one may wish to skip this at first. More precisely, the conclusion of the next theorem is quite interesting, and one may wish to skip the proof at first.

Let (M, d(x, y)) be a metric space.

**Theorem 2.5.1 (Lindelöf's theorem)** Suppose that there is a base  $\mathcal{B}$  for the topology of M with only finitely or countably many elements. If  $\{U_{\alpha}\}_{\alpha \in A}$  is any family of open subsets of M, then there is a subset  $A_1$  of A such that  $A_1$  has only finitely or countably many elements and

(2.5.2) 
$$\bigcup_{\alpha \in A_1} U_{\alpha} = \bigcup_{\alpha \in A} U_{\alpha}.$$

Note that the hypothesis of Lindelöf's theorem is the same as saying that M is separable as a metric space, as in the previous section. Lindelöf's theorem will be used in the proof of Theorem 2.7.1.

#### 2.5.1 The proof of Lindelöf's theorem

To prove Lindelöf's theorem, let  $\alpha \in A$  be given, and put

$$(2.5.3) \qquad \qquad \mathcal{B}_{\alpha} = \{ V \in \mathcal{B} : V \subseteq U_{\alpha} \}.$$

Note that (2.5.4)  $U_{\alpha} = \bigcup_{V \in \mathcal{B}_{\alpha}} V,$ 

as in (2.4.2), because  $\mathcal{B}$  is a base for the topology of M, and  $U_{\alpha}$  is an open set in M. Put

$$(2.5.5) \qquad \qquad \mathcal{B} = \bigcup_{\alpha \in A} \mathcal{B}_{\alpha}$$

It is easy to see that

(2.5.6) 
$$\bigcup_{V\in\widetilde{\mathcal{B}}} V = \bigcup_{\alpha\in A} \bigcup_{V\in\mathcal{B}_{\alpha}} V = \bigcup_{\alpha\in A} U_{\alpha},$$

using the definition (2.5.5) of  $\widetilde{\mathcal{B}}$  in the first step, and (2.5.4) in the second step. If  $V \in \widetilde{\mathcal{B}}$ , then let us pick  $\alpha(V) \in A$  such that  $V \in \mathcal{B}_{\alpha(V)}$ , which means that

$$(2.5.7) V \subseteq U_{\alpha(V)}.$$

 $\begin{array}{c} \text{Let} \\ (2.5.8) \end{array}$ 

be the collection of elements of A that have been chosen in this way. Observe that  $\widetilde{\mathcal{B}}$  has only finitely or countably many elements, because  $\mathcal{B}$  has only finitely of countably many elements, by hypothesis, and  $\widetilde{\mathcal{B}} \subseteq \mathcal{B}$ , by construction. This implies that  $A_1$  has only finitely or countably many elements as well. More precisely, the elements of  $\widetilde{\mathcal{B}}$  can be listed by a finite or infinite sequence, which can be used to list the elements of  $A_1$  by a finite or infinite sequence.

 $A_1 = \{ \alpha(V) : V \in \widetilde{\mathcal{B}} \}$ 

We also have that

(2.5.9) 
$$\bigcup_{\alpha \in A_1} U_{\alpha} = \bigcup_{V \in \widetilde{\mathcal{B}}} U_{\alpha(V)} \supseteq \bigcup_{V \in \widetilde{\mathcal{B}}} V = \bigcup_{\alpha \in A} U_{\alpha},$$

using the definition (2.5.8) of  $A_1$  in the first step, (2.5.7) in the second step, and (2.5.6) in the third step. Of course,

(2.5.10) 
$$\bigcup_{\alpha \in A_1} U_\alpha \subseteq \bigcup_{\alpha \in A} U_\alpha$$

automatically, because  $A_1 \subseteq A$ . Combining (2.5.9) and (2.5.10), we get (2.5.2), as desired.

# 2.6 Countable open coverings

Let (M, d(x, y)) be a metric space.

**Proposition 2.6.1** Suppose that  $E \subseteq M$  has the limit point property. If  $U_1, U_2$ ,  $U_3, \ldots$  is a sequence of open subsets of M such that

(2.6.2) 
$$E \subseteq \bigcup_{j=1}^{\infty} U_j,$$

then there is a positive integer n such that

(2.6.3) 
$$E \subseteq \bigcup_{j=1}^{n} U_j.$$

#### 2.7. GETTING COMPACTNESS

Suppose for the sake of a contradiction that

$$(2.6.4) E \not\subseteq \bigcup_{j=1}^{n} U_{j}$$

for every  $n \in \mathbf{Z}_+$ . Let us choose, for each  $n \in \mathbf{Z}_+$ , a point

(2.6.5) 
$$x_n \in E \setminus \Big(\bigcup_{j=1}^n U_j\Big).$$

Let

$$(2.6.6) L = \{x_n : n \in \mathbf{Z}_+\}$$

be the set of points in E that have been chosen in this way.

#### 2.6.1 Checking that *L* has infinitely many elements

Let us check that L has infinitely many elements. Otherwise, if L has only finitely many elements, then there is an  $x \in E$  such that  $x = x_n$  for infinitely many  $n \in \mathbb{Z}_+$ . Because  $x \in E$ , there is a positive integer  $j_0$  such that  $x \in U_{j_0}$ , by (2.6.2). It follows that  $x_n \neq x$  when  $n \geq j_0$ , by (2.6.5). This means that  $x_n = x$  for only finitely many n, which is a contradiction, as desired.

#### 2.6.2 Using the limit point property

If E has the limit point property, then there is a point  $p \in E$  that is a limit point of L in M. Using (2.6.2), we get that there is a positive integer  $j_1$  such that  $p \in U_{j_1}$ . Because  $U_{j_1}$  is an open set in M, by hypothesis, there is a positive real number r such that

$$(2.6.7) B(p,r) \subseteq U_{j_1}$$

We also have that B(p,r) contains infinitely many elements of L, because p is a limit point of L, as in Proposition 1.5.2. This implies that  $x_n \in B(p,r)$  for infinitely many  $n \in \mathbb{Z}_+$ .

It follows that

(2.6.8)

$$x_n \in U_{j_1}$$

for infinitely many  $n \in \mathbb{Z}_+$ , by (2.6.7). However, (2.6.8) can only hold when  $n < j_1$ , by (2.6.5). This is a contradiction, as desired.

# 2.7 Getting compactness

The proof of the next theorem uses Lindelöf's theorem, as well as other results discussed previously. One should not consider the next theorem as being completely established here unless one has gone through the proof of Lindelöf's theorem, as well as that of Corollary 2.4.6.

Let (M, d(x, y)) be a metric space.

**Theorem 2.7.1** If  $E \subseteq M$  has the limit point property, then E is compact.

Remember that the converse was given in Proposition 1.7.10.

#### **2.7.1** The case where E = M

Suppose first that M has the limit point property, and let us show that M is compact. Remember that M is totally bounded in this situation, as in Proposition 1.15.4. This implies that M is separable, as in Corollary 2.3.6. It follows that there is a base  $\mathcal{B}$  for the topology of M with only finitely or countably many elements, as in Corollary 2.4.6.

Let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of M, as a subset of itself. Thus  $\bigcup_{\alpha \in A} U_{\alpha} = M$ . Lindelöf's theorem implies that there is a subset  $A_1$  of A such that  $A_1$  has only finitely or countably many elements, and

(2.7.2) 
$$\bigcup_{\alpha \in A_1} U_{\alpha} = M.$$

If  $A_1$  has only finitely many elements, then we can stop. Otherwise, suppose that  $A_1$  is countably infinite. In this case, we can use Proposition 2.6.1 to find a finite subset  $A_2$  of  $A_1$  such that

$$(2.7.3) M \subseteq \bigcup_{\alpha \in A_2} U_{\alpha}.$$

In both cases, we get a finite subcovering of M from  $\{U_{\alpha}\}_{\alpha \in A}$ , as desired.

#### **2.7.2** Subsets of M

Suppose now that E is a subset of M with the limit point property. Remember that E may be considered as a metric space as well, with respect to the restriction of d(x, y) to  $x, y \in E$ . It is easy to see that E has the limit point property as a subset of itself, because of the corresponding property of E in M. Using this, the previous argument implies that E is compact as a subset of itself. It follows that E is compact as a subset of M, as in Proposition 1.10.12.

## 2.8 Connectedness

Let (M, d(x, y)) be a metric space.

#### **2.8.1** Separated subsets of M

**Definition 2.8.1** A pair A, B of subsets of M are said to be separated in M if

(2.8.2) 
$$\overline{A} \cap B = A \cap \overline{B} = \emptyset.$$

If we take A = (0,1) and B = (1,2) in the real line, then their closures with respect to the standard Euclidean metric on **R** are given by  $\overline{A} = [0,1]$  and  $\overline{B} = [1,2]$ . It follows that A and B are separated in **R**, although their closures are not disjoint in **R**.

#### 2.8.2 Connected sets

**Definition 2.8.3** A subset E of M is said to be connected in M if E cannot be expressed as the union of two nonempty separated subsets of M.

#### 2.8.3 Connected subsets of R

**Theorem 2.8.4** A subset E of the real line is connected with respect to the standard metric on  $\mathbf{R}$  if and only if it has the following property:

(2.8.5) if 
$$x, y \in E$$
,  $w \in \mathbf{R}$ , and  $x < w < y$ , then  $w \in E$ .

Suppose first that (2.8.5) does not hold, and let us show that E is not connected in **R**. By hypothesis, there are  $x, y \in E$  and  $w \in \mathbf{R}$  such that x < w < y and  $w \notin E$ . Put

(2.8.6) 
$$A = \{ u \in E : u < w \}$$
 and  $B = \{ u \in E : w < u \}.$ 

Thus  $x \in A$  and  $y \in B$ , so that  $A, B \neq \emptyset$ , and  $E = A \cup B$ , because  $w \notin E$ . It is easy to see that

(2.8.7) 
$$\overline{A} \subseteq \{u \in \mathbf{R} : u \le w\} \text{ and } \overline{B} \subseteq \{u \in \mathbf{R} : w \le u\}.$$

This implies that (2.8.2) holds, so that A and B are separated in **R**. It follows that E is not connected in **R**, as desired.

#### 2.8.4 Suprema, infima, and closures

Before proceeding to the other half of the theorem, let us mention the following useful fact.

**Proposition 2.8.8** If A is a nonempty subset of  $\mathbf{R}$  with an upper bound in  $\mathbf{R}$ , then  $\sup A \in \overline{A}$ . Similarly, if B is a nonempty subset of  $\mathbf{R}$  with a lower bound in  $\mathbf{R}$ , then  $\inf B \in \overline{B}$ .

This can be verified directly from the definitions, and the details are left as an exercise.

Let us now prove the "if" part of Theorem 2.8.4. Thus we suppose that (2.8.5) holds, and we would like to show that E is connected in **R**. Suppose for the sake of a contradiction that E is not connected, so that there are nonempty separated sets  $A, B \subseteq \mathbf{R}$  such that  $E = A \cup B$ . Let x be an element of A, and let y be an element of B. We may as well suppose that x < y, since otherwise we could interchange the roles of A and B.

Note that  $A \cap [x, y]$  is a nonempty subset of **R** with an upper bound in **R**, and put

(2.8.9) 
$$z = \sup(A \cap [x, y]).$$

Clearly  $x \leq z \leq y$ , because  $x \in A \cap [x, y]$ , and y is an upper bound for  $A \cap [x, y]$ . Using Proposition 2.8.8, we get that z is in the closure of  $A \cap [x, y]$  in **R**. It is easy to see that this implies that  $z \in \overline{A}$ . It follows that  $z \notin B$ , because  $\overline{A} \cap B = \emptyset$ , by hypothesis.

Using (2.8.5), we get that  $z \in E$ . This implies that  $z \in A$ , because  $z \notin B$ . Of course, z < y, because  $z \le y$  and  $z \ne y$ , since  $y \in B$  and  $z \notin B$ . We can use (2.8.5) again to get that  $(z, y) \subseteq E$ , because  $x \le z$ . It follows that

$$(2.8.10) (z,y) \subseteq B,$$

because  $E = A \cup B$ , and (z, y) is disjoint from A, by the definition (2.8.9) of z. This means that  $z \in \overline{B}$ . This contradicts the hypothesis that  $A \cap \overline{B} = \emptyset$ , as desired.

# 2.9 The Cantor set

Put  $E_0 = [0, 1]$  and  $E_1 = [0, 1/3] \cup [2/3, 1]$ . Equivalently,  $E_1$  is obtained from  $E_0$  be removing the open middle third (1/3, 2/3). Similarly, we can remove the open middle thirds (1/9, 2/9) and (7/9, 8/9) from [0, 1/3] and [2/3, 1], respectively, to get

(2.9.1) 
$$E_2 = [0, 1/9] \cup [2/9, 1/3] \cup [2/3, 7/9] \cup [8/9, 1].$$

Continuing in this way, we can define a subset  $E_j$  of **R** for every nonnegative integer j with the following properties. First, for each  $j \ge 0$ ,  $E_j$  is the union of  $2^j$  pairwise-disjoint closed intervals of length  $3^{-j}$ . Second,  $E_{j+1}$  is obtained from  $E_j$  by removing the open middle thirds of each of the  $2^j$  closed intervals in  $E_j$  just mentioned. In particular,

$$(2.9.2) E_{j+1} \subseteq E_j$$

for every  $j \ge 0$ .

The middle-thirds *Cantor set* is defined to be

(2.9.3) 
$$E = \bigcap_{j=0}^{\infty} E_j.$$

Note that  $E_j$  is a closed set in **R** with respect to the standard Euclidean metric for every  $j \ge 0$ , because it is the union of finitely many closed sets. It follows that E is a closed set in **R** too.

#### **2.9.1** Some elements of E

If j is a nonnegative integer, then let  $A_j$  be the set of  $2^{j+1}$  elements of [0,1] that occur as endpoints of the  $2^j$  closed intervals that make up  $E_j$ . Thus

 $\begin{array}{ll} A_0 = \{0,1\}, \ A_1 = \{0,1/3,2/3,1\}, \ A_2 = \{0,1/9,2/9,1/3,2/3,7/9,8/9,1\}, \ \text{etc.} \\ \text{By construction}, \\ (2.9.4) \qquad \qquad A_j \subseteq E_j \end{array}$ 

for every  $j \ge 0$ . We also have that

$$(2.9.5) A_j \subseteq A_{j+1}$$

for every  $j \ge 0$ . More precisely,  $A_{j+1}$  consists of the elements of  $A_j$ , together with the endpoints of the open middle thirds of the closed intervals that make up  $E_j$ .

 $A_i \subseteq E_l$ 

Let us check that 
$$(2.9.6)$$

for all nonnegative integers j, l. If  $j \leq l$ , then

using (2.9.5) in the first step, and (2.9.4) in the second step. Similarly, if  $j \geq l,$  then

using (2.9.4) in the first step, and (2.9.2) in the second step. It follows that

for every  $j \ge 0$ .

Observe that

is countably infinite. Clearly

$$(2.9.11) A \subseteq$$

by (2.9.9). Let us check that every element of E is a limit point of A, with respect to the standard metric on  $\mathbf{R}$ .

E,

 $\infty$ 

#### **2.9.2** Every element of E is a limit point of A

If  $x \in E$  and j is a nonnegative integer, then  $x \in E_j$ , and hence x is contained in one of the  $2^j$  closed intervals I that make up  $E_j$ . The endpoints of I are elements of  $A_j \subseteq A$ , at least one of which is different from x. The distance from x to the endpoints of I is less than or equal to the length of I, which is  $3^{-j}$ . Using this, it is easy to see that x is a limit point of A, as desired.

In particular, every element of E is a limit point of E, because of (2.9.11). This shows that E is perfect with respect to the standard metric on  $\mathbf{R}$ , because E is a closed set.

# Chapter 3

# Sequences and series

# 3.1 Complex numbers

Every complex number z can be expressed in a unique way as z = x + y i, where  $x, y \in \mathbf{R}$  and  $i^2 = -1$ . In this case, x and y are known as the *real* and *imaginary* parts of z, respectively.

#### 3.1.1 Complex conjugates and absolute values

The *complex conjugate* of z is the complex number

$$(3.1.1) \qquad \qquad \overline{z} = x - y \, i.$$

The *absolute value* or *modulus* of z is the nonnegative real number

(3.1.2) 
$$|z| = (x^2 + y^2)^{1/2}.$$

Note that the complex conjugate of  $\overline{z}$  is z, and that  $|\overline{z}| = |z|$ .

### 3.1.2 Addition and multiplication

Addition and multiplication of real numbers can be extended to the set  $\mathbf{C}$  of complex numbers in a standard way. If  $z, w \in \mathbf{C}$ , then one can check that

(3.1.3) 
$$\overline{z+w} = \overline{z} + \overline{w}$$
 and

(3.1.4)  $\overline{z w} = \overline{z} \overline{w}.$ We also have that (3.1.5)  $z \overline{z} = |z|^2$ for every  $z \in \mathbf{C}$ . It follows that

$$(3.1.6) |zw|^2 = zw\overline{zw} = z\overline{z}w\overline{w} = |z|^2|w|^2$$

for every  $z, w \in \mathbf{C}$ , so that (3.1.7)

If  $z \in \mathbf{C}$  and  $z \neq 0$ , then |z| > 0, and

which means that  $\overline{z}/|z|^2$  is the multiplicative inverse of z in **C**.

#### 3.1.3 The triangle inequality

It is well known that  $(3.1.9) \qquad \qquad |z+w| \le |z|+|w|$ 

for every  $z, w \in \mathbf{C}$ , which can be verified directly. If we identify  $(x, y) \in \mathbf{R}^2$  with  $z = x + yi \in \mathbf{C}$ , then the absolute value of a complex number corresponds exactly to the standard Euclidean norm on  $\mathbf{R}^2$ . Thus (3.1.9) is the same as the triangle inequality for the standard Euclidean norm on  $\mathbf{R}^2$ .

|zw| = |z| |w|.

It is easy to see that (3.1.10) d(z,w) = |z - w|

defines a metric on  $\mathbf{C}$ , which is the standard metric on  $\mathbf{C}$ . This corresponds exactly to the standard Euclidean metric on  $\mathbf{R}^2$ .

## **3.2** Convergent sequences

Let (M, d(x, y)) be a metric space.

**Definition 3.2.1** A sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of M is said to converge to an element x of M if for every positive real number  $\epsilon$  there is a positive integer L such that (3.2.2)  $d(x_i, x) < \epsilon$ 

for every  $j \geq L$ .

#### 3.2.1 Uniqueness of the limit

**Proposition 3.2.3** If  $\{x_j\}_{j=1}^{\infty}$  is a sequence of elements of M that converges to  $x \in M$  and to  $x' \in M$ , then x = x'.

Suppose for the sake of a contradiction that  $x \neq x'$ , so that d(x, x') > 0. Because  $\{x_j\}_{j=1}^{\infty}$  converges to x in M, there is a positive integer L such that

(3.2.4) 
$$d(x_j, x) < d(x, x')/2$$

for every  $j \ge L$ . Similarly, there is an  $L' \in \mathbf{Z}_+$  such that

(3.2.5) 
$$d(x_i, x') < d(x, x')/2$$

for every  $j \ge L'$ , because  $\{x_j\}_{j=1}^{\infty}$  converges to x' in M. If  $j \ge \max(L, L')$ , then it follows that

$$(3.2.6) \quad d(x,x') \le d(x,x_j) + d(x_j,x') < d(x,x')/2 + d(x,x')/2 = d(x,x'),$$

using the triangle inequality in the first step. In particular, this holds when  $j = \max(L, L')$ , which is a contradiction, as desired.

If  $\{x_j\}_{j=1}^{\infty}$  is a sequence of elements of M that converges to an element x of M, then x is called the *limit* of  $\{x_j\}_{j=1}^{\infty}$ , and we put

$$\lim_{j \to \infty} x_j = x.$$

We may also say that  $x_j$  tends to x as  $j \to \infty$  in this case, or  $x_j \to x$  as  $j \to \infty$ .

#### **3.2.2** Bounded sequences

**Definition 3.2.8** A sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of M is said to be bounded in M if the set

 $(3.2.9) \qquad \{x_j : j \in \mathbf{Z}_+\}$ 

of its terms is bounded in M.

#### 3.2.3 Convergent sequences are bounded

**Proposition 3.2.10** If  $\{x_j\}_{j=1}^{\infty}$  is a sequence of elements of M that converges to an element x of M, then  $\{x_j\}_{j=1}^{\infty}$  is bounded in M.

Because  $\{x_j\}_{j=1}^{\infty}$  converges to x in M, there is a positive integer L such that

(3.2.11) 
$$d(x_i, x) < 1$$

for every  $j \ge L$ . If L = 1, then it follows that the set of the terms in this sequence is contained in B(x, 1), as desired. Otherwise, put

(3.2.12) 
$$r = \max\{d(x_j, x) : j = 1, \dots, L-1\},\$$

which is a nonnegative real number. It is easy to see that

$$(3.2.13) d(x_i, x) < r+1$$

for every  $j \ge 1$ , so that the set of terms in the sequence is contained in B(x, r+1), as desired.

#### **3.2.4** Monotonic sequences of real numbers

**Proposition 3.2.14** (a) Let  $\{x_j\}_{j=1}^{\infty}$  be a sequence of real numbers that increases monotonically, in the sense that

$$(3.2.15) x_j \le x_{j+1}$$

for every  $j \geq 1$ . If the set of  $x_j$ 's,  $j \in \mathbb{Z}_+$ , has an upper bound in  $\mathbb{R}$ , then  $\{x_j\}_{j=1}^{\infty}$  converges to a real number x, with respect to the standard Euclidean metric on  $\mathbb{R}$ .

(b) Let  $\{y_j\}_{j=1}^{\infty}$  be a monotonically decreasing sequence of real numbers, so that

$$(3.2.16)$$
  $y_{j+1} \le y_j$ 

for every  $j \ge 1$ . If the set of  $y_j$ 's has a lower bound in  $\mathbf{R}$ , then  $\{y_j\}_{j=1}^{\infty}$  converges to a real number y, with respect to the standard metric on  $\mathbf{R}$ .

To prove (a), put  $A = \{x_j : j \in \mathbb{Z}_+\}$ , and  $x = \sup A$ . Let  $\epsilon > 0$  be given, and note that  $x - \epsilon$  is not an upper bound for A, by definition of the supremum. Thus there is a positive integer L such that

$$(3.2.17) x - \epsilon < x_L.$$

Of course,  $x_j \leq x$  for every  $j \geq 1$ , because x is an upper bound for A. If  $j \geq L$ , then

$$(3.2.18) x_j \ge x_L > x - \epsilon,$$

because the  $x_j$ 's increase monotonically. It follows that

$$(3.2.19) |x_j - x| = x - x_j < \epsilon$$

for every  $j \ge L$ , as desired. Part (b) can be shown in essentially the same way, or by reducing to the previous case.

# 3.3 Sums and products

In this section, we consider some basic properties of convergent sequences of complex numbers. Of course, this uses the standard metric on  $\mathbf{C}$ .

#### 3.3.1 Sums of convergent sequences

**Proposition 3.3.1** Let  $\{z_j\}_{j=1}^{\infty}$  and  $\{w_j\}_{j=1}^{\infty}$  be sequences of complex numbers that converge to complex numbers z and w, respectively. Under these conditions,  $\{z_j + w_j\}_{j=1}^{\infty}$  converges to z + w in **C**.

Thus

(3.3.2) 
$$\lim_{j \to \infty} (z_j + w_j) = \lim_{j \to \infty} z_j + \lim_{j \to \infty} w_j$$

where more precisely the limits on the right exist by hypothesis, and the existence of the limit on the left is part of the conclusion. To see this, let  $\epsilon > 0$  be given. Because  $\{z_j\}_{j=1}^{\infty}$  converges to z in  $\mathbf{C}$ , there is a positive integer  $L_1$  such that

$$(3.3.3) |z_j - z| < \epsilon/2$$

for every  $j \ge L_1$ . Similarly, because  $\{w_j\}_{j=1}^{\infty}$  converges to w in  $\mathbb{C}$ , there is an  $L_2 \in \mathbb{Z}_+$  such that (3.3.4)  $|w_j - w| < \epsilon/2$ 

for every  $j \ge L_2$ . If  $j \ge \max(L_1, L_2)$ , then it follows that

$$(3.3.5) |(z_j + w_j) - (z + w)| = |(z_j - z) + (w_j - w)| \leq |z_j - z| + |w_j - w| < \epsilon/2 + \epsilon/2 = \epsilon,$$

as desired.

#### 3.3.2 A lemma about products

**Lemma 3.3.6** If  $\{a_j\}_{j=1}^{\infty}$  is a sequence of complex numbers that converges to 0, and  $\{b_j\}_{j=1}^{\infty}$  is a bounded sequence of complex numbers, then  $\{a_j, b_j\}_{j=1}^{\infty}$  converges to 0 in **C**.

The boundedness of  $\{b_j\}_{j=1}^{\infty}$  in **C** implies that there is a positive real number B such that

$$(3.3.7) |b_j| \le B$$

for every  $j \ge 1$ . Let  $\epsilon > 0$  be given. Because  $\{a_j\}_{j=1}^{\infty}$  converges to 0 in **C**, there is a positive integer L such that

$$(3.3.8) |a_j| < \epsilon/B$$

for every  $j \ge L$ . This implies that

(3.3.9) 
$$|a_j b_j| = |a_j| |b_j| \le B |a_j| < B (\epsilon/B) = \epsilon$$

for every  $j \ge L$ , as desired.

#### 3.3.3 A lemma about constant multiples

**Lemma 3.3.10** Let  $\{z_j\}_{j=1}^{\infty}$  be a sequence of complex numbers that converges to a complex number z. If  $c \in \mathbf{C}$ , then  $\{c z_j\}_{j=1}^{\infty}$  converges to c z in  $\mathbf{C}$ .

Equivalently, this says that

(3.3.11) 
$$\lim_{j \to \infty} (c \, z_j) = c \left( \lim_{j \to \infty} z_j \right),$$

where the limit on the right exists by hypothesis, and the existence of the limit on the left is part of the conclusion.

Of course, if c = 0, then  $c z_j = 0$  for every  $j \ge 1$ , c z = 0, and the lemma is trivial. Thus we may suppose that  $c \ne 0$ . Let  $\epsilon > 0$  be given. Because  $\{z_j\}_{j=1}^{\infty}$  converges to z in  $\mathbf{C}$ , there is a positive integer L such that

$$(3.3.12) \qquad \qquad |z_j - z| < \epsilon/|c|$$

for every  $j \ge L$ . This implies that

(3.3.13) 
$$|c z_j - c z| = |c| |z_j - z| < |c| (\epsilon/|c|) = \epsilon$$

for every  $j \ge L$ , as desired.

#### **3.3.4** Products of convergent sequences

**Proposition 3.3.14** If  $\{z_j\}_{j=1}^{\infty}$  and  $\{w_j\}_{j=1}^{\infty}$  are sequences of complex numbers that converge to  $z, w \in \mathbf{C}$ , respectively, then  $\{z_j w_j\}_{j=1}^{\infty}$  converges to z w in  $\mathbf{C}$ .

As before, this can be summarized by saying that

(3.3.15) 
$$\lim_{j \to \infty} (z_j w_j) = \left(\lim_{j \to \infty} z_j\right) \left(\lim_{j \to \infty} w_j\right),$$

where the limits on the right exist by hypothesis, and the existence of the limit on the left is part of the conclusion.

Observe that

(3.3.16)  $z_j w_j = z_j (w_j - w) + z_j w$ 

for every  $j \geq 1$ . The hypothesis that  $\{w_j\}_{j=1}^{\infty}$  converges to w in  $\mathbf{C}$  is equivalent to saying that  $\{w_j - w\}_{j=1}^{\infty}$  converges to 0. We also have that  $\{z_j\}_{j=1}^{\infty}$  is a bounded sequence in  $\mathbf{C}$ , because it converges, as in Proposition 3.2.10. It follows that  $\{z_j (w_j - w)\}_{j=1}^{\infty}$  converges to 0 in  $\mathbf{C}$ , as in Lemma 3.3.6. Using Lemma 3.3.10, we get that  $\{z_j w\}_{j=1}^{\infty}$  converges to z w in  $\mathbf{C}$ . Thus the right side of (3.3.16) is the sum of two convergent sequences in  $\mathbf{C}$ . This implies that  $\{z_j w_j\}_{j=1}^{\infty}$  converges to z w, by Proposition 3.3.1, as desired.

#### 3.3.5 Sequences of reciprocals

**Proposition 3.3.17** If  $\{z_j\}_{j=1}^{\infty}$  is a sequence of nonzero complex numbers that converges to a nonzero complex number z, then  $\{1/z_j\}_{j=1}^{\infty}$  converges to 1/z in **C**.

This means that

(3.3.18) 
$$\lim_{j \to \infty} (1/z_j) = 1/\Big(\lim_{j \to \infty} z_j\Big),$$

where the existence of the limit on the right is part of the hypothesis, and the existence of the limit on the left is part of the conclusion.

Let us begin by observing that

(3.3.19) 
$$1/z_j - 1/z = (z - z_j)/(z_j z)$$

for every  $j \ge 1$ . Thus

$$(3.3.20) |1/z_j - 1/z| = |z - z_j|/(|z_j||z|)$$

for every  $j \ge 1$ . We should first deal with the factor of  $|z_j|$  in the denominator. Because  $\{z_j\}_{j=1}^{\infty}$  converges to z and  $z \ne 0$ , there is a positive integer  $L_0$  such that

$$(3.3.21) |z_j - z| < |z|/2$$

for every  $j \geq L_0$ . Using the triangle inequality, we get that

$$(3.3.22) |z| \le |z_j| + |z - z_j| < |z_j| + |z|/2$$

when  $j \ge L_0$ . This implies that

$$(3.3.23) |z|/2 < |z_j|$$

for every  $j \ge L_0$ . Combining this with (3.3.20), we get that

(3.3.24) 
$$|1/z_j - 1/z| \le (2/|z|^2) |z - z_j|$$

for every  $j \geq L_0$ .

Let  $\epsilon > 0$  be given. Because  $\{z_j\}_{j=1}^{\infty}$  converges to z, there is a positive integer L such that

(3.3.25) 
$$|z_j - z| < (|z|^2/2) \epsilon$$

for every  $j \ge L$ . If  $j \ge \max(L_0, L)$ , then it follows that

(3.3.26) 
$$|1/z_j - 1/z| < (2/|z|^2) (|z|^2/2) \epsilon = \epsilon,$$

as desired.

# 3.4 Subsequences and sequential compactness

**Definition 3.4.1** Let  $\{x_j\}_{j=1}^{\infty}$  be a sequence of elements of some set, and let  $\{j_l\}_{l=1}^{\infty}$  be a strictly increasing sequence of positive integers, so that  $j_l < j_{l+1}$  for every  $l \ge 1$ . Under these conditions,  $\{x_{j_l}\}_{l=1}^{\infty}$  is called a subsequence of  $\{x_j\}_{j=1}^{\infty}$ .

Note that a sequence may be considered as a subsequence of itself.

Let (M, d(x, y)) be a metric space. If  $\{x_j\}_{j=1}^{\infty}$  is a sequence of elements of M that converges to an element x of M, and if  $\{x_{j_l}\}_{l=1}^{\infty}$  is a subsequence of  $\{x_j\}_{j=1}^{\infty}$ , then it is easy to see that  $\{x_{j_l}\}_{l=1}^{\infty}$  converges to x in M as well. This uses the fact that  $j_l \ge l$  for every  $l \ge 1$ .

#### 3.4.1 Sequential compactness

**Definition 3.4.2** A subset K of M is said to be sequentially compact if for every sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of K there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  and an element x of K such that  $\{x_{j_l}\}_{l=1}^{\infty}$  converges to x in M.

**Proposition 3.4.3** A subset K of M is sequentially compact if and only if K has the limit point property in M.

## 3.4.2 The limit point property implies sequential compactness

Suppose first that K has the limit point property, and let us show that K is sequentially compact. Let  $\{x_j\}_{j=1}^{\infty}$  be any sequence of elements in K, and let  $L = \{x_j : j \in \mathbb{Z}_+\}$  be the set of terms in this sequence. If L has only finitely

many elements, then there is an  $x \in K$  such that  $x_j = x$  for infinitely many  $j \in \mathbf{Z}_+$ . Equivalently, this means that there is subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  such that  $x_{j_l} = x$  for every  $l \ge 1$ . Of course,  $\{x_{j_l}\}_{l=1}^{\infty}$  converges to x in this case.

Suppose now that L has infinitely many elements. If K has the limit point property, then there is a point  $x \in K$  that is a limit point of L in M. This implies that for every r > 0, there are infinitely many elements of L in B(x, r), as in Proposition 1.5.2. It follows that for each r > 0,

$$(3.4.4) d(x, x_j) < r$$

for infinitely many  $j \in \mathbf{Z}_+$ .

Using this, we can find a subsequence  $\{x_{j_n}\}_{n=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  such that

(3.4.5) 
$$d(x, x_{i_n}) < 1/n$$

for every  $n \ge 1$ . More precisely, we can first choose  $j_1 \in \mathbf{Z}_+$  so that (3.4.5) holds with n = 1. If  $j_n \in \mathbf{Z}_+$  has been chosen for some positive integer n, then we can choose  $j_{n+1} \in \mathbf{Z}_+$  so that  $j_{n+1} > j_n$  and (3.4.5) holds with n replaced with n + 1. It is easy to see that  $\{x_{j_n}\}_{n=1}^{\infty}$  converges to x in M, because of (3.4.5).

#### 3.4.3 Sequential compactness implies the limit point property

Conversely, suppose that  $K \subseteq M$  is sequentially compact, and let us show that K has the limit point property. Let L be an infinite subset of K, and let  $\{x_j\}_{j=1}^{\infty}$  be an infinite sequence of distinct elements of L. If K is sequentially compact, then there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  that converges to an element x of K in M. Let r > 0 be given, so that  $d(x_{j_l}, x) < r$  for all but finitely many  $l \in \mathbb{Z}_+$ . This implies that there are infinitely many elements of L in B(x, r), so that x is a limit point of L in M, as desired.

## **3.5** Cauchy sequences and completeness

Let (M, d(x, y)) be a metric space.

#### 3.5.1 Cauchy sequences

**Definition 3.5.1** A sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of M is said to be a Cauchy sequence if for every  $\epsilon > 0$  there is a positive integer L such that

$$(3.5.2) d(x_j, x_l) < \epsilon$$

for every  $j, l \geq L$ .

#### 3.5.2 Convergent sequences are Cauchy sequences

**Proposition 3.5.3** If a sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of M converges to an element x of M, then  $\{x_j\}_{j=1}^{\infty}$  is a Cauchy sequence in M.

Let  $\epsilon > 0$  be given. Because  $\{x_j\}_{j=1}^{\infty}$  converges to x in M, there is a positive integer L such that (3.5.4)  $d(x_i, x) < \epsilon/2$ 

for every  $j \ge L$ . If  $j, l \ge L$ , then it follows that

(3.5.5) 
$$d(x_j, x_l) \le d(x_j, x) + d(x, x_l) < \epsilon/2 + \epsilon/2 = \epsilon,$$

as desired.

#### 3.5.3 Cauchy sequences are bounded

**Proposition 3.5.6** If  $\{x_j\}_{j=1}^{\infty}$  is a Cauchy sequence in M, then  $\{x_j\}_{j=1}^{\infty}$  is bounded in M.

By hypothesis, there is a positive integer L such that

$$(3.5.7) d(x_j, x_l) < 1$$

for every  $j, l \ge L$ . If L = 1, then we get that  $x_j \in B(x_1, 1)$  for every  $j \ge 1$ , as desired. Otherwise, put

(3.5.8) 
$$r = \max\{d(x_j, x_L) : j = 1, \dots, L-1\}$$

which is a nonnegative real number. In this case, we have that  $x_j \in B(x_L, r+1)$  for every  $j \ge 1$ , as desired.

#### 3.5.4 Cauchy sequences with convergent subsequences

**Proposition 3.5.9** Let  $\{x_j\}_{j=1}^{\infty}$  be a Cauchy sequence of elements of M. If there is a subsequence  $\{x_{j_n}\}_{n=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  that converges to an element x of M, then  $\{x_j\}_{j=1}^{\infty}$  converges to x in M.

Let  $\epsilon > 0$  be given. Because  $\{x_j\}_{j=1}^{\infty}$  is a Cauchy sequence in M, there is a positive integer L such that

$$(3.5.10) d(x_j, x_l) < \epsilon/2$$

for every  $j, l \geq L$ . Similarly, there is a positive integer N such that

$$(3.5.11) d(x_{j_n}, x) < \epsilon/2$$

for every  $n \ge N$ , because  $\{x_{j_n}\}_{n=1}^{\infty}$  converges to x in M. If  $l \ge L$ ,  $n \ge N$ , and  $j_n \ge L$ , then we get that

(3.5.12) 
$$d(x_l, x) \le d(x_l, x_{j_n}) + d(x_{j_n}, x) < \epsilon/2 + \epsilon/2 = \epsilon.$$

As before,  $j_n \ge n$  for every positive integer n, so that  $j_n \ge L$  when  $n \ge L$ . In particular, if  $n = \max(L, N)$ , then  $j_n \ge L$  and  $n \ge N$ . Using this, we obtain that

$$(3.5.13)$$
  $d(x_l, x) <$ 

for every  $l \ge L$ , as desired.

**Corollary 3.5.14** If  $K \subseteq M$  is sequentially compact,  $\{x_j\}_{j=1}^{\infty}$  is a Cauchy sequence in M, and  $x_j \in K$  for every  $j \geq 1$ , then  $\{x_j\}_{j=1}^{\infty}$  converges to an element of K in M.

#### 3.5.5 Complete metric spaces

**Definition 3.5.15** If every Cauchy sequence of elements of M converges to an element of M, then M is said to be complete as a metric space.

**Corollary 3.5.16** If n is a positive integer, then  $\mathbb{R}^n$  is complete with respect to the standard Euclidean metric.

Let  $\{x_j\}_{j=1}^{\infty}$  be a Cauchy sequence of elements of  $\mathbf{R}^n$ . It is easy to see that there is a cell C in  $\mathbf{R}^n$  such that  $x_j \in C$  for every  $j \geq 1$ , because  $\{x_j\}_{j=1}^{\infty}$  is bounded in  $\mathbf{R}^n$ , as in Proposition 3.5.6. Remember that C is compact in  $\mathbf{R}^n$ , so that C has the limit point property and is sequentially compact, by Proposition 1.7.10, Theorem 1.14.1, and Proposition 3.4.3. Thus  $\{x_j\}_{j=1}^{\infty}$  converges to an element of C, as in Corollary 3.5.14.

## 3.6 Some particular sequences

In this section, we consider the convergence of some particular sequences of real or complex numbers, with respect to the standard metrics on  $\mathbf{R}$  and  $\mathbf{C}$ .

**3.6.1**  $\{1/j^p\}_{j=1}^{\infty}$ 

Let p be a positive real number, so that  $a^p$  can be defined as a positive real number for every positive real number a, as in Section 1.3. It is well known that

$$\lim_{j \to \infty} 1/j^p = 0.$$

To see this, let  $\epsilon > 0$  be given. Observe that

$$(3.6.2) 1/j^p < \epsilon$$

if and only if (3.6.3)  $j > (1/\epsilon)^{1/p}$ .

This holds for all but finitely many positive integers j, by the archimedean property of the real numbers.

This is all a bit simpler when p = 1/k for some positive integer k. If p is any positive real number, then there is a positive integer k such that  $1/k \le p$ , by the archimedean property, which can be used to reduce to that case.

 $\{a^j\}_{j=1}^\infty$ 3.6.2

If a is a nonnegative real number with a < 1, then

$$\lim_{j \to \infty} a^j = 0.$$

This is trivial when a = 0, and so we may as well suppose that a > 0.

Put b = 1/a - 1 > 0. It is easy to see that

$$(3.6.5) (1+b)^{j} \ge 1+b_{j}$$

for every positive integer j, using induction, for instance. This implies that

(3.6.6) 
$$a^j = (1+b)^{-j} \le 1/(1+bj)$$

for every  $j \ge 1$ , and (3.6.4) follows easily from this.

# **3.6.3** $\{j^{\alpha} a^{j}\}_{j=1}^{\infty}$

Let  $\alpha$  be a positive real number, and let us show that

(3.6.7) 
$$\lim_{j \to \infty} j^{\alpha} a^j = 0.$$

This follows from (3.6.6) when a > 0 and  $\alpha < 1$ .

Otherwise, the archimedean property for the real numbers implies that there is a positive integer k such that  $\alpha < k$ . Note that  $a^{1/k} < 1$ , so that

$$\lim_{j \to \infty} j^{\alpha/k} a^{j/k} = 0$$

as before. This implies that

(3.6.9) 
$$j^{\alpha} a^{j} = (j^{\alpha/k} a^{j/k})^{k} \to 0 \quad \text{as } j \to \infty,$$

as desired.

**3.6.4** 
$$\{p^{1/j}\}_{j=1}^{\infty}$$

If p is a positive real number, then

(3.6.10) 
$$\lim_{j \to \infty} p^{1/j} = 1.$$

Suppose first that  $p \ge 1$ , so that  $p^{1/j} \ge 1$  for every  $j \in \mathbf{Z}_+$ . Let  $\epsilon > 0$  be given, and observe that  $p^{1/j} < 1 + \epsilon$ 

(3.6.11)

if and only if 
$$(3.6.12) \qquad \qquad p < (1+\epsilon)^j.$$

This holds for all but finitely many  $j \in \mathbf{Z}_+$ , by (3.6.5), with  $b = \epsilon$ . Thus (3.6.10) holds when  $p \ge 1$ .

If 0 , then we can apply the previous argument to <math>1/p, to get that 1/i

(3.6.13) 
$$\lim_{j \to \infty} 1/p^{1/j} = 1$$

This implies (3.6.10), by Proposition 3.3.17.

48

**3.6.5**  $\{j^{1/j}\}_{j=1}^{\infty}$ In fact, (3.6.14)  $\lim_{j \to \infty} j^{1/j} = 1.$ 

Let  $\epsilon > 0$  be given, and note that  $j^{1/j} \ge 1$  for every  $j \in \mathbf{Z}_+$ . As before,

(3.6.15) 
$$j^{1/j} < 1 + \epsilon$$

if and only if (3.6.16)

which is the same as saying that

(3.6.17) 
$$j (1+\epsilon)^{-j} < 1.$$

The left side of (3.6.17) tends to 0 as  $j \to \infty$ , as in (3.6.7). This implies that (3.6.17) holds for all but finitely many  $j \in \mathbb{Z}_+$ , as desired.

 $j < (1+\epsilon)^j,$ 

If z is a complex number such that |z| < 1, then

$$\lim_{j \to \infty} z^j = 0.$$

More precisely,

 $(3.6.19) |z^j| = |z|^j \to 0 \quad \text{as } j \to \infty,$ 

by (3.6.4).

# 3.7 Infinite series

**Definition 3.7.1** Let  $a_1, a_2, a_3, \ldots$  be an infinite sequence of complex numbers. The infinite series  $\sum_{j=1}^{\infty} a_j$  is said to converge if the corresponding sequence of partial sums

$$(3.7.2) s_n = \sum_{j=1}^n a_j$$

converges as a sequence of complex numbers, with respect to the standard metric on  $\mathbf{C}$ . In this case, we put

(3.7.3) 
$$\sum_{j=1}^{\infty} a_j = \lim_{n \to \infty} s_n.$$

Sometimes we may wish to consider infinite series  $\sum_{j=0}^{\infty} a_j$  starting at j = 0, or some other integer.

#### 3.7.1 Sums and constant multiples

**Proposition 3.7.4** Let  $\sum_{j=1}^{\infty} a_j$  and  $\sum_{j=1}^{\infty} b_j$  be convergent series of complex numbers.

(a) Under these conditions,  $\sum_{j=1}^{\infty} (a_j + b_j)$  converges too, with

(3.7.5) 
$$\sum_{j=1}^{\infty} (a_j + b_j) = \sum_{j=1}^{\infty} a_j + \sum_{j=1}^{\infty} b_j.$$

(b) If c is a complex number, then  $\sum_{j=1}^{\infty} c a_j$  converges, with

(3.7.6) 
$$\sum_{j=1}^{\infty} c \, a_j = c \, \sum_{j=1}^{\infty} a_j.$$

If n is any positive integer, then

(3.7.7) 
$$\sum_{j=1}^{n} (a_j + b_j) = \sum_{j=1}^{n} a_j + \sum_{j=1}^{n} b_j$$

and

(3.7.8) 
$$\sum_{j=1}^{n} c \, a_j = c \, \sum_{j=1}^{n} a_j.$$

Thus Proposition 3.7.4 follows from the analogous statements for sequences of complex numbers.

#### 3.7.2 The Cauchy criterion

**Proposition 3.7.9** An infinite series  $\sum_{j=1}^{\infty} a_j$  of complex numbers converges if and only if the following condition holds: for every  $\epsilon > 0$  there is a positive integer L such that

$$(3.7.10) \qquad \qquad \left|\sum_{j=l}^{n} a_{j}\right| < \epsilon$$

for all positive integers l, n with  $n \ge l \ge L$ .

One can check that this condition is equivalent to saying that the corresponding sequence of partial sums (3.7.2) is a Cauchy sequence with respect to the standard metric on **C**. Thus the proposition follows from the completeness of **C** with respect to the standard metric, which is the same as the completeness of  $\mathbf{R}^2$  with respect to the standard Euclidean metric.

**Corollary 3.7.11** If  $\sum_{j=1}^{\infty} a_j$  is a convergent series of complex numbers, then  $\{a_j\}_{j=1}^{\infty}$  converges to 0 as a sequence of complex numbers.

This follows by taking l = n in (3.7.10).

#### 3.7.3 Absolute convergence

**Definition 3.7.12** An infinite series  $\sum_{j=1}^{\infty} a_j$  of complex numbers is said to converge absolutely if  $\sum_{j=1}^{\infty} |a_j|$  converges as an infinite series of nonnegative real numbers.

#### 3.7.4 Absolute convergence implies convergence

**Proposition 3.7.13** If  $\sum_{j=1}^{\infty} a_j$  is an absolutely convergent series of complex numbers, then  $\sum_{j=1}^{\infty} a_j$  converges.

Observe that

$$(3.7.14) \qquad \left|\sum_{j=l}^{n} a_{j}\right| \leq \sum_{j=l}^{n} |a_{j}|$$

for all positive integers l, n with  $l \leq n$ , by the triangle inequality. One can use this to obtain Proposition 3.7.13 from Proposition 3.7.9.

#### 3.7.5 Infinite series with nonnegative terms

**Proposition 3.7.15** Let  $\sum_{j=1}^{\infty} a_j$  be an infinite series of nonnegative real numbers. Under these conditions,  $\sum_{j=1}^{\infty} a_j$  converges if and only if the corresponding sequence of partial sums (3.7.2) is bounded.

If  $\sum_{j=1}^{\infty} a_j$  is a convergent series of complex numbers, then the corresponding sequence of partial sums is bounded, as in Proposition 3.2.10. If  $a_j$  is a nonnegative real number for each  $j \geq 1$ , then it is easy to see that the partial sums are monotonically increasing. In this case, the boundedness of the partial sums implies that they converge as a sequence of real numbers, as in Proposition 3.2.14.

#### 3.7.6 The comparison test

**Proposition 3.7.16 (Comparison test)** Let  $\sum_{j=1}^{\infty} a_j$  be an infinite series of complex numbers, and let  $\sum_{j=1}^{\infty} b_j$  be an infinite series of nonnegative real numbers. If  $|a_j| \leq b_j$  for every  $j \geq 1$  and  $\sum_{j=1}^{\infty} b_j$  converges, then  $\sum_{j=1}^{\infty} a_j$  converges absolutely.

Observe that

(3.7.17) 
$$\sum_{j=1}^{n} |a_j| \le \sum_{j=1}^{n} b_j$$

for every positive integer *n*. If  $\sum_{j=1}^{\infty} b_j$  converges, then its sequence of partial sums is bounded. This implies that the partial sums of  $\sum_{j=1}^{\infty} |a_j|$  are bounded as well, so that this series converges too.

#### 3.7.7 Geometric series

Let z be a complex number, and let us consider the corresponding geometric series  $\sum_{j=0}^{\infty} z^j$ . Here  $z^j$  is interpreted as being equal to 1 when j = 0.

If n is a nonnegative integer, then

$$(3.7.18) \quad (1-z) \sum_{j=0}^{n} z^{j} = \sum_{j=0}^{n} z^{j} - \sum_{j=0}^{n} z^{j+1} = \sum_{j=0}^{n} z^{j} - \sum_{j=1}^{n+1} z^{j} = 1 - z^{n+1}.$$

This implies that

(3.7.19) 
$$\sum_{j=0}^{n} z^{j} = (1 - z^{n+1})/(1 - z).$$

when  $z \neq 1$ .

If |z| < 1, then we get that  $\sum_{j=0}^{\infty} z^j$  converges, with

(3.7.20) 
$$\sum_{j=0}^{\infty} z^j = 1/(1-z),$$

by (3.6.18). Note that  $\sum_{j=0}^{\infty} z^j$  converges absolutely in this case, by the same argument for |z|.

If  $|z| \geq 1$ , then

$$(3.7.21) |z^j| = |z|^j \ge 1$$

for every  $j \ge 0$ . This means that  $\{z^j\}_{j=0}^{\infty}$  does not converge to 0, so that  $\sum_{j=0}^{\infty} z^j$  does not converge.

# 3.8 Power series

Let  $a_0, a_1, a_2, a_3, \ldots$  be an infinite sequence of complex numbers, and consider the corresponding *power series*  $\sum_{j=0}^{\infty} a_j z^j$ , for  $z \in \mathbf{C}$ .

#### 3.8.1 An absolute convergence property

**Proposition 3.8.1** If  $\sum_{j=0}^{\infty} a_j z^j$  converges absolutely for some  $z \in \mathbf{C}$ , then  $\sum_{j=0}^{\infty} a_j w^j$  converges absolutely for every  $w \in \mathbf{C}$  with  $|w| \leq |z|$ .

Indeed, if  $|w| \leq |z|$ , then

(3.8.2) 
$$|a_j w_j| = |a_j| |w|^j \le |a_j| |z|^j = |a_j z^j|$$

for every nonnegative integer j. Thus Proposition 3.8.1 follows from the comparison test.

#### Another absolute convergence property 3.8.2

**Proposition 3.8.3** If  $\sum_{j=0}^{\infty} a_j z^j$  converges for some nonzero complex number z, then  $\sum_{i=0}^{\infty} a_j w^i$  converges absolutely for every  $w \in \mathbf{C}$  with |w| < |z|.

If  $\sum_{j=0}^{\infty} a_j z^j$  converges, then  $\{a_j z^j\}_{j=0}^{\infty}$  converges to 0, by Corollary 3.7.11. This implies that  $\{a_j z^j\}_{j=0}^{\infty}$  is a bounded sequence in **C**, as in Proposition 3.2.10. It follows that there is a nonnegative real number C such that

$$(3.8.4) |a_j z^j| \le 0$$

for every  $j \ge 0$ .

Let  $w \in \mathbf{C}$  be given, and observe that

(3.8.5) 
$$|a_j w^j| = |a_j z^j| \left( |w|^j / |z|^j \right) \le C \left( |w| / |z| \right)^j$$

for every  $j \ge 0$ , by (3.8.4). If |w| < |z|, then  $\sum_{j=0}^{\infty} (|w|/|z|)^j$  is a convergent geometric series, as in the previous section. Of course, this implies that  $\sum_{j=0}^{\infty} C(|w|/|z|)^j$  converges too, as in Proposition 3.7.4. Proposition 3.8.3 now follows from the comparison test.

#### The radius of convergence 3.8.3

Let E be the set of complex numbers z such that  $\sum_{j=0}^{\infty} a_j z^j$  does not converge, and put

$$(3.8.6) E_1 = \{ |z| : z \in E \}.$$

This is a set of nonnegative real numbers, which is not empty when  $E \neq \emptyset$ .

**Definition 3.8.7** The radius of convergence  $\rho$  of  $\sum_{j=0}^{\infty} a_j z^j$  is defined by

$$(3.8.8) \qquad \qquad \rho = \inf E_1$$

when  $E \neq \emptyset$ , and by  $\rho = +\infty$  when  $E = \emptyset$ .

#### 3.8.4A characterization of the radius of convergence

**Proposition 3.8.9** (a) If  $w \in \mathbf{C}$  satisfies  $|w| > \rho$ , then  $\sum_{j=0}^{\infty} a_j w^j$  does not converge.

(b) If  $w \in \mathbf{C}$  satisfies  $|w| < \rho$ , then  $\sum_{j=0}^{\infty} a_j w^j$  converges absolutely. (c) The radius of convergence of  $\sum_{j=0}^{\infty} a_j z^j$  is uniquely determined by the properties in (a) and (b).

To prove (a), let w be a complex number such that  $\sum_{j=0}^{\infty} a_j w^j$  converges. If  $z \in \mathbf{C}$  satisfies |z| < |w|, then  $\sum_{j=0}^{\infty} a_j z^j$  converges absolutely, by Proposition 3.8.3. This implies that |w| is a lower bound for  $E_1$ , so that  $|w| \le \inf E_1 = \rho$ when  $E \neq \emptyset$ . If  $E = \emptyset$ , then  $|w| < \rho = +\infty$  holds trivially.

If  $\sum_{j=0}^{\infty} a_j w^j$  does not converge absolutely, then  $\sum_{j=0}^{\infty} a_j z^j$  does not converge for any  $z \in \mathbf{C}$  with |w| < |z|, by Proposition 3.8.3 again. This implies that  $E \neq \emptyset$ , and that  $\rho = \inf E_1 \leq |w|$ .

If  $0 \leq \rho \leq +\infty$  satisfies the conditions in (a) and (b), then  $E = \emptyset$  when  $\rho = +\infty$ , and  $E \neq \emptyset$ , inf  $E_1 = \rho$  when  $\rho < \infty$ . Alternatively, if  $0 \leq \rho', \rho'' \leq +\infty$  both satisfy both (a) and (b) in place of  $\rho$ , then one can check directly that  $\rho' = \rho''$ .

## 3.9 Cauchy's condensation test

**Theorem 3.9.1 (Cauchy's condensation test)** Let  $\{a_j\}_{j=1}^{\infty}$  be a monotonically decreasing sequence of nonnegative real numbers. Under these conditions,  $\sum_{j=1}^{\infty} a_j$  converges if and only if  $\sum_{k=0}^{\infty} 2^k a_{2^k}$  converges.

# 3.9.1 The "if" part

Suppose first that  $\sum_{k=0}^{\infty} 2^k a_{2^k}$  converges, so that the corresponding sequence of partial sums  $\sum_{k=0}^{n} 2^k a_{2^k}$  are bounded. We would like to show that the partial sums for  $\sum_{j=1}^{\infty} a_j$  are bounded, which would imply that the series converges.

If k is any nonnegative integer, then

(3.9.2) 
$$\sum_{j=2^{k}}^{2^{k+1}-1} a_{j} \leq \sum_{j=2^{k}}^{2^{k+1}-1} a_{2^{k}} = 2^{k} a_{2^{k}}.$$

If l is a positive integer, n is a nonnegative integer, and  $l < 2^{n+1}$ , then we get that

(3.9.3) 
$$\sum_{j=1}^{l} a_j \le \sum_{k=0}^{n} \sum_{j=2^k}^{2^{k+1}-1} a_j \le \sum_{k=0}^{n} 2^k a_{2^k}.$$

Thus the boundedness of the partial sums of  $\sum_{k=0}^{\infty} 2^k a_{2^k}$  implies the boundedness of the partial sums for  $\sum_{j=1}^{\infty} a_j$ , as desired.

### 3.9.2 The "only if" part

Similarly, if k is a positive integer, then

(3.9.4) 
$$2^{k-1} a_{2^k} \le \sum_{j=2^{k-1}+1}^{2^k} a_j.$$

If n is a positive integer, then we obtain that

$$(3.9.5) \quad \sum_{k=0}^{n} 2^{k} a_{2^{k}} = a_{1} + \sum_{k=1}^{n} 2^{k} a_{2^{k}} \le a_{1} + 2 \sum_{k=1}^{n} \sum_{j=2^{k-1}+1}^{2^{k}} a_{j} \le 2 \sum_{j=1}^{2^{n}} a_{j}.$$

It follows that the partial sums of  $\sum_{k=0}^{\infty} 2^k a_{2^k}$  are bounded when the partial sums of  $\sum_{j=1}^{\infty} a_j$  are bounded, as desired.

#### 3.9.3 Some examples of series of this type

Let p be a positive real number, and note that  $a_j = 1/j^p$  defines a monotonically descreasing sequence of positive real numbers. In this situation,

(3.9.6) 
$$2^k a_{2^k} = 2^{(1-p)k} = (2^{1-p})^k$$

Thus the Cauchy condensation test implies that  $\sum_{j=1}^{\infty} 1/j^p$  converges exactly when

(3.9.7) 
$$\sum_{k=0}^{\infty} (2^{1-p})^k$$

converges. The latter is a geometric series that converges exactly when

$$(3.9.8) 2^{1-p} < 1.$$

which holds if and only if p > 1.

# **3.10** Another criterion for convergence

**Theorem 3.10.1** Let  $\{a_j\}_{j=0}^{\infty}$  be a sequence of complex numbers, and suppose that the sequence of sums

(3.10.2) 
$$A_n = \sum_{j=0}^n a_j$$

is bounded. Also let  $\{b_j\}_{j=0}^{\infty}$  be a monotonically decreasing sequence of nonnegative real numbers that converges to 0. Under these conditions,  $\sum_{j=0}^{\infty} a_j b_j$  converges.

Put  $A_{-1} = 0$ , so that  $a_j = A_j - A_{j-1}$  for every nonnegative integer j. Let n be a nonnegative integer, and observe that

(3.10.3) 
$$\sum_{j=0}^{n} a_j b_j = \sum_{j=0}^{n} (A_j - A_{j-1}) b_j = \sum_{j=0}^{n} A_j b_j - \sum_{j=0}^{n} A_{j-1} b_j.$$

The last sum on the right can be reexpressed as

(3.10.4) 
$$\sum_{j=0}^{n} A_{j-1} b_j = \sum_{j=-1}^{n-1} A_j b_{j+1} = \sum_{j=0}^{n} A_j b_{j+1} - A_n b_{n+1}.$$

Thus

(3.10.5) 
$$\sum_{j=0}^{n} a_{j} b_{j} = \sum_{j=0}^{n} A_{j} b_{j} - \sum_{j=0}^{n} A_{j} b_{j+1} + A_{n} b_{n+1}$$
$$= \sum_{j=0}^{n} A_{j} (b_{j} - b_{j+1}) + A_{n} b_{n+1}.$$

Lemma 3.3.6 implies that  $\{A_n b_{n+1}\}_{n=0}^{\infty}$  converges to 0, because  $\{A_n\}_{n=0}^{\infty}$  is a bounded sequence, and  $\{b_{n+1}\}_{n=0}^{\infty}$  converges to 0.

#### 3.10.1A related series

It suffices to show that

(3.10.6) 
$$\sum_{j=0} A_j (b_j - b_{j+1})$$

converges, by the remarks in the preceding paragraph. If n is a nonnegative integer, then

nnnnn+1

 $\infty$ 

(3.10.7) 
$$\sum_{j=0}^{n} (b_j - b_{j+1}) = \sum_{j=0}^{n} b_j - \sum_{j=0}^{n} b_{j+1} = \sum_{j=0}^{n} b_j - \sum_{j=1}^{n+1} b_j = b_0 - b_{n+1}.$$

This implies that  $\sum_{j=0}^{\infty} (b_j - b_{j+1})$  converges, because  $\{b_{n+1}\}_{n=0}^{\infty}$  converges as a sequence of real numbers.

Because  $\{A_j\}_{j=0}^{\infty}$  is bounded, there is a nonnegative real number C such that

$$(3.10.8) |A_j| \le C$$

for every  $j \ge 0$ . It follows that

(3.10.9) 
$$|A_j (b_j - b_{j+1})| \le C (b_j - b_{j+1})$$

for every  $j \ge 0$ , because  $b_j - b_{j+1} \ge 0$ , by hypothesis. Of course,

(3.10.10) 
$$\sum_{j=0}^{\infty} C\left(b_j - b_{j+1}\right)$$

converges, because  $\sum_{j=0}^{\infty} (b_j - b_{j+1})$  converges. Thus the comparison test implies that (3.10.6) converges absolutely, as desired.

#### 3.10.2Leibniz' alternating series test

If  $a_j = (-1)^j$  for every nonnegative integer j, then  $A_n$  is equal to 1 when j is even, and to 0 when j is odd. In particular,  $\{A_n\}_{n=0}^{\infty}$  is bounded. In this case, Theorem 3.10.1 corresponds to Leibniz' alternating series test.

#### 3.10.3Some more complicated examples

Let z be a complex number, and suppose that

for every nonnegative integer j. If |z| < 1, then the comparison test implies that  $\sum_{j=0}^{\infty} b_j z^j$  converges absolutely when  $\{b_j\}_{j=0}^{\infty}$  is bounded. Suppose that  $z \neq 1$ , so that

(3.10.12) 
$$A_n = \sum_{j=0}^n z^j = (1 - z^{n+1})/(1 - z)$$

for every nonnegative integer n, as in (3.7.19). If |z| = 1, then

$$(3.10.13) |A_n| = |1 - z^{n+1}| / |1 - z| \le (1 + |z|^{n+1}) / |1 - z| = 2/|1 - z|$$

for every  $n \ge 0$ , so that  $\{A_n\}_{n=0}^{\infty}$  is a bounded sequence of complex numbers.

# 3.11 Extended real numbers

The set of *extended real numbers* is defined to be the set of real numbers together with two additional elements, denoted  $+\infty$  and  $-\infty$ . The standard ordering on the real line can be extended to the set of extended real numbers, by putting

$$(3.11.1) \qquad \qquad -\infty < x < +\infty$$

for every  $x \in \mathbf{R}$ .

If A is any set of extended real numbers, then the notions of upper and lower bounds for A in the set of extended real numbers can be defined in the usual way. In particular,  $+\infty$  is automatically an upper bound for A, and  $-\infty$  is automatically a lower bound for A.

## 3.11.1 Suprema and infima of sets of extended real numbers

The notions of supremum or least upper bound and infimum or greatest lower bound for A in the set of extended real numbers can be defined in the same way as before too. The supremum and infimum of A in the set of extended real numbers always exists, as follows.

If  $+\infty \in A$ , or if  $A \cap \mathbf{R}$  has no upper bound in  $\mathbf{R}$ , then  $\sup A = +\infty$ . If  $+\infty \notin A$ , and  $A \cap \mathbf{R}$  is nonempty and has an upper bound in  $\mathbf{R}$ , then  $\sup A$  is the same as the supremum of  $A \cap \mathbf{R}$  in  $\mathbf{R}$ . Otherwise, if  $A \subseteq \{-\infty\}$ , then  $\sup A = -\infty$ .

Similarly, if  $-\infty \in A$  or  $A \cap \mathbf{R}$  has no lower bound in  $\mathbf{R}$ , then  $\inf A = -\infty$ . If  $-\infty \notin A$  and  $A \cap \mathbf{R}$  is nonempty and has a lower bound in  $\mathbf{R}$ , then  $\inf A$  is the same as the infimum of  $A \cap \mathbf{R}$  in  $\mathbf{R}$ . If  $A \subseteq \{+\infty\}$ , then  $\inf A = +\infty$ .

Note that

$$(3.11.2) \qquad \qquad \inf A \le \sup A$$

when  $A \neq \emptyset$ .

### 3.11.2 Sums and products of extended real numbers

Sums and products of extended real numbers are defined in some situations, as follows. If  $x \in \mathbf{R}$ , then we put

$$(3.11.3) \quad x + (+\infty) = (+\infty) + x = +\infty, \quad x + (-\infty) = (-\infty) + x = -\infty.$$

We also put

$$(3.11.4) (+\infty) + (+\infty) = +\infty, \quad (-\infty) + (-\infty) = -\infty.$$

The sum of  $+\infty$  and  $-\infty$  is not defined.

If x is a nonzero extended real number, then we put

$$(3.11.5) x(+\infty) = (+\infty) x = +\infty, x(-\infty) = (-\infty) x = -\infty$$

when x > 0, and

(3.11.6) 
$$x(+\infty) = (+\infty)x = -\infty, \quad x(-\infty) = (-\infty)x = +\infty$$

when x < 0. The product of 0 with  $\pm \infty$  is not defined.

We put  $x/(+\infty) = x/(-\infty) = 0$  when  $x \in \mathbf{R}$ , and these quotients are not defined when  $x = \pm \infty$ . Although 1/0 is not defined, it may be appropriate to interpret it as being  $+\infty$ , when dealing with nonnegative extended real numbers.

#### 3.11.3 Sequences in R tending to $+\infty$ or $-\infty$

Let  $\{x_j\}_{j=1}^{\infty}$  be a sequence of real numbers.

**Definition 3.11.7** We say that  $x_j$  tends to  $+\infty$  as  $j \to \infty$ , or  $x_j \to +\infty$  as  $j \to \infty$ , if for every nonnegative real number R there is a positive integer L such that

$$(3.11.8) x_j > R for every j \ge L.$$

Similarly, we say that  $x_j$  tends to  $-\infty$  as  $j \to \infty$ , or  $x_j \to -\infty$  as  $j \to \infty$ , if for every nonnegative real number R there is a positive integer L such that

$$(3.11.9) x_j < -R for every j \ge L.$$

Thus, if x is any extended real number, then the condition that  $x_j \to x$  as  $j \to \infty$  is defined, using convergence with respect to the standard Euclidean metric on **R** when  $x \in \mathbf{R}$ , and the previous definition when  $x = \pm \infty$ . It is easy to see that this can hold for at most one x.

#### 3.11.4 Some properties of sequences of real numbers

**Proposition 3.11.10** Let  $\{x_j\}_{j=1}^{\infty}$ ,  $\{y_j\}_{j=1}^{\infty}$  be sequences of real numbers, let x, y be extended real numbers, and suppose that  $x_j \to x$  and  $y_j \to y$  as  $j \to \infty$ .

(a) If x + y is defined as an extended real number, then  $x_j + y_j \rightarrow x + y$  as  $j \rightarrow \infty$ .

(b) If x y is defined as an extended real number, then  $x_j y_j \to x y$  and  $j \to \infty$ .

Of course, this follows from previous results when  $x, y \in \mathbf{R}$ . It is not difficult to prove (a) and (b) directly in the other cases. Suppose that  $x = +\infty$ , for instance. If the set of  $y_j$ 's has a lower bound in  $\mathbf{R}$ , then it is easy to see that  $x_j + y_j \to \infty$  as  $j \to \infty$ . In particular, this holds when  $y_j \to y$  as  $j \to \infty$ 

58

and  $y \neq -\infty$ . Similarly, suppose that there is a positive real number a and a positive integer  $L_0$  such that

$$(3.11.11) y_j \ge a$$

for every  $j \ge L_0$ . Using this, one can check that  $x_j y_j \to +\infty$  as  $j \to \infty$ . If  $y_j \to y$  as  $j \to \infty$ , then this condition holds when y > 0.

If  $\{x_j\}_{j=1}^{\infty}$  is a sequence of nonzero real numbers such that  $|x_j| \to \infty$  as  $j \to \infty$ , then

$$(3.11.12) 1/x_j \to 0 \text{as } j \to \infty.$$

If  $\{x_j\}_{j=1}^{\infty}$  is a sequence of positive real numbers that converges to 0, then

$$(3.11.13) 1/x_j \to +\infty as j \to \infty.$$

#### 3.11.5 Subsequences of sequences in R

**Proposition 3.11.14** If  $\{x_j\}_{j=1}^{\infty}$  is any sequence of real numbers, then there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  and an extended real number x such that  $x_{j_l} \to x$  as  $l \to \infty$ .

Suppose first that there are real numbers a and b such that

$$(3.11.15) a \le x_j \le b \text{for every } j \ge 1.$$

Remember that [a, b] is a compact subset of the real line with respect to the standard Euclidean metric, which implies that [a, b] has the limit point property, and hence is sequentially compact, by previous results. This means that there is a subsequence of  $\{x_j\}_{j=1}^{\infty}$  that converges to an element of [a, b], as desired.

Next, suppose that the set of  $x_j$ 's has no upper bound in **R**. This implies that for each positive integer n,

(3.11.16) 
$$x_j > n$$
 for infinitely many  $j \ge 1$ ,

since otherwise the set of  $x_j$ 's would have a finite upper bound. We would like to find a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  such that

$$(3.11.17) x_{j_l} > l for every l \ge 1,$$

which implies that  $x_{j_l} \to \infty$  as  $j \to \infty$ . We can start by choosing  $j_1$  to be any positive integer such that  $x_{j_1} > 1$ . If  $j_l$  has been chosen for some positive integer l, then can choose  $j_{l+1}$  to be a positive integer such that  $j_{l+1} > j_l$  and  $x_{j_{l+1}} > l + 1$ , using (3.11.16).

Similarly, if the set of  $x_j$ 's does not have a lower bound in **R**, then one can find a subsequence that tends to  $-\infty$ . This can be shown in essentially the same way, or by reducing to the previous case.

# 3.12 Limits superior and inferior

Let  $\{x_j\}_{j=1}^{\infty}$  be a sequence of real numbers, and let E be the set of extended real numbers x for which there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  such that

$$(3.12.1) x_{j_l} \to x \text{ as } l \to \infty.$$

Note that  $E \neq \emptyset$ , by Proposition 3.11.14.

**Definition 3.12.2** The upper limit or limit superior of  $\{x_j\}_{j=1}^{\infty}$  is defined by

(3.12.3) 
$$\limsup_{j \to \infty} x_j = \sup E.$$

Similarly, the lower limit or limit inferior of  $\{x_j\}_{j=1}^{\infty}$  is defined by

(3.12.4) 
$$\liminf_{j \to \infty} x_j = \inf E.$$

Thus

(3.12.5) 
$$\liminf_{j \to \infty} x_j \le \limsup_{j \to \infty} x_j$$

If there is an extended real number x such that  $x_j \to x$  as  $j \to \infty$ , then

$$(3.12.6) E = \{x\},$$

(3.12.7) 
$$\limsup_{j \to \infty} x_j = \liminf_{j \to \infty} x_j = x.$$

## 3.12.1 A characterization of the upper limit of a sequence

**Proposition 3.12.8** Put  $y = \limsup_{j \to \infty} x_j$ .

(a) If  $z \in \mathbf{R}$  satisfies y < z, then  $x_j < z$  for all but finitely many  $j \ge 1$ .

(b) If  $w \in \mathbf{R}$  satisfies w < y, then  $x_j > w$  for infinitely many  $j \ge 1$ .

(c) There is only one extended real number y that satisfies the conditions in (a) and (b).

Suppose for the sake of a contradiction that  $z \in \mathbf{R}$ , y < z, and  $x_j \ge z$  for infinitely many  $j \ge 1$ . Equivalently, this means that there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  such that

$$(3.12.9) x_{j_l} \ge z \text{for every } l \ge 1.$$

Using Proposition 3.11.14, we can get a subsequence  $\{x_{jl_n}\}_{n=1}^{\infty}$  of  $\{x_{jl}\}_{l=1}^{\infty}$  and an extended real number x such that  $x_{jl_n} \to x$  as  $n \to \infty$ . It is easy to see that  $\{x_{jl_n}\}_{n=1}^{\infty}$  may also be considered as a subsequence of  $\{x_j\}_{j=1}^{\infty}$ , so that  $x \in E$ . One can also check that  $x \ge z$ , because  $x_{jl_n} \ge z$  for every  $n \ge 1$ , by (3.12.9). Thus x > y, because y < z, by hypothesis. This contradicts the fact that  $y = \sup E$ , as desired. If w < y, then w is not an upper bound for E, by definition of the supremum. This means that there is an  $x \in E$  such that w < x. In this case, there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  such that  $x_{j_l} \to x$  as  $l \to \infty$ . It is easy to see that  $x_{j_l} > w$  for all but finitely many  $l \ge 1$ , because w < x. In particular,  $x_j > w$  for infinitely many  $j \ge 1$ , as desired.

Suppose that for the sake of a contradiction that y' and y'' are distinct extended real numbers that satisfy the conditions in (a) and (b) in place of y. We may as well suppose that y' < y'', since otherwise we can interchange the roles of y' and y''. This implies that there is a real number r such that y' < r < y''. Using (a) for y' and z = r, we get that  $x_j < r$  for all but finitely many  $j \ge 1$ . However, part (b) for y'' and w = r implies that  $x_j > r$  for infinitely many  $j \ge 1$ , which is a contradiction.

#### 3.12.2 A characterization of the lower limit of a sequence

**Proposition 3.12.10** Put  $u = \liminf_{j \to \infty} x_j$ .

(a) If  $t \in \mathbf{R}$  satisfies t < u, then  $x_j > t$  for all but finitely many  $j \ge 1$ .

(b) If  $v \in \mathbf{R}$  satisfies u < v, then  $x_j < v$  for infinitely many  $j \ge 1$ .

(c) There is only one extended real number u that satisfies the conditions in (a) and (b).

This can be shown in essentially the same way as before. Alternatively, one can check that

(3.12.11) 
$$-\liminf_{j \to \infty} x_j = \limsup_{j \to \infty} (-x_j),$$

and use this to reduce to the previous proposition.

#### 3.12.3 Equality of the upper and lower limits

Corollary 3.12.12 Suppose that

(3.12.13) 
$$\limsup_{j \to \infty} x_j = \liminf_{j \to \infty} x_j,$$

and let x be their common value. Under these conditions,  $x_j \to x$  as  $j \to \infty$ .

This can be verified using part (a) of each of Propositions 3.12.8 and 3.12.10.

#### 3.12.4 Some variants of these definitions and results

The definitions of the upper and lower limits used here correspond to Definition 3.16 on p56 of [192]. The characterization of the upper limit in Proposition 3.12.8 is analogous to Theorem 3.17 on p56 of [192]. However, instead of part (b) of Proposition 3.12.8, part (a) of Theorem 3.17 in [192] states that

$$(3.12.14) y = \limsup_{j \to \infty} x_j \in E.$$

It is easy to see that this implies part (b) of Proposition 3.12.8. Conversely, it is not too difficult to show that (3.12.14) can be obtained from parts (a) and (b) of Proposition 3.12.8.

Although part (c) of Proposition 3.12.8 is stated in the same way as the last part of Theorem 3.17 in [192], its meaning is a bit different, because it uses part (b) of Proposition 3.12.8 instead of (3.12.14).

Sometimes upper and lower limits are defined or characterized in other ways, as in Definition 1.13 on p14 of [191]. In that definition, it is mentioned that the upper limit satisfies (3.12.14) and is the largest extended real number with this property, which means that the definition in [191] is equivalent to the one used in [192]. This could also be obtained from Proposition 3.12.8.

Upper and lower limits are also discussed in Section 2.9 beginning on p46 of [82], for instance, and the initial definitions are basically the same as in [191]. Theorem 2.9 L on p50 of [82] states that the upper and lower limits as defined there satisfy the conditions in parts (a) and (b) of Propositions 3.12.8 and 3.12.10, respectively, and the converse is mentioned afterwards. The fact that (3.12.14) holds is shown in the proof of Theorem 2.9 M on p51 of [82]. Exercises 2 and 3 on p51 of [82] basically show that the definitions of the upper and lower limits used there are equivalent to those used in [192] and here. Upper and lower limits of sequences of sets are discussed in Section 2.12 on p65 of [82].

#### 3.12.5 Upper limits of sums of sequences of real numbers

If  $\{a_j\}_{j=1}^{\infty}$  and  $\{b_j\}_{j=1}^{\infty}$  are sequences of real numbers, then it is well known that

(3.12.15) 
$$\limsup_{j \to \infty} (a_j + b_j) \le \limsup_{j \to \infty} a_j + \limsup_{j \to \infty} b_j,$$

as long as the right side is defined as an extended real number, which is to say that it is not a sum of  $+\infty$  and  $-\infty$ . This corresponds to part (b) of Exercise 4 on p32 of [191], and to Exercise 5 on p78 of [192].

### 3.13 Applications for infinite series

Proposition 3.13.1 (The root test) Let

$$(3.13.2) \qquad \qquad \sum_{j=1}^{\infty} a_j$$

be an infinite series of complex numbers, and put

(3.13.3) 
$$\alpha = \limsup_{j \to \infty} |a_j|^{1/j}.$$

The series (3.13.2) converges absolutely when  $\alpha < 1$ , and does not converge when  $\alpha > 1$ .

# 3.13.1 The proof of the root test, and some additional remarks

Suppose first that  $\alpha < 1$ , and let  $\beta$  be a real number such that  $\alpha < \beta < 1$ . Using part (a) of Proposition 3.12.8, we get that

(3.13.4) 
$$|a_i|^{1/j} < \beta$$

for all but finitely many  $j \ge 1$ . Equivalently, this means that

$$(3.13.5) |a_j| < \beta^j$$

for all but finitely many  $j \ge 1$ . It is easy to see that  $\sum_{j=1}^{\infty} |a_j|$  converges in this case, in comparison with the convergent geometric series  $\sum_{j=1}^{\infty} \beta^j$ .

Now suppose that  $\alpha > 1$ , and let  $\gamma$  be a real number such that  $1 \le \gamma < \alpha$ . Part (b) of Proposition 3.12.8 implies that

(3.13.6) 
$$|a_i|^{1/j} > \gamma$$

for infinitely many  $j \ge 1$ . This is the same as saying that

$$(3.13.7) |a_j| > \gamma$$

for infinitely many  $j \ge 1$ . If we take  $\gamma = 1$ , then we get that  $\{a_j\}_{j=1}^{\infty}$  does not converge to 0 as a sequence of complex numbers, so that  $\sum_{j=1}^{\infty} a_j$  does not converge. We can also take  $\gamma > 1$ , to obtain that  $\{a_j\}_{j=1}^{\infty}$  is not even bounded. If  $a_j = 1$  for every  $j \ge 1$ , then  $\alpha = 1$ , and  $\sum_{j=1}^{\infty} a_j$  does not converge. If

If  $a_j = 1$  for every  $j \ge 1$ , then  $\alpha = 1$ , and  $\sum_{j=1}^{\infty} a_j$  does not converge. If  $a_j = 1/j^2$  for every  $j \ge 1$ , then  $\sum_{j=1}^{\infty} a_j$  converges, and  $a_j^{1/j} = (j^{1/j})^{-2} \to 1$  as  $j \to \infty$ , so that  $\alpha = 1$ .

# 3.13.2 The root test and the radius of convergence of a power series

As an application of the root test, let

$$(3.13.8) \qquad \qquad \sum_{j=0}^{\infty} a_j \, z^j$$

be a power series with complex coefficients, which obviously converges absolutely when z = 0. Also let  $\alpha$  be as in (3.13.3) again, which can be defined without using the j = 0 term. If z is a nonzero complex number, then one can check that

(3.13.9) 
$$\limsup_{j \to \infty} |a_j z^j|^{1/j} = |z| \alpha.$$

The root test implies that (3.13.8) converges absolutely when  $|z|\alpha < 1$ , and that (3.13.8) does not converge when  $|z|\alpha > 1$ . It follows that

$$(3.13.10) \qquad \qquad \rho = 1/\alpha$$

is the radius of convergence of (3.13.8), which is interpreted as being  $+\infty$  when  $\alpha = 0$ .

#### 3.13.3 The ratio test

**Proposition 3.13.11 (The ratio test)** An infinite series  $\sum_{j=1}^{\infty} a_j$  of nonzero complex numbers converges absolutely when

(3.13.12) 
$$\limsup_{j \to \infty} (|a_{j+1}|/|a_j|) < 1,$$

and does not converge when

(3.13.13) 
$$\liminf_{j \to \infty} (|a_{j+1}|/|a_j|) > 1.$$

If (3.13.12) holds, then let  $\beta$  be a real number such that

(3.13.14) 
$$\limsup_{j \to \infty} (|a_{j+1}|/|a_j|) < \beta < 1.$$

Part (a) of Proposition 3.12.8 implies that there is a positive integer  $L_1$  such that

$$(3.13.15) |a_{j+1}|/|a_j| < \beta$$

for every  $j \geq L_1$ . This means that

$$(3.13.16) |a_{j+1}| < \beta |a_j|$$

for every  $j \ge L_1$ , so that

$$(3.13.17) |a_j| \le \beta^{j-L_1} |a_{L_1}|$$

for every  $j \ge L_1$ . Using this, the convergence of  $\sum_{j=1}^{\infty} |a_j|$  can be obtained from the convergence of the geometric series  $\sum_{j=1}^{\infty} \beta^j$ .

If (3.13.13) holds, then let  $\gamma$  be a real number such that

(3.13.18) 
$$\liminf_{j \to \infty} (|a_{j+1}|/|a_j|) > \gamma \ge 1.$$

Part (a) of Proposition 3.12.10 implies that there is a positive integer  $L_2$  such that

$$(3.13.19) |a_{j+1}|/|a_j| > \gamma$$

for every  $j \ge L_2$ . Thus (3.13.20)  $|a_{j+1}| > \gamma |a_j|$ 

for every  $j \ge L_2$ , and hence

$$(3.13.21) |a_j| \ge \gamma^{j-L_2} |a_{L_2}|$$

for every  $j \geq L_2$ . In particular, this implies that  $\{a_j\}_{j=1}^{\infty}$  does not converge to 0 when  $\gamma = 1$ , so that  $\sum_{j=1}^{\infty} a_j$  does not converge. If we take  $\gamma > 1$ , then we get that  $|a_j| \to \infty$  as  $j \to \infty$ .

#### **Rearrangements of infinite series** 3.14

Let  $\pi$  be a one-to-one mapping from the set of positive integers onto itself. Thus  $\pi(j) \in \mathbf{Z}_+$  for every  $j \in \mathbf{Z}_+$ , and every positive integer k can be expressed as  $\pi(j)$  for exactly one  $j \in \mathbf{Z}_+$ , which may be denoted  $\pi^{-1}(k)$ .

If  $\sum_{j=1}^{\infty} a_j$  is an infinite series of complex numbers, then

(3.14.1) 
$$\sum_{j=1}^{\infty} a_{\pi(j)}$$

is called a *rearrangement* of  $\sum_{j=1}^{\infty} a_j$ . If  $a_j = 0$  for all but finitely many  $j \ge 1$ , then it is easy to see that  $a_{\pi(j)} = 0$  for all but finitely many  $j \ge 1$  too, and that

(3.14.2) 
$$\sum_{j=1}^{\infty} a_{\pi(j)} = \sum_{j=1}^{\infty} a_j.$$

#### 3.14.1**Rearrangements with nonnegative terms**

1

Suppose for the moment that  $a_j$  is a nonnegative real number for each  $j \ge 1$ . If l, n are positive integers such that

(3.14.3) 
$$\max\{\pi(j) : 1 \le j \le l\} \le n,$$

then

(3.14.4) 
$$\sum_{j=1}^{l} a_{\pi(j)} \le \sum_{j=1}^{n} a_j.$$

Similarly, if (3.14.5)

$$\max\{\pi^{-1}(j): 1 \le j \le l\} \le n,$$

then

(3.14.6) 
$$\sum_{j=1}^{l} a_j \le \sum_{j=1}^{n} a_{\pi(j)}.$$

If  $\sum_{j=1}^{\infty} a_j$  converges, then one can use (3.14.4) and (3.14.6) to check that  $\sum_{j=1}^{\infty} a_{\pi(j)}$  converges, and that (3.14.2) holds.

#### 3.14.2 Rearrangements of absolutely convergent series

If  $\sum_{j=1}^{\infty} a_j$  is an absolutely convergent series of complex numbers, then it is easy to see that  $\sum_{j=1}^{\infty} a_{\pi(j)}$  is absolutely convergent as well, by applying the remarks in the preceding paragraph to  $\sum_{j=1}^{\infty} |a_j|$ . In order to show that (3.14.2) holds in this situation, suppose first that  $a_j$ 

is a real number for each  $j \ge 1$ . Put

(3.14.7) 
$$a_j^+ = \max(a_j, 0)$$

and

66

(3.14.8) 
$$a_j^- = -\min(a_j, 0) = \max(-a_j, 0)$$

for each  $j \ge 1$ . Note that  $a_j^+, a_j^- \ge 0$ ,

(3.14.9) 
$$a_j = a_j^+ - a_j^-, \text{ and } |a_j| = a_j^+ + a_j^-$$

for each j. If follows that  $\sum_{j=1}^{\infty} a_j$  converges absolutely if and only if  $\sum_{j=1}^{\infty} a_j^+$ and  $\sum_{j=1}^{\infty} a_j^-$  both converge, in which case

(3.14.10) 
$$\sum_{j=1}^{\infty} a_j = \sum_{j=1}^{\infty} a_j^+ - \sum_{j=1}^{\infty} a_j^-.$$

One can use this to obtain (3.14.2) from the analogous statements for  $\sum_{j=1}^{\infty} a_j^+$ 

and  $\sum_{j=1}^{\infty} a_j^{-1}$ . If the  $a_j$ 's are complex numbers, then one can reduce to the real case, by

Alternatively, if l and n are positive integers that satisfy (3.14.5), then one can check that

(3.14.11) 
$$\left|\sum_{j=1}^{n} a_{\pi(j)} - \sum_{j=1}^{n} a_{j}\right| \le \sum_{j=l+1}^{\infty} |a_{j}|.$$

One can use this to get (3.14.2) when  $\sum_{j=1}^{\infty} a_j$  converges absolutely.

#### Convergence without absolute convergence 3.14.3

Suppose that  $\sum_{j=1}^{\infty} a_j$  is an infinite series of real numbers that converges, but does not converge absolutely. One can check that neither  $\sum_{i=1}^{\infty} a_i^+$  nor  $\sum_{i=1}^{\infty} a_i^$ converge under these conditions.

One can use this to show that there are rearrangements of  $\sum_{j=1}^{\infty} a_j$  that do not converge, as well as rearrangements whose sum can be any real number. This corresponds to Theorem 3.5 D and Exercises 4, 5 on p77, 80 of [82], respectively, and Theorem 3.54 on p76 of [192].

See [45, 51, 71, 75, 102, 103, 138, 187, 197, 203] for more on rearrangements of infinite series.

#### 3.15Cauchy products of infinite series

Let  $\sum_{j=0}^{\infty} a_j$  and  $\sum_{l=0}^{\infty} b_l$  be infinite series of complex numbers. Put

(3.15.1) 
$$c_n = \sum_{j=0}^n a_j \, b_{n-j}$$

for every nonnegative integer *n*. The corresponding infinite series  $\sum_{n=0}^{\infty} c_n$  is called the *Cauchy product* of  $\sum_{j=0}^{\infty} a_j$  and  $\sum_{l=0}^{\infty} b_l$ .

If  $a_j = 0$  for all but finitely many  $j \ge 0$ , and  $b_l = 0$  for all but finitely many  $l \ge 0$ , then one can verify that  $c_n = 0$  for all but finitely many  $n \ge 0$ , and that

(3.15.2) 
$$\sum_{n=0}^{\infty} c_n = \left(\sum_{j=0}^{\infty} a_j\right) \left(\sum_{l=0}^{\infty} b_l\right).$$

#### 3.15.1 Cauchy products of series with nonnegative terms

Suppose for the moment that  $a_j$  is a nonnegative real number for every  $j \ge 0$ , and that  $b_l$  is a nonnegative real number for every  $l \ge 0$ , so that  $c_n$  is a nonnegative real number for every  $n \ge 0$ . If N is a nonnegative integer, then it is easy to see that

(3.15.3) 
$$\sum_{n=0}^{N} c_n \le \left(\sum_{j=0}^{N} a_j\right) \left(\sum_{l=0}^{N} b_l\right).$$

Similarly, if  $N_1$  and  $N_2$  are nonnegative integers, then one can check that

(3.15.4) 
$$\left(\sum_{j=0}^{N_1} a_j\right) \left(\sum_{l=0}^{N_2} b_l\right) \le \sum_{n=0}^{N_1+N_2} c_n.$$

If  $\sum_{j=0}^{\infty} a_j$  and  $\sum_{l=0}^{\infty} b_l$  converge, then one can use (3.15.3) to get that  $\sum_{n=0}^{\infty} c_n$  converges, with

(3.15.5) 
$$\sum_{n=0}^{\infty} c_n \le \left(\sum_{j=0}^{\infty} a_j\right) \left(\sum_{l=0}^{\infty} b_l\right).$$

In this case, one can also use (3.15.4) to get that

(3.15.6) 
$$\left(\sum_{j=0}^{\infty} a_j\right) \left(\sum_{l=0}^{\infty} b_l\right) \le \sum_{n=0}^{\infty} c_n,$$

so that (3.15.2) holds.

#### 3.15.2 Cauchy products and absolute convergence

If the  $a_j$ 's and  $b_l$ 's are complex numbers, then

(3.15.7) 
$$|c_n| \le \sum_{j=0}^n |a_j| |b_{n-j}|$$

for every  $n \ge 0$ . The sum on the right is the same as the *n*th term of the Cauchy product of  $\sum_{j=0}^{\infty} |a_j|$  and  $\sum_{l=0}^{\infty} |b_l|$ . If these two series converge, then their Cauchy product converges too, as in the preceding paragraph.

Under these conditions,  $\sum_{n=0}^{\infty} |c_n|$  converges as well, by the comparison test. More precisely, we have that

(3.15.8) 
$$\sum_{n=0}^{\infty} |c_n| \le \left(\sum_{j=0}^{\infty} |a_j|\right) \left(\sum_{l=0}^{\infty} |b_l|\right).$$

Thus  $\sum_{n=0}^{\infty} c_n$  converges absolutely when  $\sum_{j=0}^{\infty} a_j$  and  $\sum_{l=0}^{\infty} b_l$  converge absolutely. One can check that (3.15.2) holds in this situation, by reducing to the case of convergent series of nonnegative real numbers, as in the previous section. This corresponds to the theorem on p78 of [82], although (3.15.2) is

established another way there. If at least one of  $\sum_{j=0}^{\infty} a_j$  and  $\sum_{l=0}^{\infty} b_l$  converges absolutely, and the other converges, then Theorem 3.50 on p74 of [192] states that  $\sum_{n=0}^{\infty} c_n$  converges, and that (3.15.2) holds. Example 3.49 on p73 of [192] shows that  $\sum_{n=0}^{\infty} c_n$  may not converge, even if  $\sum_{j=0}^{\infty} a_j$  and  $\sum_{l=0}^{\infty} b_l$  both converge. See [139] for more on Cauchy products.

# Chapter 4

# **Continuous** mappings

#### **4.1** Continuity at a point

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces, and let f be a function defined on M with values in N. This is also known as a *mapping* from M into N, which may be expressed by  $f: M \to N$ .

**Definition 4.1.1** We say that f is continuous at a point  $x \in M$  if for every  $\epsilon > 0$  there is a  $\delta > 0$  such that

(4.1.2) 
$$\rho(f(x), f(y)) < \epsilon$$

for every  $y \in M$  with  $d(x, y) < \delta$ .

#### 4.1.1Characterizing continuity in terms of convergent sequences

**Proposition 4.1.3** A mapping f from M into N is continuous at  $x \in M$  if and only if for every sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of M that converges to x,  ${f(x_j)}_{j=1}^{\infty}$  converges to f(x) in N.

Suppose first that f is continuous at x, and let  $\{x_j\}_{j=1}^{\infty}$  be a sequence of elements of M that converges to x. Also let  $\epsilon > 0$  be given, and let  $\delta > 0$  be as in Definition 4.1.1. Because  $\{x_j\}_{j=1}^{\infty}$  converges to x in M, there is a positive integer L such that δ

$$(4.1.4) d(x,x_j) < \epsilon$$

for every  $j \ge L$ . This implies that

(4.1.5) 
$$\rho(f(x), f(x_j)) < \epsilon$$

for every  $j \ge L$ , by (4.1.2), as desired.

Conversely, we would like to show that f is continuous at x when it has this property in terms of sequences. Let  $\epsilon > 0$  be given, and suppose for the sake of a contradiction that there is no  $\delta > 0$  as in Definition 4.1.1. This means that for every  $\delta > 0$  there is a point  $x(\delta) \in M$  such that  $d(x, x(\delta)) < \delta$  and

(4.1.6) 
$$\rho(f(x), f(x(\delta))) \ge \epsilon.$$

If j is a positive integer, then we can apply this with  $\delta = 1/j$ , to get a point  $x_j \in M$  such that

$$(4.1.7) d(x, x_j) < 1/j$$

and

(4.1.8) 
$$\rho(f(x), f(x_j)) \ge \epsilon.$$

This leads to a sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of M that converges to x in M, and for which  $\{f(x_j)\}_{j=1}^{\infty}$  does not converge to f(x) in N, as desired.

#### 4.1.2 Continuous complex-valued functions

**Proposition 4.1.9** Let  $(M, d(\cdot, \cdot))$  be a metric space, and let f, g be complexvalued functions on M. Suppose that f and g are continuous at a point  $x \in M$ , with respect to the standard metric on  $\mathbb{C}$ . Under these conditions, f + g and f g are continuous at x as well. If, for every  $w \in M$ ,  $f(w) \neq 0$ , then 1/f is continuous at x too.

This follows from Proposition 4.1.3, and previous results about convergent sequences of complex numbers.

#### 4.1.3 Some examples of continuous functions

**Definition 4.1.10** Let  $(M, d(\cdot, \cdot))$  and  $(N, \rho(\cdot, \cdot))$  be metric spaces. A mapping f from M into N is said to be continuous on M if f is continuous at every point  $x \in M$ .

One can use Proposition 4.1.9 to show that

(4.1.11) polynomial functions on 
$$\mathbf{R}^n$$
 are continuous,

with respect to the standard Euclidean metric on  $\mathbf{R}^n$ . Similarly,

(4.1.12) rational functions are continuous,

when the denominator is not zero.

If z and w are complex numbers, then one can check that

$$(4.1.13) ||z| - |w|| \le |z - w|,$$

using the triangle inequality. One can use this to show that the absolute value defines a continuous real-valued function on the complex plane, with respect to the standard Euclidean metrics on  $\mathbf{R}$  and  $\mathbf{C}$ .

### 4.2 Compositions and inverse images

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces.

#### 4.2.1 Continuity and open sets

**Proposition 4.2.1** A mapping f from M into N is continuous if and only if for every open set  $V \subseteq N$ , its inverse image

(4.2.2) 
$$f^{-1}(V) = \{x \in M : f(x) \in V\}$$

is an open set in M.

Suppose that f is continuous, and let an open set  $V \subseteq N$  be given. Let  $x \in f^{-1}(V)$  be given, so that  $f(x) \in V$ . Because V is an open set in N, there is an  $\epsilon > 0$  such that

$$(4.2.3) \qquad \{z \in N : \rho(f(x), z) < \epsilon\} \subseteq V.$$

Using the continuity of f at x, we get that there is a  $\delta > 0$  such that if  $y \in M$  satisfies  $d(x,y) < \delta$ , then  $\rho(f(x), f(y)) < \epsilon$ , and hence  $f(y) \in V$ . This shows that

(4.2.4) 
$$\{y \in M : d(x,y) < \delta\} \subseteq f^{-1}(V),$$

as desired.

Conversely, suppose that f has the property described in the proposition, and let us show that f is continuous on M. Let  $x \in M$  be given, and let us show that f is continuous at x. To do this, let  $\epsilon > 0$  be given, and note that

(4.2.5) 
$$V = \{ z \in N : \rho(f(x), z) < \epsilon \}$$

is an open set in N, because it is an open ball. Thus  $f^{-1}(V)$  is an open set in M, by hypothesis. Of course,  $f(x) \in V$ , by construction, so that  $x \in f^{-1}(V)$ . It follows that there is a  $\delta > 0$  such that (4.2.4) holds, by the definition of an open set. If  $y \in M$  satisfies  $d(x, y) < \delta$ , then we get that  $y \in f^{-1}(V)$ , so that  $f(y) \in V$ , as desired.

#### 4.2.2 Continuity and closed sets

**Corollary 4.2.6** A mapping f from M into N is continuous if and only if for every closed set  $E \subseteq N$ ,  $f^{-1}(E)$  is a closed set in M.

It is easy to see that

(4.2.7) 
$$f^{-1}(N \setminus E) = M \setminus f^{-1}(E)$$

for every  $E \subseteq N$ . Using this, one can check that the condition in Corollary 4.2.6 if equivalent to the one in Proposition 4.2.1, because a subset of a metric space is a closed set if and only if its complement is an open set.

#### 4.2.3 Compositions of continuous mappings

**Proposition 4.2.8** Let  $(M_1, d_1(\cdot, \cdot))$ ,  $(M_2, d_2(\cdot, \cdot))$ , and  $(M_3, d_3(\cdot, \cdot))$  be metric spaces, let f be a continuous mapping from  $M_1$  into  $M_2$ , and let g be a continuous mapping from  $M_2$  into  $M_3$ . Under these conditions, the composition  $g \circ f$  of f and g is continuous as a mapping from  $M_1$  into  $M_3$ .

More precisely,  $g \circ f$  is defined by putting  $(g \circ f)(w) = g(f(w))$  for every  $w \in M_1$ .

To prove the proposition, let  $x \in M_1$  be given, and let us check that  $g \circ f$  is continuous at x. Let  $\epsilon > 0$  be given, and let us use the continuity of g at f(x) to get that there is an  $\eta > 0$  such that

(4.2.9) 
$$d_3(g(f(x)), g(u)) < \epsilon$$

for every  $u \in M_2$  with  $d_2(f(x), u) < \eta$ . The continuity of f at x implies that there is a  $\delta > 0$  such that

(4.2.10) 
$$d_2(f(x), f(y)) < \eta$$

for every  $y \in M_1$  with  $d_1(x, y) < \delta$ . In this case, we can take u = f(y) in (4.2.9), to get that

(4.2.11) 
$$d_3(g(f(x)), g(f(y))) < \epsilon_3$$

as desired.

#### 4.2.4 Another proof using convergent sequences

Alternatively, if  $\{x_j\}_{j=1}^{\infty}$  is any sequence of elements of  $M_1$  that converges to x, then

(4.2.12)  $\{f(x_j)\}_{j=1}^{\infty}$  converges to f(x)

in  $M_2$ , because f is continuous at x. This implies that

(4.2.13) 
$$\{g(f(x_j))\}_{j=1}^{\infty}$$
 converges to  $g(f(x))$ 

in  $M_3$ , because g is continuous at f(x), as desired.

#### 4.2.5 A proof using open sets

We can also use the characterization of continuity in Proposition 4.2.1. Let W be an open subset of  $M_3$ , so that

(4.2.14) 
$$g^{-1}(W)$$
 is an open subset of  $M_2$ ,

by the continuity of g. This implies that

(4.2.15) 
$$f^{-1}(g^{-1}(W))$$
 is an open set in  $M_1$ ,

by the continuity of f. It is easy to see that

(4.2.16) 
$$(g \circ f)^{-1}(W) = f^{-1}(g^{-1}(W)),$$

directly from the definitions. Thus

(4.2.17) 
$$(g \circ f)^{-1}(W)$$
 is an open set in  $M_1$ ,

as desired.

#### 4.3 Images of compact sets

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces.

**Theorem 4.3.1** If f is a continuous mapping from M into N and K is a compact subset of M, then

(4.3.2) 
$$f(K) = \{f(x) : x \in K\}$$

is a compact subset of N.

Let  $\{U_{\alpha}\}_{\alpha \in A}$  be an arbitrary open covering of f(K) in N. Thus  $U_{\alpha}$  is an open subset of N for every  $\alpha \in A$ , so that  $f^{-1}(U_{\alpha})$  is an open set in M for every  $\alpha \in A$ , by Proposition 4.2.1. One can check that

(4.3.3) 
$$K \subseteq \bigcup_{\alpha \in A} f^{-1}(U_{\alpha}),$$

because  $f(K) \subseteq \bigcup_{\alpha \in A} U_{\alpha}$ , by hypothesis. This shows that  $\{f^{-1}(U_{\alpha})\}_{\alpha \in A}$  is an open covering of K in M.

If K is compact in M, then it follows that there are finitely many indices  $\alpha_1, \ldots, \alpha_n \in A$  such that

n

(4.3.4) 
$$K \subseteq \bigcup_{j=1}^{n} f^{-1}(U_{\alpha_j})$$

Using this, one can verify that

(4.3.5) 
$$f(K) \subseteq \bigcup_{j=1}^{n} U_{\alpha_j};$$

as desired.

#### 4.3.1 Continuity and sequential compactness

Alternatively, suppose that K is sequentially compact in M, and let us show that

(4.3.6) f(K) is sequentially compact in N.

Let  $\{z_j\}_{j=1}^{\infty}$  be an arbitrary sequence of elements of f(K). If j is any positive integer, then let us choose  $x_j \in K$  such that

$$(4.3.7) f(x_j) = z_j.$$

Because K is sequentially compact, there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  that converges to an element x of K in M. Thus

(4.3.8) 
$$\{z_{j_l}\}_{l=1}^{\infty} = \{f(x_{j_l})\}_{l=1}^{\infty}$$

is a subsequence of  $\{z_j\}_{j=1}^{\infty}$  that converges to  $f(x) \in f(K)$  in N, because f is continuous at x, by hypothesis.

#### 4.3.2 The extreme value theorem

**Corollary 4.3.9 (Extreme value theorem)** Suppose that f is a continuous real-valued function on M, with respect to the standard Euclidean metric on  $\mathbf{R}$ . If K is a nonempty compact subset of M, then there are points  $p, q \in K$  such that

$$(4.3.10) f(p) \le f(x) \le f(q)$$

for every  $x \in M$ .

Theorem 4.3.1 implies that f(K) is a compact subset of the real line, so that f(K) is closed and bounded. Of course,  $f(K) \neq \emptyset$ , because  $K \neq \emptyset$ , by hypothesis, so that the supremum and infimum of f(K) exist in **R**. In this situation, the supremum and infimum of f(K) are elements of f(K), because f(K) is a closed set in **R**, as in Proposition 2.8.8. This is the same as saying that the maximum and minimum of f are attained on K, as desired.

## 4.4 Uniform continuity

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces.

**Definition 4.4.1** A mapping f from M into N is said to be uniformly continuous on M if for every  $\epsilon > 0$  there is a  $\delta > 0$  such that

$$(4.4.2) \qquad \qquad \rho(f(x), f(y)) < \epsilon$$

for every  $x, y \in M$  with  $d(x, y) < \delta$ .

Note that uniform continuity implies ordinary continuity. In order to get an example where the converse does not hold, one can take  $M = N = \mathbf{R}$  with the standard Euclidean metric, and put  $f(x) = x^2$  for every  $x \in \mathbf{R}$ . One can check that f is not uniformly continuous on  $\mathbf{R}$ , using the fact that

(4.4.3) 
$$f(x) - f(y) = x^2 - y^2 = (x+y)(x-y)$$

for every  $x, y \in \mathbf{R}$ .

Alternatively, let us take M to be the open unit interval (0, 1) in the real line, equipped with the restriction of the standard Euclidean metric on  $\mathbf{R}$  to (0, 1). If we put f(x) = 1/x for every  $x \in (0, 1)$ , then one can verify that f is not uniformly continuous on (0, 1).

#### 4.4.1 Uniform continuity and compactness

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be arbitrary metric spaces again.

**Theorem 4.4.4** If f is a continuous mapping from M into N, and if M is compact, then f is uniformly continuous on M.

Let  $\epsilon > 0$  be given. If  $x \in M,$  then there is a positive real number  $\delta(x)$  such that

$$(4.4.5) \qquad \qquad \rho(f(x), f(y)) < \epsilon/2$$

for every  $y \in M$  with  $d(x, y) < \delta(x)$ . Let B(x) be the open ball in M centered at x with radius  $\delta(x)/2$ . The collection of these open balls B(x),  $x \in M$ , is an open covering of M, because  $x \in B(x)$  for every  $x \in M$ , and open balls in Mare open sets. If M is compact, then there are finitely many elements  $x_1, \ldots, x_l$ of M such that

(4.4.6) 
$$M \subseteq \bigcup_{j=1}^{l} B(x_j)$$

Put

(4.4.7) 
$$\delta = \min\{\delta(x_j)/2 : 1 \le j \le l\}$$

which is a positive real number. Let w and y be elements of M such that  $d(w, y) < \delta$ . Using (4.4.6), we get that there is a  $j \in \{1, \ldots, l\}$  such that  $w \in B(x_j)$ , so that  $d(x_j, w) < \delta(x_j)/2$ . It follows that

(4.4.8) 
$$d(x_j, y) \le d(x_j, w) + d(w, y) < \delta(x_j)/2 + \delta \le \delta(x_j).$$

This implies that

$$(4.4.9) \quad \rho(f(w), f(y)) \le \rho(f(w), f(x_j)) + \rho(f(x_j), f(y)) < \epsilon/2 + \epsilon/2 = \epsilon,$$

as desired.

#### 4.4.2 Another proof using sequential compactness

Alternatively, let  $\epsilon > 0$  be given again, and suppose for the sake of a contradiction that there is no  $\delta > 0$  that satisfies the condition in the definition of uniform continuity. This means that for every  $\delta > 0$  there are points  $x(\delta), y(\delta) \in M$  such that  $d(x(\delta), y(\delta)) < \delta$  and

(4.4.10) 
$$\rho(f(x(\delta)), f(y(\delta))) \ge \epsilon.$$

If j is a positive integer, then we can take  $\delta = 1/j$ , to get points  $x_j, y_j \in M$ such that  $d(x_j, y_j) < 1/j$  and

(4.4.11) 
$$\rho(f(x_j), f(y_j)) \ge \epsilon.$$

Because M is compact, and thus sequentially compact, there is a subsequence  $\{x_{j_l}\}_{l=1}^{\infty}$  of  $\{x_j\}_{j=1}^{\infty}$  that converges to  $x \in M$ . Using the same sequence of indices

 $\{j_l\}_{l=1}^{\infty}$ , we get a corresponding subsequence  $\{y_{j_l}\}_{l=1}^{\infty}$  of  $\{y_j\}_{j=1}^{\infty}$ . It is easy to see that  $\{y_{j_l}\}_{l=1}^{\infty}$  converges to x in M in this situation as well.

This implies that  $\{f(x_{j_l})\}_{l=1}^{\infty}$  and  $\{f(y_{j_l})\}_{l=1}^{\infty}$  both converge to f(x) in N, because f is continuous at x, by hypothesis. Of course,

$$(4.4.12) \qquad \rho(f(x_{j_l}), f(y_{j_l})) \le \rho(f(x_{j_l}), f(x)) + \rho(f(x), f(y_{j_l}))$$

for every  $l \ge 1$ , by the triangle inequality. The right side of (4.4.12) is as small as we want when l is sufficiently large, because  $\{f(x_{j_l})\}_{l=1}^{\infty}$  and  $\{f(y_{j_l})\}_{l=1}^{\infty}$ converge to f(x) in N. This contradicts (4.4.11), as desired.

#### 4.5 Images of connected sets

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces.

**Theorem 4.5.1** If f is a continuous mapping from M into N, and if E is a connected subset of M, then f(E) is connected as a subset of N.

Suppose for the sake of a contradiction that f(E) is not connected in N. This means that there are nonempty separated subsets A, B of N such that  $f(E) = A \cup B$ . Put

(4.5.2) 
$$A_1 = f^{-1}(A) \cap E, \quad B_1 = f^{-1}(B) \cap E.$$

It is easy to see that  $A_1 \cup B_1 = E$ , by construction. We also have that  $A_1, B_1 \neq \emptyset$ , because A and B are nonempty subsets of f(E).

We would like to check that  $A_1$  and  $B_1$  are separated in M. Observe that  $A_1 \subseteq f^{-1}(A) \subseteq f^{-1}(\overline{A})$ , where  $\overline{A}$  is the closure of A in N. Remember that  $\overline{A}$  is a closed set in N, so that  $f^{-1}(\overline{A})$  is a closed set in M, as in Corollary 4.2.6. Using this, one can check that the closure  $\overline{A_1}$  of  $A_1$  in M is contained in  $f^{-1}(\overline{A})$ . This implies that

(4.5.3) 
$$\overline{A_1} \cap B_1 \subseteq f^{-1}(\overline{A}) \cap f^{-1}(B).$$

However,  $f^{-1}(\overline{A}) \cap f^{-1}(B) = f^{-1}(\overline{A} \cap B) = f^{-1}(\emptyset) = \emptyset$ , because A and B are separated in N. It follows that  $\overline{A_1} \cap B_1 = \emptyset$ . One can show that  $A_1 \cap \overline{B_1} = \emptyset$  in the same way, so that  $A_1$  and  $B_1$  are separated in M. This means that E is not connected in M, as desired.

#### 4.5.1 The intermedate value theorem

**Corollary 4.5.4 (Intermediate value theorem)** Let a and b be real numbers with a < b, and let f be a continuous real-valued function on the closed interval [a,b], with respect to the standard Euclidean metric on  $\mathbf{R}$  and its restriction to [a,b]. If  $t \in \mathbf{R}$  satisfies f(a) < t < f(b) or f(b) < t < f(a), then there is an  $x \in (a,b)$  such that f(x) = t.

Let us first extend f to a real-valued function on the real line, by putting f(y) = f(a) when  $y \in \mathbf{R}$  satisfies y < a, and f(z) = f(b) when  $z \in \mathbf{R}$  satisfies z > b. It is easy to see that this extension is continuous on  $\mathbf{R}$ . Remember that [a, b] is connected as a subset of the real line, as in Theorem 2.8.4. It follows that f([a, b]) is connected in  $\mathbf{R}$  as well, by the previous theorem. If t is as in the statement of the corollary, then we get that  $t \in f([a, b])$ , using Theorem 2.8.4 again.

## 4.6 Uniform convergence

Let M be a set, and let  $(N, \rho(u, v))$  be a metric space. Also let  $\{f_j\}_{j=1}^{\infty}$  be a sequence of mappings from M into N, and let f be another mapping from M into N.

#### 4.6.1 Pointwise and uniform convergence

**Definition 4.6.1** We say that  $\{f_j\}_{j=1}^{\infty}$  converges to f pointwise on M if for every  $x \in M$ ,  $\{f_j(x)\}_{j=1}^{\infty}$  converges to f(x) in N. We say that  $\{f_j\}_{j=1}^{\infty}$  converges to f uniformly on M if for every  $\epsilon > 0$  there is a positive integer L such that

(4.6.2)  $\rho(f_j(x), f(x)) < \epsilon$ 

for every  $x \in M$  and  $j \geq L$ .

It is easy to see that uniform convergence implies pointwise convergence. To get an example where the converse does not hold, let us take M to be the closed unit interval [0, 1],  $N = \mathbf{R}$  with the standard Euclidean metric, and  $f_j(x) = x^j$  for every  $x \in [0, 1]$  and  $j \ge 1$ . We have seen that  $\{f_j\}_{j=1}^{\infty}$  converges pointwise to the real-valued function f defined on [0, 1] by putting f(x) = 0 when  $0 \le x < 1$ , and f(1) = 1. However, one can check that  $\{f_j\}_{j=1}^{\infty}$  does not converge to f uniformly on [0, 1].

#### 4.6.2 Uniform convergence and continuity

**Theorem 4.6.3** Let  $(M, d(\cdot, \cdot))$  and  $(N, \rho(\cdot, \cdot))$  be metric spaces, and let

$$(4.6.4) {f_j}_{j=1}^\infty$$

be a sequence of mappings from M into N that converges uniformly to a mapping f from M into N. If  $x \in M$  and  $f_j$  is continuous at x for every  $j \ge 1$ , then f is continuous at x too.

Let  $\epsilon > 0$  be given. Because  $\{f_j\}_{j=1}^{\infty}$  converges to f uniformly on M, there is a positive integer L such that

$$(4.6.5) \qquad \qquad \rho(f_j(w), f(w)) < \epsilon/3$$

for every  $w \in M$  and  $j \ge L$ . By hypothesis,  $f_L$  is continuous at x, and so there is a  $\delta_L > 0$  such that

$$(4.6.6) \qquad \qquad \rho(f_L(x), f_L(y)) < \epsilon/3$$

for every  $y \in M$  with  $d(x, y) < \delta_L$ . It follows that

$$\rho(f(x), f(y)) \leq \rho(f(x), f_L(x)) + \rho(f_L(x), f_L(y)) + \rho(f_L(y), f(y)) 
(4.6.7) < \epsilon/3 + \epsilon/3 + \epsilon/3 = \epsilon$$

for every  $y \in M$  with  $d(x, y) < \delta_L$ , as desired.

#### 4.6.3 Weierstrass' criterion for uniform convergence

The following criterion for uniform convergence is due to Weierstrass.

**Proposition 4.6.8** Let  $a_1, a_2, a_3, \ldots$  be an infinite sequence of complex-valued functions on a set M, and let  $A_1, A_2, A_3, \ldots$  be an infinite sequence of nonnegative real numbers. Suppose that

$$(4.6.9) |a_j(x)| \le A_j$$

for every  $x \in M$  and  $j \ge 1$ , and that  $\sum_{j=1}^{\infty} A_j$  converges. Note that  $\sum_{j=1}^{\infty} a_j(x)$  converges absolutely for every  $x \in M$ , by the comparison test. Under these conditions, the sequence of partial sums  $\sum_{j=1}^{n} a_j(x)$  converges to  $\sum_{j=1}^{\infty} a_j(x)$  uniformly on M as  $n \to \infty$ , with respect to the standard metric on  $\mathbf{C}$ .

If  $x \in M$  and n is a positive integer, then one can verify that

$$(4.6.10) \quad \left|\sum_{j=1}^{\infty} a_j(x) - \sum_{j=1}^{n} a_j(x)\right| = \left|\sum_{j=n+1}^{\infty} a_j(x)\right| \le \sum_{j=n+1}^{\infty} |a_j(x)| \le \sum_{j=n+1}^{\infty} A_j.$$

Of course,

(4.6.11) 
$$\sum_{j=n+1}^{\infty} A_j = \sum_{j=1}^{\infty} A_j - \sum_{j=1}^{n} A_j$$

tends to 0 as  $n \to \infty$ , because  $\sum_{j=1}^{\infty} A_j$  converges. The proposition follows easily from these two statements.

# 4.7 Continuity of power series

Let  $\sum_{j=0}^{\infty} c_j z^j$  be a power series with coefficients in **C**.

#### 4.7. CONTINUITY OF POWER SERIES

#### 4.7.1 Continuity on a closed disk

**Proposition 4.7.1** Suppose that  $\sum_{j=0}^{\infty} |c_j| r^j$  converges for some positive real number r. Consider the complex-valued function defined on the closed disk

$$\{z \in \mathbf{C} : |z| \le r\}$$

by

(4.7.3) 
$$f(z) = \sum_{j=0}^{\infty} c_j \, z^j,$$

where the series converges absolutely by the comparison test. Under these conditions, f is continuous on (4.7.2), with respect to the standard metric on  $\mathbf{C}$  and its restriction to (4.7.2).

If j is a nonnegative integer, then

(4.7.4) 
$$|c_j z^j| = |c_j| |z|^j \le |c_j| r^j$$

on (4.7.2). Thus the sequence of partial sums

$$(4.7.5) \qquad \qquad \sum_{j=0}^{n} c_j z^j$$

converges to f(z) uniformly on (4.7.2), by Weierstrass' criterion. We also have that (4.7.5) is continuous on the complex plane for every  $n \ge 0$ , and in particular the restrictions of these functions to (4.7.2) are continuous. The continuity of f on (4.7.2) now follows from Theorem 4.6.3.

#### 4.7.2 Continuity on an open disk, or the whole plane

**Proposition 4.7.6** Let R be a positive extended real number, and suppose that  $\sum_{j=0}^{\infty} |c_j| r^j$  converges for every positive real number r with r < R. Consider the complex-valued function defined on

$$\{z \in \mathbf{C} : |z| < R\}$$

by (4.7.3), where the series converges absolutely by the comparison test. Under these conditions, f is continuous on (4.7.7), with respect to the standard metric on **C**, and its restriction to (4.7.7).

Let a point  $z_0 \in \mathbf{C}$  with  $|z_0| < R$  be given, and let us check that f is continuous at  $z_0$ , as a complex-valued function defined on (4.7.7). The restriction of f to (4.7.2) is continuous when 0 < r < R, as in the previous proposition. If  $|z_0| < r < R$ , then one can check that the continuity of f at  $z_0$  as a function on (4.7.2) implies the continuity of f at  $z_0$  as a function on (4.7.7).

#### 4.8 The supremum metric

Let M be a set, and let  $(N, \rho(u, v))$  be a metric space.

#### 4.8.1 Bounded functions

**Definition 4.8.1** A mapping f from M into N is said to be bounded if f(M) is a bounded set in N.

Now let (M, d(x, y)) be a metric space as well. The space of all continuous mappings from M into N may be denoted C(M, N). The space of all mappings from M into N that are both bounded and continuous may be denoted  $C_b(M, N)$ . If M is compact, then every continuous mapping from M into N is bounded, by Theorem 4.3.1.

#### 4.8.2 Defining the supremum metric

Suppose from now on in this section that  $M \neq \emptyset$ . If f and g are bounded continuous mappings from M into N, then one can check that  $\rho(f(x), g(x))$  is bounded as a real-valued function on M. Put

(4.8.2) 
$$\theta(f,g) = \sup\{\rho(f(x),g(x)) : x \in M\},\$$

which is a nonnegative real number. If f(x) = g(x) for every  $x \in M$ , then  $\rho(f(x), g(x)) = 0$  for every  $x \in M$ , and  $\theta(f, g) = 0$ . Conversely, if  $\theta(f, g) = 0$ , then  $\rho(f(x), g(x)) = 0$  for every  $x \in M$ , so that f(x) = g(x) for every  $x \in M$ .

It is easy to see that

(4.8.3) 
$$\theta(f,g) = \theta(g,f)$$

because  $\rho(\cdot, \cdot)$  is symmetric. Let h be another bounded continuous mapping from M into N, and observe that

$$(4.8.4) \quad \rho(f(x), h(x)) \le \rho(f(x), g(x)) + \rho(g(x), h(x)) \le \theta(f, g) + \theta(g, h)$$

for every  $x \in M$ . This implies that

(4.8.5) 
$$\theta(f,h) \le \theta(f,g) + \theta(g,h).$$

Thus  $\theta(\cdot, \cdot)$  defines a metric on  $C_b(M, N)$ , which is known as the *supremum* metric.

#### 4.8.3 Convergence with respect to the supremum metric

**Proposition 4.8.6** A sequence  $\{f_j\}_{j=1}^{\infty}$  of bounded continuous mappings from M into N converges to  $f \in C_b(M, N)$  with respect to the supremum metric if and only if  $\{f_j\}_{j=1}^{\infty}$  converges to f uniformly on M.

Suppose first that  $\{f_j\}_{j=1}^{\infty}$  converges to f with respect to the supremum metric, and let  $\epsilon > 0$  be given. By definition of convergence of a sequence in a metric space, there is a positive integer L such that

(4.8.7) 
$$\theta(f_i, f) < \epsilon$$

for every  $j \ge L$ . If  $x \in M$ , then it follows that

(4.8.8) 
$$\rho(f_j(x), f(x)) \le \theta(f_j, f) < \epsilon$$

for every  $j \ge L$ . This means that  $\{f_j\}_{j=1}^{\infty}$  converges to f uniformly on M, as desired.

Conversely, suppose that  $\{f_j\}_{j=1}^{\infty}$  converges to f uniformly on M. Thus, for each  $\epsilon > 0$ , there is a positive integer  $L(\epsilon)$  such that

(4.8.9) 
$$\rho(f_j(x), f(x)) < \epsilon$$

for every  $x \in M$  and  $j \geq L(\epsilon)$ . It follows that

(4.8.10) 
$$\theta(f_j, f) \le \epsilon$$

for every  $j \ge L(\epsilon)$ , by the definition of the supremum metric. This means that

(4.8.11) 
$$\theta(f_j, f) \le \epsilon/2 < \epsilon$$

for every  $j \ge L(\epsilon/2)$ , so that  $\{f_j\}_{j=1}^{\infty}$  converges to f with respect to the supremum metric.

# **4.9** Completeness of $C_b(M, N)$

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be nonempty metric spaces.

**Theorem 4.9.1** If N is complete as a metric space with respect to  $\rho(\cdot, \cdot)$ , then  $C_b(M, N)$  is complete with respect to the supremum metric.

#### 4.9.1 Pointwise convergence of Cauchy sequences

Let  $\{f_j\}_{j=1}^{\infty}$  be a Cauchy sequence of elements of  $C_b(M, N)$ , with respect to the supremum metric. This means that for every  $\epsilon > 0$  there is a positive integer  $L(\epsilon)$  such that (4.9.2)  $\theta(f_i, f_l) < \epsilon$ 

for every  $j, l \ge L(\epsilon)$ . If  $x \in M$ , then we get that

(4.9.3) 
$$\rho(f_j(x), f_l(x)) \le \theta(f_j, f_l) < \epsilon$$

for every  $j, l \ge L(\epsilon)$ . This shows that  $\{f_j(x)\}_{j=1}^{\infty}$  is a Cauchy sequence in N. It follows that  $\{f_j(x)\}_{j=1}^{\infty}$  converges in N, because N is complete by hypothesis.

#### 4.9.2 Checking uniform convergence

Put

(4.9.4) 
$$f(x) = \lim_{j \to \infty} f_j(x)$$

for every  $x \in M$ , which defines a mapping from M into N. One can check that

(4.9.5) 
$$\rho(f(x), f_l(x)) \le \epsilon$$

for every  $x \in M$  and  $l \ge L(\epsilon)$ , using (4.9.3) and (4.9.4). More precisely, this uses the fact that

(4.9.6) 
$$\rho(f(x), f_l(x)) \le \rho(f(x), f_j(x)) + \rho(f_j(x), f_l(x))$$

for all  $j \ge 1$ , by the triangle inequality. This implies that

(4.9.7)  $\{f_l\}_{l=1}^{\infty}$  converges to f uniformly on M.

It follows that

(4.9.8) f is continuous on M,

by Theorem 4.6.3, and because  $f_l$  is continuous on M for every  $l \ge 1$ , by hypothesis.

#### 4.9.3 Boundedness of the limit

One can verify that

(4.9.9)

f is bounded on M,

using (4.9.5) with  $\epsilon = 1$  and l = L(1), and the boundedness of  $f_{L(1)}$  on M, by hypothesis. Thus  $f \in C_b(M, N)$ , and  $\{f_l\}_{l=1}^{\infty}$  converges to f with respect to the supremum metric, because of uniform convergence, as in the previous section.

### 4.10 The limit of a function

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces, let E be a subset of M, and suppose that  $p \in M$  is a limit point of E. Also let f be a function defined on E with values in N, and let q be an element of N.

**Definition 4.10.1** We say that the limit of f(x), as  $x \in E$  approaches p, is equal to q, if for every  $\epsilon > 0$  there is a  $\delta > 0$  such that

 $\lim_{x \to p} f(x) = q,$ 

(4.10.2)  $\rho(f(x),q) < \epsilon$ 

for every  $x \in E$  with  $d(p, x) < \delta$  and  $x \neq p$ .

In this case, we put 
$$(4.10.3) \qquad \qquad \lim_{x \in E \atop x \to p} f(x) = q,$$

or simply (4.10.4)

when E = M, or it is clear which set E being used.

#### 4.10.1Characterizing limits in terms of convergent sequences

**Proposition 4.10.5** Under the conditions just mentioned, (4.10.3) holds if and only if for every sequence  $\{x_j\}_{j=1}^{\infty}$  of elements of E that converges to p in M and satisfies  $x_j \neq p$  for each  $j \geq 1$ , we have that  $\{f(x_j)\}_{j=1}^{\infty}$  converges to q in N.

This is analogous to Proposition 4.1.3, and we omit the details. One can use this to get the uniqueness of the limit, when it exists, from the corresponding statement for convergent sequences.

#### Limits of complex-valued functions 4.10.2

**Proposition 4.10.6** Let f and g be complex-valued functions on E, and suppose that the limits of f(x) and g(x), as  $x \in E$  approaches p, exist, with respect to the standard metric on C. Under these conditions,

(4.10.7) 
$$\lim_{\substack{x \in E \\ x \to p}} (f(x) + g(x)) = \lim_{\substack{x \in E \\ x \to p}} f(x) + \lim_{\substack{x \in E \\ x \to p}} g(x)$$

and

(4.10.8) 
$$\lim_{\substack{x \in E \\ x \to p}} (f(x) g(x)) = \left(\lim_{\substack{x \in E \\ x \to p}} f(x)\right) \left(\lim_{\substack{x \in E \\ x \to p}} g(x)\right).$$

We also have that

(4.10.9) 
$$\lim_{\substack{x \in E \\ x \to p}} 1/f(x) = \left(\lim_{\substack{x \in E \\ x \to p}} f(x)\right)^{-1}$$

when  $f(x) \neq 0$  for every  $x \in E$ , and the limit on the right is not zero.

This follows from the previous proposition, and the corresponding statements for convergent sequences of complex numbers.

#### 4.10.3 Limits and continuity

Let f be a mapping from M into a metric space N again. If  $p \in M$  is a limit point of M, then it is easy to see that f is continuous at p if and only if

(4.10.10) 
$$\lim_{x \to p} f(x) = f(p)$$

If p is not a limit point of M, then one can check that f is automatically continuous at p.

#### 4.11**One-sided** limits

Let a and b be real numbers with a < b, and let  $(N, \rho(u, v))$  be a metric space. Also let f be a function defined on the open interval (a, b) in the real line with values in N. If  $p \in \mathbf{R}$  satisfies  $a \leq p < b$ , then put

(4.11.1) 
$$E_{+}(p) = \{ x \in \mathbf{R} : p < x < b \}.$$

The limit of f(x) as  $x \in (a, b)$  approaches p from the right is defined by

(4.11.2) 
$$f(p+) = \lim_{x \to p+} f(x) = \lim_{\substack{x \in E_+(p) \\ x \to p}} f(x)$$

when the limit on the right side exists, with respect to the standard Euclidean metric on **R**. Similarly, if  $p \in \mathbf{R}$  satisfies a , then put

(4.11.3) 
$$E_{-}(p) = \{x \in \mathbf{R} : a < x < p\}$$

The limit of f(x) as  $x \in (a, b)$  approaches p from the left is defined by

(4.11.4) 
$$f(p-) = \lim_{x \to p-} f(x) = \lim_{\substack{x \in E_{-}(p) \\ x \to p}} f(x).$$

when the limit on the right side exists. If  $p \in (a, b)$ , then it is easy to see that

$$(4.11.5) \qquad \qquad \lim_{x \to p} f(x)$$

exists if and only if the one-sided limits (4.11.2) and (4.11.4) exist and are equal, in which case (4.11.5) is equal to the common value of the one-sided limits.

#### 4.11.1 Monotonically increasing functions

**Definition 4.11.6** A real-valued function f on (a,b) is said to be monotonically increasing on (a,b) if for every  $x, y \in (a,b)$  with x < y, we have that

$$(4.11.7) f(x) \le f(y).$$

**Proposition 4.11.8** Let f be a monotonically increasing real-valued function on (a, b). If  $p \in (a, b)$ , then the one-sided limits of f at p exist, with respect to the standard Euclidean metric on  $\mathbf{R}$ , and satisfy

(4.11.9) 
$$f(p-) \le f(p) \le f(p+).$$

If q is another element of (a, b), and p < q, then

(4.11.10) 
$$f(p+) \le f(q-)$$

Let  $p \in (a, b)$  be given, and put

$$(4.11.11) A_+(p) = \{f(x) : p < x < b\}$$

and

$$(4.11.12) A_{-}(p) = \{f(x) : a < x < p\}$$

Note that f(p) is a lower bound for  $A_+(p)$ , and an upper bound for  $A_-(p)$ , because f is monotonically increasing on (a, b). In this situation, the one-sided limits of f at p are given by

(4.11.13) 
$$f(p+) = \inf A_+(p)$$

and

(4.11.14) 
$$f(p-) = \sup A_{-}(p)$$

To prove (4.11.13), let  $\epsilon > 0$  be given. Observe that there is a point  $x(\epsilon)$  in (p, b) such that

(4.11.15) 
$$f(x(\epsilon)) < \inf A_+(p) + \epsilon,$$

by definition of the infimum. If  $p < x \leq x(\epsilon)$ , then

(4.11.16) 
$$\inf A_+(p) \le f(x) \le f(x(\epsilon)) < \inf A_+(p) + \epsilon,$$

using the monotonicity of f in the second step. This implies (4.11.13), and the argument for (4.11.14) is analogous.

It is easy to obtain (4.11.9) from (4.11.13) and (4.11.14). If p < q < b, then let x be an element of (p,q). Using (4.11.13) and the analogue of (4.11.14) for q, we get that

(4.11.17) 
$$f(p+) \le f(x) \le f(q-).$$

More precisely, this also uses the fact that f(x) is an element of  $A_+(p)$  and  $A_-(q)$ . This implies (4.11.10).

#### 4.11.2 Continuity of monotone functions

**Theorem 4.11.18** If f is a monotonically increasing real-valued function on (a,b), then f is continuous at all but finitely or countably many points in (a,b), with respect to the standard Euclidean metric on  $\mathbf{R}$  and its restriction to (a,b).

Because the one-sided limits of f exist at every point in (a, b), f is continuous at  $p \in (a, b)$  if and only if

(4.11.19) 
$$f(p+) = f(p-) = f(p)$$

It follows that f is not continuous at p exactly when

$$(4.11.20) f(p-) < f(p+),$$

by (4.11.9). In this case, we can choose a rational number  $r(p) \in (f(p-), f(p+))$ . If f is not continuous at  $q \in (a, b)$  and p < q, then

$$(4.11.21) r(p) < f(p+) \le f(q-) < r(q),$$

by (4.11.10). Thus we get a one-to-one correspondence between the set of discontinuities of f in (a, b) and a set of rational numbers, which implies the theorem.

# 4.12 Prescribing jump discontinuities

Consider the real-valued function defined on the real line by

(4.12.1) 
$$I(x) = 0$$
 when  $x \le 0$   
= 1 when  $x > 0$ .

This may be called the *unit step function* on  $\mathbf{R}$ , as in Definition 6.14 on p129 of [192]. Note that I(x) is monotonically increasing on  $\mathbf{R}$ , and that it is discontinuous at 0.

If  $u \in \mathbf{R}$ , then

(4.12.2) 
$$I(x-u) = 0 \quad \text{when } x \le u$$
$$= 1 \quad \text{when } x > u$$

is monotonically increasing on  $\mathbf{R}$ , and discontinuous at u. One can use linear combinations of these functions with positive coefficients to get monotonically increasing functions on  $\mathbf{R}$  with discontinuities at any finite set of points.

#### 4.12.1 Some infinite series of functions

Let  $\{t_j\}_{j=1}^{\infty}$  be an infinite sequence of real numbers, and let  $\{c_j\}_{j=1}^{\infty}$  be an infinite sequence of complex numbers such that

$$(4.12.3)\qquad\qquad\qquad\sum_{j=1}^{\infty}|c_j|$$

converges. Put

(4.12.4) 
$$f(x) = \sum_{j=1}^{\infty} c_j I(x - t_j)$$

for each  $x \in \mathbf{R}$ , where the series on the right converges absolutely, by the comparison test. If  $c_j$  is a nonnegative real number for each j, then it is easy to see that f is a monotonically increasing real-valued function on  $\mathbf{R}$ .

The sequence of partial sums

(4.12.5) 
$$\sum_{j=1}^{n} c_j I(x-t_j)$$

converges to f uniformly on  $\mathbf{R}$ , by Weierstrass' criterion, as in Section 4.6. If  $x \in \mathbf{R}$  and  $x \neq t_j$  for each j, then it follows that

$$(4.12.6) f ext{ is continuous at } x,$$

as before. If the  $t_j$ 's are distinct elements of  $\mathbf{R}$ , and  $c_j \neq 0$  for each j, then one can check that f is discontinuous at  $t_l$  for each l. More precisely, the one-sided limits of f at  $t_l$  exist for each l, and satisfy

(4.12.7) 
$$f(t_l+) - f(t_l-) = c_l$$

in this case.

This type of construction is described another way in Remark 4.31 on p97 of [192]. This description is mentioned in Theorem 6.16 on p130 of [192], and some of its properties are indicated in Exercise 8 on p166 of [192].

86

## 4.13 Some limits of limits

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces, let E be a subset of M, and suppose that  $p \in M$  is a limit point of E. Also let  $\{f_j\}_{j=1}^{\infty}$  be a sequence of mappings from E into N that converges uniformly to a mapping f from E into N. Suppose that for each  $j \geq 1$ ,

(4.13.1) 
$$q_j = \lim_{\substack{x \in E \\ x \to p}} f_j(x),$$

where the existence of the limit on the right is part of the hypothesis. Under these conditions, one can show that

(4.13.2) 
$$\{q_j\}_{j=1}^{\infty}$$
 is a Cauchy sequence in N.

# **4.13.1** When $\{q_j\}_{j=1}^{\infty}$ converges in N

Suppose in addition that

(4.13.3) 
$$\{q_j\}_{j=1}^{\infty}$$
 converges to an element  $q$  of  $N$ ,

which holds automatically when  ${\cal N}$  is complete. Using this, one can also show that

(4.13.4) 
$$\lim_{\substack{x \in E \\ x \to p}} f(x) = q.$$

This corresponds to Theorem 7.11 on p149 of [192] when N is the complex plane, with the standard metric, and to part of Exercise 17 on p168 of [192] when N is any complete metric space. This is analogous to Theorem 4.6.3, and it can be used to obtain that result as well.

## 4.14 Path connectedness

Let (M, d(x, y)) be a metric space.

#### 4.14.1 Continuous paths and connectedness

**Proposition 4.14.1** Let a and b be real numbers with a < b, and let p be a continuous mapping from the closed interval [a, b] into M, with respect to the restriction of the standard Euclidean metric on **R** to [a, b]. Under these conditions, p([a, b]) is a connected subset of M.

As in an earlier proof, it is helpful to extend p to a mapping from the real line into M, by putting p(t) = p(a) when t < a, and p(t) = p(b) when t > b. It is easy to see that this extension is continuous as a mapping from  $\mathbf{R}$  into M. It follows that p([a, b]) is a connected subset of M, because [a, b] is connected in  $\mathbf{R}$ .

#### 4.14.2 Path connected sets

**Definition 4.14.2** A subset E of M is said to be path connected if for every pair of points  $x, y \in E$  there is a continuous mapping p from the closed unit interval [0,1] into M such that p(0) = x, p(1) = y, and  $p(t) \in E$  for every  $t \in [0,1]$ . This uses the restriction of the standard Euclidean metric on  $\mathbf{R}$  to [0,1].

#### 4.14.3 Path connected sets are connected

**Proposition 4.14.3** If  $E \subseteq M$  is path connected, then E is connected in M.

Suppose for the sake of a contradiction that E is not connected in M, so that there are nonempty separated subsets A, B of M such that  $A \cup B = E$ . Let x be an element of A, and let y be an element of B. If E is path connected in M, then there is a continuous mapping p from [0, 1] into M such that p(0) = x, p(1) = y, and  $p([0, 1]) \subseteq E$ .

Put (4.14.4)  $A_1 = A \cap p([0,1]), \quad B_1 = B \cap p([0,1]).$ Observe that (4.14.5)  $A_1 \cup B_1 = p([0,1]),$ 

because  $p([0,1]) \subseteq E = A \cup B$ . By construction,  $x \in A_1$  and  $y \in B_1$ , so that  $A_1, B_1 \neq \emptyset$ . One can check that  $A_1$  and  $B_1$  are separated in M, because A and B are separated in M, and  $A_1 \subseteq A$ ,  $B_1 \subseteq B$ . This implies that p([0,1]) is not connected in M, which is a contradiction, as desired.

#### 4.14.4 Convex subsets of $\mathbb{R}^n$

**Definition 4.14.6** Let n be a positive integer. A subset E of  $\mathbb{R}^n$  is said to be convex if for every  $x, y \in E$  and  $t \in \mathbb{R}$  with  $0 \le t \le 1$ , we have that

$$(4.14.7) (1-t) x + t y \in E.$$

It is easy to see that convex subsets of  $\mathbf{R}^n$  are path connected, with respect to the standard Euclidean metric on  $\mathbf{R}^n$ . Note that connected subsets of the real line are convex.

#### 4.14.5 More on path connectedness

**Proposition 4.14.8** Let M and N be metric spaces, and let f be a continuous mapping from M into N. If E is a path-connected subset of M, then f(E) is path connected in N.

This follows from the definition of path connectedness, and the fact that compositions of continuous mappings are continuous.

It is well known and not too difficult to show that connected open subsets of  $\mathbf{R}^n$  are connected. However, there are examples of connected subsets of  $\mathbf{R}^2$  that are not path connected.

### 4.15 Integral metrics and other topics

In the first part of this section, the reader is supposed to be familiar with the Riemann integral of a continuous function on a closed interval in the real line. Let a, b be real numbers, with a < b, and let

(4.15.1) 
$$C([a,b]) = C([a,b], \mathbf{R})$$

be the space of continuous real-valued functions on [a, b]. More precisely, this uses the standard Euclidean metric on **R**, and its restriction to [a, b].

### **4.15.1** The $L^1$ metric on C([a, b])

If  $f, g \in C([a, b])$ , then put

(4.15.2) 
$$d_1(f,g) = \int_a^b |f(x) - g(x)| \, dx.$$

One can check that this defines a metric on C([a, b]). In particular,  $d_1(f, g) > 0$  unless f = g on [a, b].

#### 4.15.2 Comparison with the supremum metric

Remember that continuous real-valued functions on [a, b] are automatically bounded, because [a, b] is compact. The supremum metric on C([a, b]) is defined by

(4.15.3) 
$$\theta(f,g) = \sup\{|f(x) - g(x)| : a \le x \le b\},\$$

as in Section 4.8. It is easy to see that

(4.15.4) 
$$d_1(f,g) \le (b-a)\,\theta(f,g)$$

for every  $f, g \in C([a, b])$ .

In particular, if  $\{f_j\}_{j=1}^{\infty}$  is a sequence of continuous real-valued functions on [a, b] that converges uniformly to a continuous real-valued function f on [a, b], then  $\{f_j\}_{j=1}^{\infty}$  converges to f with respect to  $d_1(\cdot, \cdot)$ .

However, C([a, b]) is not complete as a metric space with respect to (4.15.2). To get a complete metric space, one can use the *Lebesgue integral*.

#### 4.15.3 Isometric mappings and completions

Let (M, d(x, y)) and  $(N, \rho(u, v))$  be metric spaces. A mapping  $\phi$  from M into N is said to be an *isometry* if

(4.15.5) 
$$\rho(\phi(x), \phi(y)) = d(x, y)$$

for every  $x, y \in M$ . A completion of M may be defined as an isometry from M onto a dense subset of a complete metric space. It is well known that every metric space has a completion, which is unique up to isometric equivalence.

#### 4.15.4 The Baire category theorem

Suppose that M is complete. If  $U_1, U_2, U_3, \ldots$  is an infinite sequence of dense open subsets of M, then the *Baire category theorem* states that  $\bigcap_{j=1}^{\infty} U_j$  is dense in M too.

#### 4.15.5 The contraction mapping theorem

A mapping  $\phi$  from M into itself is said to be a *contraction* if there is a nonnegative real number c < 1 such that

$$(4.15.6) d(\phi(x), \phi(y)) \le c d(x, y)$$

for every  $x, y \in M$ . In this case, if M is also nonempty and complete, then the contraction mapping theorem states that there is a unique  $w \in M$  such that  $\phi(w) = w$ .

#### 4.15.6 Brouwer's fixed-point theorem

Let n be a positive integer, and let

(4.15.7) 
$$B_n = \left\{ x \in \mathbf{R}^n : \sum_{j=1}^n x_j^2 \le 1 \right\}$$

be the closed unit ball in  $\mathbb{R}^n$ . Suppose that f is a continuous mapping from  $B_n$  into itself, with respect to the restriction of the standard Euclidean metric on  $\mathbb{R}^n$  to  $B_n$ . Brouwer's fixed-point theorem says that there is an  $x \in B_n$  such that f(x) = x. This can be obtained from the intermediate value theorem when n = 1.

#### 4.15.7 Weierstrass' approximation theorem

Let a and b be real numbers, with a < b. Suppose that f is a continuous real-valued function on [a, b], with respect to the standard Euclidean metric on **R** and its restriction to [a, b]. Under these conditions, a famous theorem of Weierstrass says that f can be uniformly approximated by polynomials on [a, b].

# Chapter 5

# Differentiating functions of one variable

## 5.1 Functions on subsets of R

Let E be a nonempty subset of the real line, and let f be a real-valued function on E. In this and the next sections, we always use the standard Euclidean metric on  $\mathbf{R}$ , and its restriction to E.

#### 5.1.1 The derivative of a real-valued function

If

(5.1.1) 
$$x \in E$$
 is a limit point of  $E$ ,

then the *derivative* of f at x as a function on E may be defined as usual by

(5.1.2) 
$$f'(x) = \lim_{\substack{w \in E \\ w \to x}} \frac{f(w) - f(x)}{w - x},$$

when this limit exists. In this case, f is said to be *differentiable* at x, as a function on E.

Of course, one is often particularly interested in functions defined on something like an interval in the real line. Some of the usual arguments work for other types of subsets of  $\mathbf{R}$ , and some arguments involve more properties of E. See [129, 130] for some topics related to functions defined on rational numbers.

#### 5.1.2 Differentiability implies continuity

If f is differentiable at x as a function on E, then

(5.1.3) f is continuous at x as a function on E.

Indeed, under these conditions, we have that

$$\lim_{\substack{w \in E \\ w \to x}} (f(w) - f(x)) = \lim_{\substack{w \in E \\ w \to x}} \left( \left( \frac{f(w) - f(x)}{w - x} \right) (w - x) \right)$$
  
(5.1.4)
$$= \left( \lim_{\substack{w \in E \\ w \to x}} \left( \frac{f(w) - f(x)}{w - x} \right) \right) \left( \lim_{\substack{w \in E \\ w \to x}} (w - x) \right)$$
  
$$= f'(x) \cdot 0 = 0,$$

where the second step is as in Section 4.10.

#### 5.1.3 Differentiability of sums and products

Let g be another real-valued function on E that is differentiable at x. It is easy to see that

(5.1.5) 
$$f + g$$
 is differentiable at  $x$ ,

with

(5.1.6) 
$$(f+g)'(x) = f'(x) + g'(x)$$

This uses the analogous statement for limits of sums in Section 4.10. One can also check that

(5.1.7) 
$$fg$$
 is differentiable at  $x$ ,

with

(5.1.8) 
$$(fg)'(x) = f'(x)g(x) + f(x)g'(x),$$

which is the usual *product rule* for the derivative. This can be obtained in a standard way from the usual results about limits of products in Section 4.10.

#### 5.1.4 Differentiability of reciprocals

Suppose that (5.1.9)	$f(w) \neq 0$
for each $w \in E$ , so that	
(5.1.10)	1/f

defines a real-valued function on E too. In this case, one can show that

(5.1.11) 
$$1/f$$
 is differentiable at  $x$ ,

with

(5.1.12) 
$$(1/f)'(x) = -f'(x)/f(x)^2.$$

This uses the fact that 1/f is continuous at x as a function on E, because of (5.1.3). Note that (5.1.12) could be obtained from the product rule, if one knows (5.1.11).

More precisely, one can consider 1/f as a real-valued function on the set

(5.1.13) 
$$\{w \in E : f(w) \neq 0\}$$

even if f(w) = 0 for some  $w \in E$ . If  $f(x) \neq 0$ , then (5.1.13) contains every  $w \in E$  that is sufficiently close to x, because f is continuous at x. In particular, this means that x is a limit point of (5.1.13), because x is a limit point of E in **R**, by hypothesis.

Of course, the *quotient rule* for the derivative can be obtained from (5.1.12) and the product rule.

## 5.2 The chain rule

Let E be a nonempty subset of the real line again, and let f be a real-valued function on E. Suppose that  $x \in E$  is a limit point of E, and that f is differentiable at x. Let  $E_1$  be another subset of the real line, and suppose that

$$(5.2.1) f(E) \subseteq E_1,$$

and that

(5.2.2) 
$$f(x)$$
 is a limit point of  $E_1$ .

Also let g be a real-valued function on  $E_1$ , and put

 $(5.2.3) h = g \circ f,$ 

which is a real-valued function on E.

#### 5.2.1 The statement and some indications of the proof

 $\mathbf{If}$ 

(5.2.4) g is differentiable at f(x),

then the *chain rule* states that

(5.2.5) *h* is differentiable at *x*,

with

(5.2.6) 
$$h'(x) = g'(f(x)) f'(x).$$

This is very easy to see when g is a linear function on **R**. Using this, one can reduce to the case where

(5.2.7) g'(f(x)) = 0,

which is a bit simpler.

#### 5.2.2 Differentiating inverse functions

Suppose for the moment that f is a one-to-one function on E, and that g is the inverse of f on

(5.2.8)  $E_1 = f(E),$ 

so that (5.2.9) h(w) = g(f(w)) = w

for every  $w \in E$ . It is easy to see that (5.2.2) holds in this case, because f is continuous at x, as in the previous section. If (5.2.4) holds, then

(5.2.10) 
$$g'(f(x)) f'(x) = h'(x) = 1.$$

This means that  $f'(x) \neq 0$ , and that

(5.2.11) 
$$g'(f(x)) = 1/f'(x)$$

#### 5.2.3 Functions on open intervals

Suppose for the moment again that

(5.2.12) 
$$E = (a, b)$$

for some real numbers a, b with a < b, although one could also permit  $a = -\infty$ or  $b = +\infty$ . Suppose also that f is differentiable at every  $w \in (a, b)$ , with

(5.2.13) 
$$f'(w) > 0.$$

One can use the mean value theorem to get that f is strictly increasing on (a, b).

One can also check that f((a, b)) is an open interval in **R**, or an open halfline, or **R**. If g is the inverse of f on f((a, b)), then part of Exercise 2 on p114 of [192] states that

(5.2.14) g is differentiable at every point in f((a, b)).

#### 5.2.4 More on the differentiability of the inverse function

Let E be any nonempty subset of  $\mathbf{R}$  again, and suppose that  $x \in E$  is a limit point of E, as before. Suppose that f is one-to-one on E, and let g be the inverse of f on (5.2.8). Suppose also that f is differentiable at x, as usual, with  $f'(x) \neq 0$ . If

(5.2.15) g is continuous at f(x),

as a real-valued function on (5.2.8), then one can show that (5.2.4) holds. This corresponds to Theorem 7.5 E on p174 of [82], at least when E is an interval in **R**, which may be unbounded, as well as open, closed, or half-open and half-closed, as on p2 of [82].

More precisely, (5.2.15) implies that if  $w \in E$  and f(w) is close to f(x), then

(5.2.16) 
$$w = g(f(w))$$
 is close to  $x = g(f(x))$ .

Otherwise, let  $r_0$  be a positive real number, and let  $g_0$  be the restriction of g to

(5.2.17) 
$$f(E \cap (x - r_0, x + r_0)).$$

If  $r_0$  is small enough, then one can check that  $g_0$  is continuous at f(x) as a function on (5.2.17), using the hypothesis that  $f'(x) \neq 0$ .

## 5.3 One-sided derivatives

Let E be a nonempty subset of **R** again, and let f be a real-valued function on E. If  $x \in E$ , then put

(5.3.1)	$E^+(x) = \{ w \in E : w \ge x \}$
and $(5.3.2)$	$E^{-}(x) = \{ w \in E : w \le x \}.$
If (5.3.3)	$x$ is a limit point of $E^+(x)$ ,

then the derivative of f at x from the right as a function on E is defined by

(5.3.4) 
$$f'_{+}(x) = \lim_{\substack{w \in E^{+}(x) \\ w \to x}} \frac{f(w) - f(x)}{w - x},$$

when this limit exists.

Similarly, if

(5.3.5) x is a limit point of  $E^{-}(x)$ ,

then the *derivative of* f at x from the left as a function on E is defined by

(5.3.6) 
$$f'_{-}(x) = \lim_{\substack{w \in E^{-}(x) \\ w \to x}} \frac{f(w) - f(x)}{w - x},$$

when this limit exists. These are the same as the derivatives at x of the restrictions of f to  $E^+(x)$  and  $E^-(x)$ , respectively, when they exist.

If x is a limit point of  $E^+(x)$  or  $E^-(x)$ , then it is easy to see that x is a limit point of E. One can check that the converse holds as well.

Suppose for the moment that

(5.3.7) 
$$x$$
 is a limit point of  $E^+(x)$  and  $E^-(x)$ ,

so that x is a limit point of E in particular. If f is differentiable at x as a function on E, then it is easy to see that the derivatives of f at x from the left and right exist, with

(5.3.8) 
$$f'_+(x) = f'_-(x) = f'(x).$$

Conversely, if the derivatives of f at x from the left and right exist and are equal to each other, then f is differentiable at x as a function on E, with the derivative of f at x as in (5.3.8). This is very similar to a remark in Section 4.11.

### 5.4 More on one-sided derivatives

Let us continue with the same notation and hypotheses as in the previous section. Suppose for the moment again that

(5.4.1) 
$$f$$
 has a local maximum at  $x$ ,

as a function on E, so that (5.4.2)  $f(w) \le f(x)$ 

for all  $w \in E$  that are sufficiently close to x. If x is a limit point of  $E^+(x)$ , and if the derivative of f at x from the right as a function on E exists, then it is easy to see that

(5.4.3) 
$$f'_+(x) \le 0.$$

Similarly, if x is a limit point of  $E^{-}(x)$ , and if the derivative of f at x from the left as a function on E exists, then

(5.4.4) 
$$f'_{-}(x) \ge 0.$$

If (5.3.7) holds, and if f is differentiable at x as a function on E, then it follows that

(5.4.5) 
$$f'(x) = 0,$$

because of (5.3.8).

Similarly, suppose for the moment that

(5.4.6) 
$$f$$
 has a local minimum at  $x$ ,

as a function on E, so that

$$(5.4.7) f(x) \le f(w)$$

for all  $w \in E$  that are sufficiently close to x. If x is a limit point of  $E^+(x)$ , and if the derivative of f at x from the right as a function on E exists, then

(5.4.8) 
$$f'_+(x) \ge 0$$

If x is a limit point of  $E^{-}(x)$ , and if the deriviative of f at x from the left as a function on E exists, then

(5.4.9) 
$$f'_{-}(x) \le 0.$$

If (5.3.7) holds, and if f is differentiable at x as a function on E, then (5.4.5) holds, because of (5.3.8), as before. Of course, these statements correspond to those in the preceding paragraph for -f.

These remarks are related to Rolle's theorem and the mean value theorem when E is a closed interval in **R**. Remember that Rolle's theorem uses the extreme value theorem, as in Corollary 4.3.9. One might consider analogous types of arguments when E is the Cantor set, as in Section 2.9, for instance.

See [18, 47, 56, 127, 210] for some additional topics concerning the mean value theorem. Some related matters are discussed in [66, 68].

#### 5.5. THE DARBOUX PROPERTY

#### 5.4.1 The generalized mean value theorem

Let a and b be real numbers with a < b, and let f, g be continuous real-valued functions on [a, b] that are differentiable at every point in (a, b). Under these conditions, the *generalized mean value theorem* states that there is an  $x \in (a, b)$  such that

(5.4.10) 
$$(f(b) - f(a))g'(x) = (g(b) - g(a))f'(x),$$

as in Theorem 5.9 on p107 of [192]. To see this, put

(5.4.11) 
$$h(t) = (f(b) - f(a))g(t) - (g(b) - g(a))f(t)$$

for every  $t \in [a, b]$ . This is a continuous real-valued function on [a, b] that is differentiable at every point in (a, b), with

(5.4.12) 
$$h(a) = f(b) g(a) - f(a) g(b) = h(b).$$

One can use Rolle's theorem to get a point  $x \in (a, b)$  such that h'(x) = 0, and (5.4.10) follows from this.

#### 5.5 The Darboux property

Let a and b be real numbers with a < b, and let f be a real-valued function on [a, b] that is differentiable at every point in [a, b]. Suppose that  $\lambda \in \mathbf{R}$  satisfies

$$(5.5.1) f'(a) < \lambda < f'(b)$$

$$(5.5.2) f'(b) < \lambda < f'(a).$$

Under these conditions, there is an  $x \in (a, b)$  such that

$$(5.5.3) f'(x) = \lambda$$

To see this, it is convenient to suppose that (5.5.1) holds, and otherwise one can use an analogous argument when (5.5.2) holds, or reduce to this case. Let g be the real-valued function defined on [a, b] by

(5.5.4) 
$$g(w) = f(w) - \lambda w,$$

and observe that g is differentiable on [a, b], with

$$(5.5.5) g' = f' - \lambda.$$

In particular,

or

(5.5.6)	$g'(a) = f'(a) - \lambda < 0$
and	

(5.5.7) 
$$g'(b) = f'(b) - \lambda > 0,$$

because of (5.5.1).

The extreme value theorem implies that g attains its minimum at a point  $x \in [a, b]$ . One can check that  $x \in (a, b)$  in this case, using (5.5.6) and (5.5.7), as in the previous section. It follows that

(5.5.8) 
$$g'(x) = 0,$$

as in the previous section again. This implies (5.5.3), by (5.5.5). Of course, the same conclusion can be obtained from the intermediate value theorem when f' is continuous on [a, b].

#### 5.5.1 Another property of the derivative

Let f be a real-valued function on the real line that is differentiable at every point in **R**. Put

(5.5.9) 
$$f_j(x) = j \left( f(x + (1/j)) - f(x) \right)$$

for every  $x \in \mathbf{R}$  and positive integer j, and observe that

(5.5.10) 
$$\lim_{j \to \infty} f_j(x) = f'(x)$$

for every  $x \in \mathbf{R}$ . Pointwise limits of continuous functions are said to be of *Baire class one*. Thus f' is of Baire class one. Some related references include [31, 39, 40, 41, 62].

#### 5.6 Some examples

Let l be a positive integer, and consider the function  $f_l$  defined on the real line by

(5.6.1) 
$$f_l(x) = x^2 \sin(x^{-l})$$
 when  $x \neq 0$   
= 0 when  $x = 0$ .

We shall use standard properties of trigonometric functions here, although one could avoid this by using other nonconstant periodic differentiable functions on **R**. If  $x \neq 0$ , then  $f_l$  is differentiable at x, with

(5.6.2) 
$$f'_{l}(x) = 2x \sin(x^{-l}) + x^{2} (-l x^{-l-1}) \cos(x^{-l})$$
$$= 2x \sin(x^{-l}) - l x^{1-l} \cos(x^{-l}).$$

One can check directly that  $f_l$  is differentiable at 0, with

(5.6.3) 
$$f_l'(0) = 0.$$

Note that  
(5.6.4) 
$$\lim_{x \to 0} (2x \sin(x^{-l})) = 0,$$

because  $\sin(x^{-l})$  is bounded for  $x \neq 0$ . If l = 1, then (5.6.2) reduces to

(5.6.5) 
$$f_1'(x) = 2x \sin(1/x) - \cos(1/x)$$

for  $x \neq 0$ . It is easy to see that  $f'_1$  is not continuous at 0. If l > 1, then  $f'_l(x)$  is not even bounded for  $x \neq 0$  near 0.

Clearly (5.6.5) is bounded on bounded subsets of  $\mathbf{R} \setminus \{0\}$ . Of course, it is well known that

(5.6.6) 
$$\lim_{t \to 0} (t^{-1} \sin(t)) = \sin'(0) = 1.$$

One can use this with t = 1/x to get that

(5.6.7) 
$$f'_1(x) \to 1 \text{ as } x \to \pm \infty.$$

In particular, this implies that (5.6.5) is bounded on  $\mathbf{R} \setminus \{0\}$ . Similarly, if l > 1, then one can use (5.6.6) with  $t = x^{-l}$  to get that

(5.6.8) 
$$f'_l(x) \to 0 \text{ as } x \to \pm \infty.$$

See [123] for more on some related examples, and their properties.

# 5.7 Lipschitz mappings

Let  $(M, d(\cdot, \cdot))$  and  $(N, \rho(\cdot, \cdot))$  be metric spaces. A mapping f from M into N is said to be *Lipschitz* if there is a nonnegative real number C such that

(5.7.1) 
$$\rho(f(x), f(w)) \le C d(x, w)$$

for all  $x, w \in M$ . In this case, we may say that f is Lipschitz with constant C, to indicate the role of C. It is easy to see that

(5.7.2) Lipschitz mappings are uniformly continuous.

Note that

(5.7.3) 
$$f$$
 is Lipschitz with constant  $C = 0$  on  $M$  if and only if  $f$  is constant on  $M$ .

#### 5.7.1 Real-valued functions on subsets of the real line

Let E be a subset of the real line, and let f be a real-valued function on E. Note that f is Lipschitz with constant  $C \ge 0$  on E exactly when

(5.7.4) 
$$|f(x) - f(w)| \le C |x - w|$$

for all  $x, w \in E$ . This uses the standard Euclidean metric on **R**, and its restriction to *E*.

Suppose that  $x \in E$  is a limit point of E, and that f is differentiable at x. If f is Lipschitz with constant C on E, then one can check that

$$(5.7.5) |f'(x)| \le C.$$

Let a and b be real numbers with a < b, and let f be a continuous real-valued function on [a, b]. Suppose that f is differentiable at every point in (a, b), and that there is a nonnegative real number C such that (5.7.5) holds for every  $x \in (a, b)$ . Under these conditions, one can verify that

(5.7.6) 
$$f$$
 is Lipschitz with constant  $C$  on  $[a, b]$ ,

using the mean value theorem.

Let E be a subset of the real line again, and let f be a real-valued function on E. Suppose that  $x_0 \in E$  is a limit point of E, and that f is differentiable at  $x_0$ . If  $\epsilon_0$  is a positive real number, then one can check that there is a positive real number  $\delta_0$  such that

(5.7.7) 
$$\left|\frac{f(w) - f(x_0)}{w - x_0}\right| < |f'(x_0)| + \epsilon_0$$

for every  $w \in E$  with  $w \neq x_0$  and  $|w - x_0| < \delta_0$ . This implies that

(5.7.8) 
$$|f(w) - f(x_0)| \le (|f'(x_0)| + \epsilon_0) |w - x_0|$$

for every  $w \in E$  with  $|w - x_0| < \delta_0$ , because (5.7.8) holds trivially when  $w = x_0$ . This may be considered as a type of pointwise Lipschitz condition for f at  $x_0$ . In particular, this implies that f is continuous at  $x_0$ . Remember that the continuity of f at  $x_0$  was also mentioned in Subsection 5.1.2.

# Chapter 6

# Integrating functions of one variable

# 6.1 Suprema and infima of functions

Let E be a nonempty set, and let f be a real-valued function on E. Suppose that f is bounded on E, so that f(E) is a nonempty bounded subset of the real line. Under these conditions, we may use the notation

(6.1.1) 
$$\sup_{x \in E} f(x) = \sup f(E) = \sup \{ f(x) : x \in E \}$$

and

(6.1.2) 
$$\inf_{x \in E} f(x) = \inf f(E) = \inf \{ f(x) : x \in E \}.$$

Of course,

(6.1.3) 
$$\inf_{x \in E} f(x) \le \sup_{x \in E} f(x).$$

#### 6.1.1 Multiplication by a real number

If  $t \in \mathbf{R}$ , then it is easy to see that

(6.1.4) t f is a bounded real-valued function on E

too. If t > 0, then one can check that

(6.1.5) 
$$\sup_{x \in E} (t f(x)) = t \left( \sup_{x \in E} f(x) \right)$$

and

(6.1.6) 
$$\inf_{x \in E} (t f(x)) = t \left( \inf_{x \in E} f(x) \right).$$

Similarly, if t < 0, then

(6.1.7) 
$$\sup_{x \in E} (t f(x)) = t \left( \inf_{x \in E} f(x) \right)$$

and

and  
(6.1.8) 
$$\inf_{x \in E} (t f(x)) = t \left( \sup_{x \in E} f(x) \right).$$

#### 6.1.2 Sums of bounded functions

Let g be another bounded real-valued function on E, and observe that

(6.1.9) 
$$f + g$$
 is bounded on  $E$ 

as well. One can verify that

(6.1.10) 
$$\sup_{x \in E} (f(x) + g(x)) \le \left(\sup_{x \in E} f(x)\right) + \left(\sup_{x \in E} g(x)\right)$$

and

(6.1.11) 
$$\inf_{x \in E} (f(x) + g(x)) \ge \left(\inf_{x \in E} f(x)\right) + \left(\inf_{x \in E} g(x)\right).$$

#### More on suprema and infima of functions 6.2

Let us continue with the same notation and hypotheses as in the previous section. We also have that

(6.2.1) 
$$\sup_{x \in E} (f(x) + g(x)) \ge \left(\sup_{x \in E} f(x)\right) + \left(\inf_{x \in E} g(x)\right)$$

and

(6.2.2) 
$$\inf_{x \in E} (f(x) + g(x)) \le \left(\inf_{x \in E} f(x)\right) + \left(\sup_{x \in E} g(x)\right).$$

This can be verified directly, or using the previous inequalities and some of the earlier remarks with t = -1.

#### Products of nonnegative functions 6.2.1

Suppose for the moment that

$$(6.2.3) f,g \ge 0 \text{ on } E.$$

Under these conditions, one can check that

(6.2.4) 
$$\sup_{x \in E} (f(x) g(x)) \le \left( \sup_{x \in E} f(x) \right) \left( \sup_{x \in E} g(x) \right)$$

(6.2.5) 
$$\inf_{x \in E} (f(x) g(x)) \ge \left(\inf_{x \in E} f(x)\right) \left(\inf_{x \in E} g(x)\right).$$

#### 6.2.2 Some monotonicity properties

Suppose for the moment again that

$$(6.2.6) f(x) \le g(x)$$

for every  $x \in E$ . In this case, it is easy to see that

(6.2.7) 
$$\sup_{x \in E} f(x) \le \sup_{x \in E} g(x)$$

and

(6.2.8) 
$$\inf_{x \in E} f(x) \le \inf_{x \in E} g(x).$$

#### 6.2.3 Some more monotonicity properties

If  $E_0$  is a nonempty subset of E, then

(6.2.9) 
$$\sup_{x \in E_0} f(x) \le \sup_{x \in E} f(x)$$

and (6.2.10)  $\inf_{x \in E} f(x) \le \inf_{x \in E_0} f(x).$ 

## 6.3 Darboux sums

Let a and b be real numbers, with  $a \leq b$ , and let

(6.3.1) 
$$\mathcal{P} = \{t_j\}_{j=0}^l$$

be a *partition* of [a, b]. More precisely, this is a finite sequence of real numbers with

$$(6.3.2) a = t_0 \le t_1 \le \dots \le t_{l-1} \le t_l = b.$$

It is convenient to include the case where a = b here, but the various sums and integrals that we shall consider will be equal to 0 when this happens. Similarly, if  $t_{j-1} = t_j$  for some  $j \ge 1$ , then the corresponding term in each of the sums that we shall consider will be equal to 0.

#### 6.3.1 Upper and lower Darboux sums

Let f be a bounded real-valued function on [a, b], and put

(6.3.3) 
$$M_j = \sup\{f(x) : t_{j-1} \le x \le t_j\}$$

and

(6.3.4) 
$$m_j = \inf\{f(x) : t_{j-1} \le x \le t_j\}$$

for each  $j = 1, \ldots, l$ . Consider

(6.3.5) 
$$U(\mathcal{P}, f) = \sum_{j=1}^{l} M_j \left( t_j - t_{j-1} \right)$$

and

(6.3.6) 
$$L(\mathcal{P}, f) = \sum_{j=1}^{l} m_j (t_j - t_{j-1}),$$

as on p121 of [192], for instance. These are the *upper and lower Darboux sums* associated to f and  $\mathcal{P}$ , respectively.

#### 6.3.2 Riemann sums

If

(6.3.7) 
$$w_j \in [t_{j-1}, t_j], \ 1 \le j \le l,$$

then

(6.3.8) 
$$L(\mathcal{P}, f) \le \sum_{j=1}^{l} f(w_j) (t_j - t_{j-1}) \le U(\mathcal{P}, f).$$

The sum in the middle is the *Riemann sum* associated to  $f, w_1, \ldots, w_l$ , and the partition  $\mathcal{P}$ .

## 6.4 Darboux–Stieltjes sums

Let us continue with the same notation and hypotheses as in the previous section. Also let  $\alpha$  be a monotonically increasing real-valued function on [a, b], so that

(6.4.1) 
$$\alpha(t_j) - \alpha(t_{j-1}) \ge 0$$

for each  $j = 1, \ldots, l$ . Consider

(6.4.2) 
$$U(\mathcal{P}, f, \alpha) = \sum_{j=1}^{l} M_j \left( \alpha(t_j) - \alpha(t_{j-1}) \right)$$

and

(6.4.3) 
$$L(\mathcal{P}, f, \alpha) = \sum_{j=1}^{l} m_j \left( \alpha(t_j) - \alpha(t_{j-1}) \right),$$

as on p122 of [192]. These are the upper and lower Darboux sums or Darboux-Stieltjes sums associated to f,  $\mathcal{P}$ , and  $\alpha$ , respectively.

#### 6.4.1 Riemann–Stieltjes sums

If  $w_1, \ldots, w_l$  are as in (6.3.7), then

(6.4.4) 
$$L(\mathcal{P}, f, \alpha) \leq \sum_{j=1}^{l} f(w_j) \left( \alpha(t_j) - \alpha(t_{j-1}) \right) \leq U(\mathcal{P}, f, \alpha).$$

The sum in the middle is the *Riemann–Stieltjes sum* associated to f, the points  $w_1, \ldots, w_l$ , the partition  $\mathcal{P}$ , and  $\alpha$ .

#### 6.4.2 A basic sum using $\alpha$

Of course, if  $\alpha(x) = x$  on [a, b], then (6.4.2) and (6.4.3) are the same as (6.3.5) and (6.3.6), respectively. Note that

(6.4.5) 
$$\sum_{j=1}^{l} (\alpha(t_j) - \alpha(t_{j-1})) = \sum_{j=1}^{l} \alpha(t_j) - \sum_{j=1}^{l} \alpha(t_{j-1})$$
$$= \sum_{j=1}^{l} \alpha(t_j) - \sum_{j=0}^{l-1} \alpha(t_j) = \alpha(b) - \alpha(a).$$

# 6.5 More on Darboux sums

Let us continue with the same notation and hypotheses as in the previous two sections. Clearly

$$(6.5.1) m_j \le M_j$$

for each j. This implies that

(6.5.2) 
$$L(\mathcal{P}, f, \alpha) \le U(\mathcal{P}, f, \alpha),$$

which could also be obtained from (6.4.4). In particular,

(6.5.3) 
$$L(\mathcal{P}, f) \le U(\mathcal{P}, f),$$

which follows from (6.3.8) as well.

#### 6.5.1 Some basic estimates for upper and lower sums

If we put $(6.5.4)$	$M = \sup\{f(x) : a \le x \le b\}$
and $(6.5.5)$	$m = \inf\{f(x) : a \le x \le b\},\$
(6.5.6)	$M = \max_{1 \le j \le l} M_j$

and

(6.5.7) 
$$m = \min_{1 \le j \le l} m_j.$$

Using this, we get that

(6.5.8) 
$$U(\mathcal{P}, f, \alpha) \leq \sum_{j=1}^{l} M\left(\alpha(t_j) - \alpha(t_{j-1})\right) = M\left(\alpha(b) - \alpha(a)\right)$$

and

(6.5.9) 
$$L(\mathcal{P}, f, \alpha) \ge \sum_{j=1}^{l} m\left(\alpha(t_j) - \alpha(t_{j-1})\right) = m\left(\alpha(b) - \alpha(a)\right).$$

In particular,

(6.5.10) 
$$U(\mathcal{P}, f) \leq M (b-a)$$
  
and  
(6.5.11) 
$$L(\mathcal{P}, f) \geq m (b-a),$$

as on p121 of [192]. Note that equality holds in these four inequalities when  $\mathcal{P}$  is the partition of [a, b] consisting only of a and b, so that l = 1.

#### 6.5.2 Refinements of partitions

Let  $\mathcal{P}^* = \{t_k^*\}_{k=0}^{l^*}$  be another partition of [a, b]. We say that

(6.5.12) 
$$\mathcal{P}^*$$
 is a *refinement* of  $\mathcal{P}$ 

if for each j = 0, 1..., l there is a  $k \in \{0, 1, ..., l^*\}$  such that

(6.5.13) 
$$t_j = t_k^*$$
.

This means that  $[t_{j-1}, t_j]$  may be partitioned further by  $\mathcal{P}^*$  for each  $j = 1, \ldots, l$ . In this case, one can show that

(6.5.14) 
$$U(\mathcal{P}^*, f, \alpha) \le U(\mathcal{P}, f, \alpha)$$

and

(6.5.15) 
$$L(\mathcal{P}, f, \alpha) \le L(\mathcal{P}^*, f, \alpha),$$

as in Theorem 6.4 on p123 of [192].

# 6.6 Upper and lower integrals

Let us continue with the same notation and hypotheses as in the previous three sections. The *upper and lower Riemann integrals* of f on [a, b] are defined by

(6.6.1) 
$$\overline{\int}_{a}^{b} f \, dx = \inf_{\mathcal{P}} U(\mathcal{P}, f)$$

106

and

(6.6.2) 
$$\underbrace{\int_{-a}^{b} f \, dx}_{\mathcal{P}} = \sup_{\mathcal{P}} L(\mathcal{P}, f),$$

respectively, as on p121 of [192]. More precisely, this infimum and supremum are taken over all partitions  $\mathcal{P}$  of [a, b].

#### Some basic estimates for upper and lower integrals 6.6.1

Remember that

(6.6.3) 
$$m(b-a) \le L(\mathcal{P}, f) \le U(\mathcal{P}, f) \le M(b-a)$$

-b

ch

for all such partitions  $\mathcal{P}$ , as in (6.5.3), (6.5.10), and (6.5.11), where M and m are as in (6.5.4) and (6.5.5), respectively. This implies that the infimum and supremum in (6.6.1) and (6.6.2) exist in **R**, with

(6.6.4) 
$$m(b-a) \le \overline{\int}_{a}^{b} f \, dx$$

and

(6.6.5) 
$$\underline{\int}_{-a}^{b} f \, dx \le M \, (b-a).$$

We also have that

(6.6.6) 
$$\int_{-a}^{a} f \, dx \le M \, (b-a)$$
 and

(6.6.7) 
$$\int_{-a}^{a} f \, dx \ge m \, (b-a)$$

#### 6.6.2Upper and lower Riemann–Stieltjes integrals of f

Similarly, the upper and lower Riemann-Stieltjes integrals of f with respect to  $\alpha$  on [a, b] are defined by

(6.6.8) 
$$\overline{\int}_{a}^{b} f \, d\alpha = \inf_{\mathcal{P}} U(\mathcal{P}, f, \alpha)$$

and

(6.6.9) 
$$\underline{\int}_{-a}^{b} f \, d\alpha = \sup_{\mathcal{P}} L(\mathcal{P}, f, \alpha),$$

respectively, as on p122 of [192]. These are the same as in (6.6.1) and (6.6.2)when  $\alpha(x) = x$  on [a, b].

Note that

$$(6.6.10) \quad m(\alpha(b) - \alpha(a)) \le L(\mathcal{P}, f, \alpha) \le U(\mathcal{P}, f, \alpha) \le M(\alpha(b) - \alpha(a))$$

for all partitions  $\mathcal{P}$  of [a, b], as in (6.5.2), (6.5.8), and (6.5.9). This implies that the infimum and supremum in (6.6.8) and (6.6.9) exist in **R**, with

(6.6.11) 
$$m(\alpha(b) - \alpha(a)) \le \overline{\int}_{a}^{b} f \, d\alpha$$

and

(6.6.12) 
$$\underline{\int}_{-a}^{b} f \, d\alpha \leq M \, (\alpha(b) - \alpha(a)).$$

In addition,

(6.6.13) 
$$\overline{\int}_{a}^{b} f \, d\alpha \leq M \left( \alpha(b) - \alpha(a) \right)$$

and

(6.6.14) 
$$\int_{-a}^{b} f \, d\alpha \ge m \left(\alpha(b) - \alpha(a)\right)$$

#### 6.7 **Common refinements**

Let us continue with the same notation and hypotheses as in the previous four sections. Let  $\mathcal{P}_1$  and  $\mathcal{P}_2$  be any two partitions of [a, b], as in the previous section. It is not too difficult to see that there is a partition  $\mathcal{P}^*$  of [a, b] such that

(6.7.1) 
$$\mathcal{P}^*$$
 is a refinement of both  $\mathcal{P}_1, \mathcal{P}_2$ .

The smallest such *common refinement* of  $\mathcal{P}_1$  and  $\mathcal{P}_2$  can be obtained using all of the points in [a, b] that occur in  $\mathcal{P}_1$  and  $\mathcal{P}_2$ , listed in order.

#### 6.7.1Using common refinements

Under these conditions, we have that

(6.7.2) 
$$L(\mathcal{P}_1, f, \alpha) \le L(\mathcal{P}^*, f, \alpha) \le U(\mathcal{P}^*, f, \alpha) \le U(\mathcal{P}_2, f, \alpha).$$

This uses (6.5.15) in the first step, (6.5.2) in the second step, and (6.5.14) in the third step. One can use this to get that

(6.7.3) 
$$\underline{\int}_{a}^{b} f \, d\alpha \leq \overline{\int}_{a}^{b} f \, d\alpha,$$

as in Theorem 6.5 on p124 of [192]. In particular,

(6.7.4) 
$$\underline{\int}_{a}^{b} f \, dx \leq \overline{\int}_{a}^{b} f \, dx.$$

108

#### 6.7.2 Some remarks about upper and lower sums

If  $\mathcal{P} = \{t_j\}_{j=0}^l$  is a partition of [a, b], then

(6.7.5) 
$$U(\mathcal{P}, f, \alpha) - L(\mathcal{P}, f, \alpha) = \sum_{j=1}^{l} (M_j - m_j) \left( \alpha(t_j) - \alpha(t_{j-1}) \right),$$

by (6.4.2) and (6.4.3). Remember that  $M_j$  and  $m_j$  are as in (6.3.3) and (6.3.4), respectively, so that (6.7.6)  $M_j - m_j \ge 0$ 

for each j in particular.

#### 6.7.3 A property of refinements

Let  $\mathcal{P}$ ,  $\mathcal{P}^*$  be partitions of [a, b], and suppose that  $\mathcal{P}^*$  is a refinement of  $\mathcal{P}$ . Observe that

(6.7.7) 
$$U(\mathcal{P}^*, f, \alpha) - L(\mathcal{P}^*, f, \alpha) \le U(\mathcal{P}, f, \alpha) - L(\mathcal{P}, f, \alpha),$$

because of (6.5.14) and (6.5.15). This is related to part (a) of Theorem 6.7 on p125 of [192], and its proof.

#### 6.8 Riemann–Stieltjes integrals

We continue with the same notation and hypotheses as in the previous sections. If

(6.8.1) 
$$\underline{\int}_{a}^{b} f \, dx = \overline{\int}_{a}^{b} f \, dx,$$

then f is said to be *Riemann integrable* on [a, b], as on p121 of [192]. In this case, the *Riemann integral* of f on [a, b] is denoted

(6.8.2) 
$$\int_{a}^{b} f \, dx,$$

and defined to be the common value in (6.8.1). The Riemann integral may also be expressed as

(6.8.3) 
$$\int_{a}^{b} f(x) \, dx$$

#### 6.8.1 Riemann–Stieltjes integrability

Similarly, if

(6.8.4) 
$$\underline{\int}_{-a}^{b} f \, d\alpha = \overline{\int}_{-a}^{b} f \, d\alpha,$$

then f is said to be *Riemann–Stieltjes integrable* with respect to  $\alpha$  on [a, b], as on p122 of [192]. The corresponding *Riemann–Stieltjes integral* 

(6.8.5) 
$$\int_{a}^{b} f \, d\alpha$$

is defined to be the common value in (6.8.4), and this may also be expressed as

(6.8.6) 
$$\int_a^b f(x) \, d\alpha(x).$$

This is the same as in the previous paragraph when  $\alpha(x) = x$  on [a, b], as usual.

#### 6.8.2 A characterization of integrability

One can show that f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b] if and only if for every  $\epsilon > 0$  there is a partition  $\mathcal{P}$  of [a, b] such that

(6.8.7) 
$$U(\mathcal{P}, f, \alpha) - L(\mathcal{P}, f, \alpha) < \epsilon,$$

as in Theorem 6.6 on p124 of [192]. More precisely, it is somewhat easier to see that f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b] if and only if for every  $\epsilon > 0$  there are partitions  $\mathcal{P}_1$ ,  $\mathcal{P}_2$  of [a, b] such that

(6.8.8) 
$$U(\mathcal{P}_2, f, \alpha) - L(\mathcal{P}_1, f, \alpha) < \epsilon.$$

In order to get (6.8.7) from (6.8.8), one can take  $\mathcal{P}$  to be a common refinement of  $\mathcal{P}_1$  and  $\mathcal{P}_2$ , and use (6.5.14), (6.5.15).

## 6.9 Continuity and integrability

Let us continue with the same notation and hypotheses as in the previous sections. If

(6.9.1) 
$$f$$
 is continuous on  $[a, b]$ ,

then one can show that

(6.9.2) f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b],

as in Theorem 6.8 on p125 of [192]. This uses the fact that f is uniformly continuous on [a, b], because [a, b] is compact.

If c is a real number, then c may be considered as a constant function on [a, b], and it is easy to see that

(6.9.3) 
$$\int_{a}^{b} c \, d\alpha = c \left( \alpha(b) - \alpha(a) \right).$$

6.9.1 Some more criteria for integrability

If $(6.9.4)$	f is continuous at all but finitely many points in $[a, b]$ ,		
and if			
(6.9.5)	$\alpha$ is continuous at those points,		
then one can show that (6.9.2) holds, as in Theorem 6.10 on p126 of [192]. If			
(6.9.6)	f is monotonic on $[a, b]$ ,		
and			
(6.9.7)	$\alpha$ is continuous on $[a, b]$ ,		
1 (0.0			

then (6.9.2) holds, as in Theorem 6.9 on p126 of [192].

#### 6.9.2 A related example

If

(6.9.8)	f(x)	=	0	when $x \in [a, b]$ is irrational
		=	1	when $x \in [a, b]$ is rational,

\_\_\_\_h

then can check that

(6.9.9) 
$$\int_{-a}^{b} f \, d\alpha = \alpha(b) - \alpha(a)$$

and

(6.9.10) 
$$\underline{\int}_{a}^{b} f \, d\alpha = 0.$$

This implies that f is not Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b] when  $\alpha$  is not constant on [a, b]. In particular, f is not Riemann integrable on [a, b] when a < b, as in Exercise 4 on p138 of [192].

# 6.10 More on Riemann–Stieltjes integrals

We continue with the same notation and hypotheses as in the previous sections. Let I(x) be the unit step function on the real line, as in Section 4.12. Also let w be a real number with (6, 10, 1)

(6.10.1)	$a \leq w < b$ ,
and put	

(6.10.2)  $\alpha_w(x) = I(x-w)$ 

for  $x \in [a, b]$ , which is a monotonically increasing real-valued function on [a, b]. If

(6.10.3) f is continuous at w,

then (6.9.2) holds, with  $\alpha = \alpha_w$ , and

(6.10.4) 
$$\int_{a}^{b} f \, d\alpha_{w} = f(w).$$

This corresponds to Theorem 6.15 on p130 of [192], which is stated for  $w \neq a$ .

The argument for w = a is very similar, but if w = b, then  $\alpha_w \equiv 0$  on [a, b], because of the way that I(x) is defined. Some related results are mentioned in Exercise 3 on p138 of [192]. In particular, one can consider functions defined on **R** in the same way as I(x) when  $x \neq 0$ , and equal to 1 or 1/2 at x = 0.

#### 6.10.1 A remark about approximating integrals by sums

Suppose that (6.9.2) holds, and let  $\mathcal{P} = \{t_j\}_{j=0}^l$  be a partition of [a, b]. Of course,

(6.10.5) 
$$L(\mathcal{P}, f, \alpha) \le \int_{a}^{b} f \, d\alpha \le U(\mathcal{P}, f, \alpha),$$

by definition of the integral. If  $w_j \in [t_{j-1}, t_j]$ ,  $1 \leq j \leq l$ , then one can use (6.4.4) and (6.10.5) to get that

(6.10.6) 
$$\left|\sum_{j=1}^{l} f(w_j) \left(\alpha(t_j) - \alpha(t_{j-1})\right) - \int_a^b f \, d\alpha\right| \le U(\mathcal{P}, f, \alpha) - L(\mathcal{P}, f, \alpha).$$

In particular, if (6.8.7) holds for some  $\epsilon > 0$ , then

(6.10.7) 
$$\left|\sum_{j=1}^{l} f(w_j) \left(\alpha(t_j) - \alpha(t_{j-1})\right) - \int_a^b f \, d\alpha\right| < \epsilon.$$

This corresponds to part (c) of Theorem 6.7 on p125 of [192].

# 6.11 Multiplying functions by constants

Let us continue with the same notation and hypotheses as in the previous sections again. Let r be a positive real number, so that

(6.11.1) r f is a bounded real-valued function on [a, b].

If  $\mathcal{P}$  is a partition of [a, b], then one can check that

(6.11.2) 
$$U(\mathcal{P}, r f, \alpha) = r U(\mathcal{P}, f, \alpha)$$

and

(6.11.3) 
$$L(\mathcal{P}, rf, \alpha) = rL(\mathcal{P}, f, \alpha),$$

using some remarks in Subsection 6.1.1.

This implies that

(6.11.4) 
$$\overline{\int}_{a}^{b} r f \, d\alpha = r \overline{\int}_{a}^{b} f \, d\alpha$$

and

(6.11.5) 
$$\underline{\int}_{a}^{b} r f \, d\alpha = r \underline{\int}_{a}^{b} f \, d\alpha,$$

using the same remarks from Subsection 6.1.1. If f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then it follows that

(6.11.6) rf is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b]

too, with

(6.11.7) 
$$\int_{a}^{b} r f \, d\alpha = r \, \int_{a}^{b} f \, d\alpha,$$

as in part (a) of Theorem 6.12 on p128 of [192].

#### 6.11.1Multiplication by -1

Of course,

(6.11.8)-f is a bounded real-valued function on [a, b]

as well. If  $\mathcal{P}$  is a partition of [a, b], then one can verify that

(6.11.9) 
$$U(\mathcal{P}, -f, \alpha) = -L(\mathcal{P}, f, \alpha)$$

and

(6.11.10) 
$$L(\mathcal{P}, -f, \alpha) = -U(\mathcal{P}, f, \alpha),$$

using some other remarks in Subsection 6.1.1.

This means that

(6.11.11) 
$$\overline{\int}_{a}^{b} - f \, d\alpha = -\underline{\int}_{a}^{b} f \, d\alpha$$

and

(6.11.12) 
$$\underline{\int}_{a}^{b} - f \, d\alpha = -\overline{\int}_{a}^{b} f \, d\alpha,$$

using these same remarks from Subsection 6.1.1. If f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then we get that

(6.11.13) -f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b],

with

(6.11.14) 
$$\int_{a}^{b} -f \, d\alpha = -\int_{a}^{b} f \, d\alpha.$$

#### **6.11.2** Multiplying $\alpha$ by r > 0

Let r be a positive real number again, and note that

(6.11.15)  $r \alpha$  is a monotonically increasing real-valued function on [a, b].

If  $\mathcal{P}$  is a partition of [a, b], then it is easy to see that

(6.11.16) 
$$U(\mathcal{P}, f, r \alpha) = r U(\mathcal{P}, f, \alpha)$$

and

(6.11.17) 
$$L(\mathcal{P}, f, r \alpha) = r L(\mathcal{P}, f, \alpha)$$

This implies that

(6.11.18) 
$$\overline{\int}_{a}^{b} f d(r \alpha) = r \overline{\int}_{a}^{b} f d\alpha$$

and

(6.11.19) 
$$\int_{-a}^{b} f d(r \alpha) = r \int_{-a}^{b} f d\alpha$$

using some remarks from Subsection 6.1.1 again. If f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then we get that

(6.11.20) f is Riemann–Stieltjes integrable with respect to  $r \alpha$  on [a, b],

with

(6.11.21) 
$$\int_{a}^{b} f d(r \alpha) = r \int_{a}^{b} f d\alpha$$

This corresponds to part of part (e) of Theorem 6.12 on p128 of [192].

### 6.12 Adding two functions

We continue with the same notation and hypotheses as in the previous sections. Let g be another bounded real-valued function on [a, b], so that

(6.12.1) 
$$f + g$$
 is bounded on  $[a, b]$ 

as well. If  $\mathcal{P}$  is a partition of [a, b], then one can check that

(6.12.2) 
$$U(\mathcal{P}, f+g, \alpha) \le U(\mathcal{P}, f, \alpha) + U(\mathcal{P}, g, \alpha)$$

and

(6.12.3) 
$$L(\mathcal{P}, f+g, \alpha) \ge L(\mathcal{P}, f, \alpha) + L(\mathcal{P}, g, \alpha)$$

using the remarks about suprema and infima of sums of functions in Subsection 6.1.2.

Let  $\mathcal{P}_1$  and  $\mathcal{P}_2$  be arbitrary partitions of [a, b], and let  $\mathcal{P}^*$  be a common refinement of them, as in Section 6.7. Observe that

$$(6.12.4) \overline{\int}_{a}^{b} (f+g) \, d\alpha \leq U(\mathcal{P}^{*}, f+g, \alpha) \leq U(\mathcal{P}^{*}, f, \alpha) + U(\mathcal{P}^{*}, g, \alpha)$$
$$\leq U(\mathcal{P}_{1}, f, \alpha) + U(\mathcal{P}_{2}, g, \alpha),$$

where the first step is as in Subsection 6.6.2, the second step is as in (6.12.2), and the third step is as in Subsection 6.5.2. One can use this to get that

(6.12.5) 
$$\overline{\int}_{a}^{b} (f+g) \, d\alpha \leq \overline{\int}_{a}^{b} f \, d\alpha + \overline{\int}_{a}^{b} g \, d\alpha.$$

Similarly,

$$(6.12.6) \frac{\int_{-a}^{b} (f+g) \, d\alpha}{\geq} L(\mathcal{P}^*, f+g, \alpha) \geq L(\mathcal{P}^*, f, \alpha) + L(\mathcal{P}^*, g, \alpha)$$
$$\geq L(\mathcal{P}_1, f, \alpha) + L(\mathcal{P}_2, g, \alpha).$$

This implies that

(6.12.7) 
$$\underline{\int}_{a}^{b} (f+g) \, d\alpha \ge \underline{\int}_{a}^{b} f \, d\alpha + \underline{\int}_{a}^{b} g \, d\alpha.$$

#### 6.12.1 Sums of integrable functions

If f and g are Riemann–Stieltjes integrable with respect to  $\alpha$  on [a,b], then it follows that

(6.12.8) f + g is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b],

with

(6.12.9) 
$$\int_{a}^{b} (f+g) \, d\alpha = \int_{a}^{b} f \, d\alpha + \int_{a}^{b} g \, d\alpha.$$

This corresponds to another part of part (a) of Theorem 6.12 on p128 of [192].

#### 6.12.2 Some related results

Let f and g be any two bounded real-valued functions on [a, b] again. If  $\mathcal{P}$  is a partition of [a, b], then one can verify that

(6.12.10) 
$$U(\mathcal{P}, f+g, \alpha) \ge U(\mathcal{P}, f, \alpha) + L(\mathcal{P}, g, \alpha)$$

and

(6.12.11) 
$$L(\mathcal{P}, f+g, \alpha) \le L(\mathcal{P}, f, \alpha) + U(\mathcal{P}, g, \alpha),$$

using the remarks at the beginning of Section 6.2. These inequalities can also be obtained from those mentioned at the beginning of the section using some of the remarks in the previous section. One can use these inequalities to get that

(6.12.12) 
$$\overline{\int}_{a}^{b} (f+g) \, d\alpha \ge \overline{\int}_{a}^{b} f \, d\alpha + \underline{\int}_{a}^{b} g \, d\alpha$$

and

(6.12.13) 
$$\underline{\int}_{a}^{b} (f+g) \, d\alpha \leq \underline{\int}_{a}^{b} f \, d\alpha + \overline{\int}_{a}^{b} g \, d\alpha,$$

as before. This could also be obtained from the analogous inequalities mentioned earlier and some of the remarks in the previous section.

If g is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then we get that -b -b ,

(6.12.14) 
$$\overline{\int}_{a}^{b} (f+g) \, d\alpha = \overline{\int}_{a}^{b} f \, d\alpha + \int_{a}^{b} g \, d\alpha$$

and

(6.12.15) 
$$\underline{\int}_{a}^{b} (f+g) \, d\alpha = \underline{\int}_{a}^{b} f \, d\alpha + \int_{a}^{b} g \, d\alpha.$$

#### Another monotonically increasing function 6.13

We continue with the same notation and hypotheses as in the previous sections, starting in Section 6.3. Let  $\beta$  be another monotonically increasing real-valued function on [a, b], so that

<u>\_</u>b

<u>-</u>b

(6.13.1) 
$$\alpha + \beta$$
 is monotonically increasing on  $[a, b]$ 

too. If  $\mathcal{P}$  is a partition of [a, b], then it is easy to see that

(6.13.2) 
$$U(\mathcal{P}, f, \alpha + \beta) = U(\mathcal{P}, f, \alpha) + U(\mathcal{P}, f, \beta)$$

and

(6.13.3) 
$$L(\mathcal{P}, f, \alpha + \beta) = L(\mathcal{P}, f, \alpha) + L(\mathcal{P}, f, \beta).$$

This implies that

(6.13.4) 
$$U(\mathcal{P}, f, \alpha + \beta) \ge \overline{\int}_{a}^{b} f \, d\alpha + \overline{\int}_{a}^{b} f \, d\beta$$

and

(6.13.5) 
$$L(\mathcal{P}, f, \alpha + \beta) \leq \underline{\int_{a}^{b} f \, d\alpha} + \underline{\int_{a}^{b} f \, d\beta}$$

It follows that

(6.13.6) 
$$\overline{\int}_{a}^{b} f \, d(\alpha + \beta) \ge \overline{\int}_{a}^{b} f \, d\alpha + \overline{\int}_{a}^{b} f \, d\beta$$
and

(6.13.7) 
$$\underline{\int}_{a}^{b} f \, d(\alpha + \beta) \leq \underline{\int}_{a}^{b} f \, d\alpha + \underline{\int}_{a}^{b} f \, d\beta.$$

#### 6.13.1 More on these upper and lower integrals

Let  $\mathcal{P}_1$  and  $\mathcal{P}_2$  be arbitrary partitions of [a, b], and let  $\mathcal{P}^*$  be a common refinement of them. Note that

$$(6.13.8) \overline{\int}_{a}^{b} f \, d(\alpha + \beta) \leq U(\mathcal{P}^{*}, f, \alpha + \beta) = U(\mathcal{P}^{*}, f, \alpha) + U(\mathcal{P}^{*}, f, \beta)$$
$$\leq U(\mathcal{P}_{1}, f, \alpha) + U(\mathcal{P}_{2}, f, \beta),$$

which implies that

(6.13.9) 
$$\overline{\int}_{a}^{b} f \, d(\alpha + \beta) \leq \overline{\int}_{a}^{b} f \, d\alpha + \overline{\int}_{a}^{b} f \, d\beta.$$

Combining this with (6.13.6) we obtain that

(6.13.10) 
$$\overline{\int}_{a}^{b} f \, d(\alpha + \beta) = \overline{\int}_{a}^{b} f \, d\alpha + \overline{\int}_{a}^{b} f \, d\beta.$$

Similarly,

$$\underbrace{\int_{a}^{b} f d(\alpha + \beta)}_{a} \geq L(\mathcal{P}^{*}, f, \alpha + \beta) = L(\mathcal{P}^{*}, f, \alpha)L(\mathcal{P}^{*}, f, \beta) \\ \geq L(\mathcal{P}_{1}, f, \alpha) + L(\mathcal{P}_{2}, f, \beta),$$

(6.13.11) and thus (6.13.12)

$$\underbrace{\int_{-a}^{b} f \, d(\alpha + \beta)}_{a} \geq \underbrace{\int_{-a}^{b} f \, d\alpha}_{a} + \underbrace{\int_{-a}^{b} f \, d\beta}_{a}.$$

This means that

$$\underline{\int}_{a}^{b} f \, d(\alpha + \beta) = \underline{\int}_{a}^{b} f \, d\alpha + \underline{\int}_{a}^{b} f \, d\beta,$$

because of (6.13.7).

#### 6.13.2 Integrability with respect to $\alpha + \beta$

One can use (6.13.10) and (6.13.13) to get that

(6.13.14) f is Riemann–Stieltjes integrable with respect to  $\alpha+\beta$  on [a,b] if and only if

(6.13.15) 
$$f$$
 is Riemann–Stieltjes integrable with respect to  
both  $\alpha$  and  $\beta$  on  $[a, b]$ .

In this case, we have that

(6.13.16) 
$$\int_{a}^{b} f d(\alpha + \beta) = \int_{a}^{b} f d\alpha + \int_{a}^{b} f d\beta,$$

as in part (e) of Theorem 6.12 on p128 of [192].

# Chapter 7

# More on differentiation and integration

#### Some inequalities with integrals 7.1

Let us continue with the same notation and hypotheses as in the previous chapter, starting in Section 6.3. Let g be another bounded real-valued function on [a, b] again, and suppose that

$$(7.1.1) f(x) \le g(x)$$

for every  $x \in [a, b]$ . If  $\mathcal{P}$  is a partition of [a, b], then it is easy to see that

(7.1.2) 
$$U(\mathcal{P}, f, \alpha) \le U(\mathcal{P}, g, \alpha)$$

and (7.1.3)

$$L(\mathcal{P}, f, \alpha) \leq L(\mathcal{P}, g, \alpha).$$

This implies that

(7.1.4) 
$$\overline{\int}_{a}^{b} f \, d\alpha \leq U(\mathcal{P}, g, \alpha)$$

and

(7.1.5) 
$$L(\mathcal{P}, f, \alpha) \leq \underline{\int}_{a}^{b} g \, d\alpha,$$

by definition of the upper and lower integrals, as in Subsection 6.6.2. It follows that L ,

(7.1.6) 
$$\overline{\int}_{a}^{b} f \, d\alpha \leq \overline{\int}_{a}^{b} g \, d\alpha$$

and

(7.1.7) 
$$\underline{\int}_{a}^{b} f \, d\alpha \leq \underline{\int}_{a}^{b} g \, d\alpha,$$

118

#### 7.2. SUBINTERVALS OF [A, B]

because  $\mathcal{P}$  is arbitrary. If f and g are Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then we get that

(7.1.8) 
$$\int_{a}^{b} f \, d\alpha \le \int_{a}^{b} g \, d\alpha$$

This corresponds to part (b) of Theorem 6.12 on p128 of [192].

#### 7.1.1 A simple estimate for the absolute value of the integral

If f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then

(7.1.9) 
$$\left(\inf_{a \le x \le b} f(x)\right) \left(\alpha(b) - \alpha(a)\right) \le \int_{a}^{b} f \, d\alpha \le \left(\sup_{a \le x \le b} f(x)\right) \left(\alpha(b) - \alpha(a)\right)$$

as in Subsection 6.6.2. The first inequality implies that

(7.1.10) 
$$-\int_{a}^{b} f \, d\alpha \leq -\left(\inf_{a \leq x \leq b} f(x)\right) (\alpha(b) - \alpha(a))$$
$$= \left(\sup_{a \leq x \leq b} (-f(x))\right) (\alpha(b) - \alpha(a)),$$

where the second step is as in Subsection 6.1.1.

One can check that

(7.1.11) 
$$\sup_{a \le x \le b} |f(x)| = \max\left(\sup_{a \le x \le b} f(x), \sup_{a \le x \le b} (-f(x))\right).$$

It follows that

(7.1.12) 
$$\left| \int_{a}^{b} f \, d\alpha \right| \leq \left( \sup_{a \leq x \leq b} |f(x)| \right) (\alpha(b) - \alpha(a)).$$

This corresponds to part (d) of Theorem 6.12 on p128 of [192].

# 7.2 Subintervals of [a, b]

We continue with the same notation and hypotheses as in the previous sections again, starting in Section 6.3. Let c be a real number with

$$(7.2.1) a \le c \le b,$$

so that [a, c] and [c, b] are closed intervals in **R** with

(7.2.2) 
$$[a,c] \cup [c,b] = [a,b].$$

If  $\mathcal{P}_1$  and  $\mathcal{P}_2$  are partitions of [a, c] and [c, b], respectively, then we can combine them to get a partition  $\mathcal{P}_{1,2}$  of [a, b]. Of course, the restrictions of

f to [a, c] and [c, b] define bounded real-valued functions on the two intervals. Similarly, the restrictions of  $\alpha$  to [a, c] and [c, b] define monotonically increasing real-valued functions on these intervals. It is easy to see that

(7.2.3) 
$$U(\mathcal{P}_{1,2}, f, \alpha) = U(\mathcal{P}_1, f, \alpha) + U(\mathcal{P}_2, f, \alpha)$$

and

(7.2.4) 
$$L(\mathcal{P}_{1,2}, f, \alpha) = L(\mathcal{P}_1, f, \alpha) + L(\mathcal{P}_2, f, \alpha).$$

This implies that

(7.2.5) 
$$\overline{\int}_{a}^{b} f \, d\alpha \leq U(\mathcal{P}_{1}, f, \alpha) + U(\mathcal{P}_{2}, f, \alpha)$$

and

(7.2.6) 
$$\underline{\int_{a}^{b} f \, d\alpha} \ge L(\mathcal{P}_{1}, f, \alpha) + L(\mathcal{P}_{2}, f, \alpha)$$

One can use this to get that

(7.2.7) 
$$\overline{\int}_{a}^{b} f \, d\alpha \leq \overline{\int}_{a}^{c} f \, d\alpha + \overline{\int}_{c}^{b} f \, d\alpha$$

and

(7.2.8) 
$$\underline{\int}_{a}^{b} f \, d\alpha \ge \underline{\int}_{a}^{c} f \, d\alpha + \underline{\int}_{c}^{b} f \, d\alpha.$$

#### 7.2.1 Starting with an arbitrary partition of [a, b]

Let  $\mathcal{P}$  be any partition of [a, b], and let  $\mathcal{P}^*$  be a partition of [a, b] that is a refinement of  $\mathcal{P}$  that contains c, which can be obtained by adding c to  $\mathcal{P}$ , if necessary. Using  $\mathcal{P}^*$ , we get partitions  $\mathcal{P}_1^*$  and  $\mathcal{P}_2^*$  of [a, c] and [c, b], respectively. Observe that

(7.2.9) 
$$U(\mathcal{P}, f, \alpha) \ge U(\mathcal{P}^*, f, \alpha) = U(\mathcal{P}_1^*, f, \alpha) + U(\mathcal{P}_2^*, f, \alpha)$$
$$\ge \int_a^c f \, d\alpha + \int_c^b f \, d\alpha,$$

where the first step is as in Subsection 6.5.2, the second step is as in (7.2.3), and the third step is as in Subsection 6.6.2. This implies that

(7.2.10) 
$$\overline{\int}_{a}^{b} f \, d\alpha \ge \overline{\int}_{a}^{c} f \, d\alpha + \overline{\int}_{c}^{b} f \, d\alpha,$$

so that

(7.2.11) 
$$\overline{\int}_{a}^{b} f \, d\alpha = \overline{\int}_{a}^{c} f \, d\alpha + \overline{\int}_{c}^{b} f \, d\alpha.$$

Similarly,

(7.2.12) 
$$L(\mathcal{P}, f, \alpha) \leq L(\mathcal{P}^*, f, \alpha) = L(\mathcal{P}_1^*, f, \alpha) + L(\mathcal{P}_2^*, f, \alpha)$$
$$\leq \underbrace{\int_{-a}^{c} f \, d\alpha}_{a} + \underbrace{\int_{-c}^{b} f \, d\alpha}_{c},$$

which implies that

(7.2.13) 
$$\underline{\int}_{-a}^{b} f \, d\alpha \leq \underline{\int}_{-a}^{c} f \, d\alpha + \underline{\int}_{-c}^{b} f \, d\alpha,$$

and thus

(7.2.14) 
$$\underline{\int}_{a}^{b} f \, d\alpha = \underline{\int}_{a}^{c} f \, d\alpha + \underline{\int}_{c}^{b} f \, d\alpha$$

These statements could also be obtained from the previous ones using the remarks in Section 6.11.

#### 7.2.2 Integrability on [a, b]

One can use (7.2.11) and (7.2.14) to get that f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b] if and only if

(7.2.15) the restrictions of f to [a, c] and [c, b]are Riemann–Stieltjes integrable with respect to  $\alpha$ .

In this case, we have that

(7.2.16) 
$$\int_{a}^{b} f \, d\alpha = \int_{a}^{c} f \, d\alpha + \int_{c}^{b} f \, d\alpha,$$

as in part (c) of Theorem 6.12 on p128 of [192].

#### **7.2.3** Integrability on subintervals of [a, b]

If f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then one can repeat the argument in the preceding paragraph to get that

(7.2.17) the restriction of f to any closed subinterval of [a, b]is Riemann–Stieltjes integrable with respect to  $\alpha$ 

as well.

# 7.3 Derivatives of integrals

We continue with the same notation and hypotheses as before, starting in Section 6.3. Suppose that f is Riemann–Stieltjes integrable with respect to  $\alpha$  on

[a, b]. If  $x \in [a, b]$ , then the restriction of f to [a, x] is Riemann–Stieltjes integrable with respect to  $\alpha$ , as in the previous section, and we put

(7.3.1) 
$$F(x) = \int_a^x f \, d\alpha.$$

Similarly, if  $y \in [x, b]$ , then the restrictions of f to [a, y] and [x, y] are Riemann–Stieltjes integrable with respect to  $\alpha$ , and

(7.3.2) 
$$F(y) - F(x) = \int_{x}^{y} f \, d\alpha,$$

as in the previous section.

Observe that

(7.3.3) 
$$|F(y) - F(x)| = \left| \int_{x}^{y} f \, d\alpha \right| \leq \left( \sup_{x \le v \le y} |f(v)| \right) (\alpha(y) - \alpha(x)) \\ \leq \left( \sup_{a \le v \le b} |f(v)| \right) (\alpha(y) - \alpha(x)),$$

where the second step is as in Subsection 7.1.1. One can use this to get that

(7.3.4) F is continuous at any point in [a, b] at which  $\alpha$  is continuous.

This basically corresponds to the first part of Theorem 6.20 on p133 of [192].

#### 7.3.1 Using the Riemann integral

Suppose from now on in this section that a < b, and that  $\alpha(u) = u$  on [a, b], so that f is Riemann integrable on [a, b]. Put

(7.3.5) 
$$F(x) = \int_{a}^{x} f(t) dt$$

for each  $x \in [a, b]$ , as before. If  $y \in [x, b]$ , then

(7.3.6) 
$$F(y) - F(x) = \int_{x}^{y} f(t) dt,$$

as before. If

(7.3.7) 
$$f$$
 is continuous at a point  $w \in [a, b]$ ,

then it is well known and not too difficult to show that

(7.3.8) 
$$F$$
 is differentiable at  $w$ ,

with 
$$(7.3.9)$$

(3.9) 
$$F'(w) = f(w).$$

#### 7.4. INTEGRALS OF DERIVATIVES

This is the second part of Theorem 6.20 on p133 of [192]. More precisely, suppose first that

$$(7.3.10) a \le w < b,$$

and let h > 0 be given, with  $h \le b - w$ . Observe that

(7.3.11) 
$$\frac{F(w+h) - F(w)}{h} = \frac{1}{h} \int_{w}^{w+h} f(t) dt,$$

because of (7.3.6). If f is continuous at w, then one can check that

(7.3.12) 
$$\lim_{h \to 0+} \frac{F(w+h) - F(x)}{h} = f(w)$$

Of course, if f is constant on [a, b], then (7.3.11) is the same as the constant value of f. One can use this to reduce to the case where f(w) = 0, which is a bit simpler.

Similarly, suppose now that

$$(7.3.13) a < w \le b,$$

and let h > 0 be given, with  $h \le w - a$ . In this case,

(7.3.14) 
$$\frac{F(w) - F(w-h)}{h} = \frac{1}{h} \int_{w-h}^{w} f(t) dt$$

using (7.3.6) again. If f is continuous at w, then one can verify that

(7.3.15) 
$$\lim_{h \to 0+} \frac{F(w) - F(w - h)}{h} = f(w),$$

as before.

# 7.4 Integrals of derivatives

Let a and b be real numbers with a < b, and let  $F_1$  be a real-valued function on [a, b]. Suppose that

(7.4.1)  $F_1$  is differentiable at every point in [a, b],

and put

(7.4.2) 
$$f = F_1'$$

on [a, b]. If (7.4.3) f is Riemann integrable on [a, b],

then Theorem 6.21 on p134 of [192] says that

(7.4.4) 
$$\int_{a}^{b} f(x) \, dx = F_1(b) - F_1(a).$$

This is a version of the fundamental theorem of calculus. Of course, (7.4.3) includes the condition that f be bounded on [a, b].

#### **7.4.1** The case where f is continuous on [a, b]

If f is continuous on [a, b], and F is as in (7.3.5), then F is differentiable at every point in [a, b], as in the previous section, and we have that

(7.4.5) 
$$(F_1 - F)' = F_1' - F' = 0$$

on [a, b]. This means that  $F_1 - F$  is constant on [a, b], by the mean value theorem. In this case, (7.4.4) follows from the definition of F.

#### 7.4.2 When the derivative is Riemann integrable on [a, b]

Otherwise, let  $\mathcal{P} = \{t_j\}_{j=0}^l$  be a partition of [a, b], and note that

(7.4.6) 
$$\sum_{j=1}^{l} (F_1(t_j) - F_1(t_{j-1})) = F_1(b) - F_1(a),$$

as in (6.4.5). The mean value theorem implies that for each j = 1, ..., l there is an  $r_j \in [t_{j-1}, t_j]$  such that

(7.4.7) 
$$F_1(t_j) - F_1(t_{j-1}) = F'_1(r_j) (t_j - t_{j-1}).$$

This means that

(7.4.8) 
$$\sum_{j=1}^{l} F_1'(r_j) \left( t_j - t_{j-1} \right) = F_1(b) - F_1(a),$$

by (7.4.6).

If f is bounded on [a, b], then

(7.4.9) 
$$L(\mathcal{P}, f) \le \sum_{j=1}^{l} F_1'(r_j) \left( t_j - t_{j-1} \right) \le U(\mathcal{P}, f),$$

by definition of the upper and lower Darboux sums associated to f and  $\mathcal{P}$ , as in Subsection 6.3.1. Thus

(7.4.10) 
$$L(\mathcal{P}, f) \le F_1(b) - F_1(a) \le U(\mathcal{P}, f),$$

by (7.4.8). It follows that

(7.4.11) 
$$\underline{\int}_{a}^{b} f(x) \, dx \leq F_1(b) - F_1(a) \leq \overline{\int}_{a}^{b} f(x) \, dx,$$

by definition of the upper and lower integrals, as in Section 6.6. If (7.4.3) holds, then we get that (7.4.4) holds.

#### 7.5 More on Riemann–Stieltjes integrability

Let us continue with the same notation and hypotheses as before, starting in Section 6.3. Let  $a_1$ ,  $b_1$  be real numbers with  $a_1 \leq b_1$ , and suppose that

$$(7.5.1) a_1 \le f \le b_1$$

on [a, b]. Also let  $\phi$  be a continuous real-valued function on  $[a_1, b_1]$ , and note that  $\phi$  is bounded on  $[a_1, b_1]$ , because  $[a_1, b_1]$  is compact. Thus  $\phi \circ f$  is a bounded real-valued function on [a, b].

#### **7.5.1** Integrability of $\phi \circ f$

If f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then

(7.5.2)  $\phi \circ f$  is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b]

too. This is Theorem 6.11 on p127 of [192]. This uses the fact that  $\phi$  is uniformly continuous on  $[a_1, b_1]$ , because  $[a_1, b_1]$  is compact. Basically, one can show that

(7.5.3) 
$$U(\mathcal{P},\phi\circ f,\alpha) - L(\mathcal{P},\phi\circ f,\alpha)$$

is arbitrarily small when

(7.5.4) 
$$U(\mathcal{P}, f, \alpha) - L(\mathcal{P}, f, \alpha)$$

is sufficiently small.

#### 7.5.2 Integrability of products

In particular, if f is a bounded real-valued function on [a, b] that is Riemann–Stieltjes integrable with respect to  $\alpha$ , then

(7.5.5)  $f^2$  is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b].

Let g be another bounded real-valued function on [a, b] that is Riemann–Stieltjes integrable with respect to  $\alpha$ . Under these conditions, part (a) of Theorem 6.13 on p129 of [192] says that

(7.5.6) fg is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b].

This uses the fact that

(7.5.7) 
$$fg = (1/2)\left((f+g)^2 - f^2 - g^2\right).$$

#### Integrability of |f|7.5.3

Similarly,

(7.5.8)|f| is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b],

as in part (b) of Theorem 6.13 on p129 of [192]. Note that

(7.5.9) 
$$\int_{a}^{b} f \, d\alpha, - \int_{a}^{b} f \, d\alpha \leq \int_{a}^{b} |f| \, d\alpha,$$

because  $f, -f \leq |f|$  on [a, b]. This means that

(7.5.10) 
$$\left| \int_{a}^{b} f \, d\alpha \right| \leq \int_{a}^{b} |f| \, d\alpha,$$

as in [192].

#### 7.6 Differentiability of $\alpha$

Let us continue with the same notation and hypotheses as in the previous sections. Suppose that a < b,

(7.6.1) $\alpha$  is differentiable at every point in [a, b],

 $\alpha'$  is bounded on [a, b], and

(7.6.2) $\alpha'$  is Riemann integrable on [a, b].

Under these conditions, Theorem 6.17 on 131 of [192] states that a bounded real-valued function f on [a, b] is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b] if and only if

(7.6.3) 
$$f \alpha'$$
 is Riemann integrable on  $[a, b]$ ,

in which case

(7.6.4) 
$$\int_{a}^{b} f \, d\alpha = \int_{a}^{b} f(x) \, \alpha'(x) \, dx.$$

Note that (7.6.3) holds when f is Riemann integrable on [a, b], because of (7.6.2), as in the previous section.

#### 7.7Changing variables

Let  $a_0$  and  $b_0$  be real numbers with  $a_0 \leq b_0$ , and let  $\phi$  be a continuous realvalued function on  $[a_0, b_0]$  that is also monotonically increasing on  $[a_0, b_0]$ . If y, z are real numbers with  $a_0 \leq y \leq z \leq b_0$ , then

(7.7.1) 
$$\phi([y,z]) = [\phi(y),\phi(z)].$$

More precisely,

(7.7.2) 
$$\phi([y,z]) \subseteq [\phi(y),\phi(z)],$$

because  $\phi$  is monotonically increasing on  $[a_0, b_0]$ , and

 $[\phi(y), \phi(z)] \subseteq \phi([y, z]),$ (7.7.3)

by the intermediate value theorem.

#### 7.7.1Compositions with $\phi$

Put  $a = \phi(a_0)$  and  $b = \phi(b_0)$ , so that  $a \leq b$ . Let  $\alpha$  be a monotonically increasing real-valued function on [a, b], as usual, and put

$$(7.7.4) \qquad \qquad \beta = \alpha \circ \phi.$$

It is easy to see that this is a monotonically increasing real-valued function on  $[a_0, b_0]$ . Also let f be a bounded real-valued function on [a, b], as before, so that

(7.7.5) $g = f \circ \phi$ 

is a bounded real-valued function on  $[a_0, b_0]$ .

#### 7.7.2 Partitions of $[a_0, b_0]$

Let  $\mathcal{P}_0 = \{t_{0,j}\}_{i=0}^{l_0}$  be a partition of  $[a_0, b_0]$ . Note that

(7.7.6) 
$$\mathcal{P}_0^{\phi} = \{\phi(t_{0,j})\}_{j=0}^{l_0}$$

is a partition of [a, b]. One can check that

(7.7.7) 
$$U(\mathcal{P}_0, g, \beta) = U(\mathcal{P}_0^{\phi}, f, \alpha)$$

and

(7.7.8) 
$$L(\mathcal{P}_0, g, \beta) = L(\mathcal{P}_0^{\phi}, f, \alpha),$$

as in (38) on p133 of [192]. This uses the fact that

(7.7.9) 
$$\phi([t_{0,j-1}, t_{0,j}]) = [\phi(t_{0,j-1}), \phi(t_{0,j})]$$

for each  $j = 1, \ldots, l_0$ , as in (7.7.1).

#### Partitions of [a, b] 7.7.3

Observe that every partition of [a, b] is of the form (7.7.6) for some partition  $\mathcal{P}_0$ of  $[a_0, b_0]$ . One can use this and (7.7.7), (7.7.8) to get that

(7.7.10) 
$$\overline{\int}_{a_0}^{b_0} g \, d\beta = \overline{\int}_a^b f \, d\alpha$$

and

(7.7.11) 
$$\underline{\int}_{a_0}^{b_0} g \, d\beta = \underline{\int}_{a}^{b} f \, d\alpha.$$

#### 7.7.4 Integrability on $[a_0, b_0]$

In particular, if f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b], then

(7.7.12) g is Riemann–Stieltjes integrable with respect to  $\beta$  on  $[a_0, b_0]$ ,

with

(7.7.13) 
$$\int_{a_0}^{b_0} g \, d\beta = \int_a^b f \, d\alpha,$$

as in Theorem 6.19 on p132 of [192]. Although this theorem is stated for the case where  $\phi$  is strictly increasing on  $[a_0, b_0]$ , we do not really need this here. If  $\phi$  is strictly increasing on  $[a_0, b_0]$ , then we get a one-to-one correspondence between partitions of  $[a_0, b_0]$  and [a, b], but this is not really needed.

#### 7.7.5 Some additional remarks

Of course, if

(7.7.14)  $\phi(y) = \phi(z)$ 

for some  $y, z \in [a_0, b_0]$  with y < z, then  $\phi$  is constant on [y, z], because  $\phi$  is monotonically increasing on  $[a_0, b_0]$ . This implies that g and  $\beta$  are constant on [y, z] as well.

More precisely, the previous argument shows that (7.7.12) holds if and only if f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b]. If  $\phi$  is strictly increasing on  $[a_0, b_0]$ , then one can also look at the "only if" part in terms of the inverse of  $\phi$ .

#### 7.8 Some sums of step functions

We continue with the same notation and hypotheses as before, starting in Section 6.3 again. Let I(x) be the unit step function on the real line, as in Section 4.12, and let  $\{w_j\}_{j=1}^{\infty}$  be a sequence of elements of (a, b). Also let  $\{c_j\}_{j=1}^{\infty}$  be an infinite sequence of nonnegative real numbers such that

(7.8.1) 
$$\sum_{j=1}^{\infty} c_j \text{ converges},$$

and put

(7.8.2) 
$$\alpha(x) = \sum_{j=1}^{\infty} c_j I(x - w_j)$$

for each  $x \in [a, b]$ , where the series on the right converges by the comparison test. This defines a monotonically increasing real-valued function on [a, b].

#### 7.8.1 Riemann–Stieltjes integrals with respect to $\alpha$

If f is continuous on [a, b], then Theorem 6.16 on p130 of [192] states that

(7.8.3) 
$$\int_a^b f \, d\alpha = \sum_{j=1}^\infty c_j f(w_j).$$

Note that the series on the right converges absolutely, by the comparison test, because f is bounded on [a, b]. More precisely, in [192], the  $w_j$ 's are asked to be distinct, but this does not really seem to be needed, although there may not be any reason to consider repetitions. One could permit  $w_j$  to be equal to a for some j, but not b, because of the way that I(x) is defined, as in Section 6.10.

One could use analogous arguments to get that f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b] when

(7.8.4) 
$$f$$
 is continuous at  $w_j$  for each  $j$ ,

in addition to being bounded on [a, b], and that (7.8.3) holds in this case.

#### 7.8.2 Some related sums

Let n be a positive integer, and put

(7.8.5) 
$$\alpha_n(x) = \sum_{j=1}^n c_j I(x - w_j)$$

for each  $x \in [a, b]$ . This is a monotonically increasing real-valued function on [a, b]. If f is continuous at each of  $w_1, \ldots, w_n$ , then it is easy to see that f is Riemann–Stieltjes integrable with respect to  $\alpha_n$  on [a, b], with

(7.8.6) 
$$\int_{a}^{b} f \, d\alpha_{n} = \sum_{j=1}^{n} c_{j} f(w_{j}),$$

using remarks in Section 6.10 and Subsections 6.11.2 and 6.13.2. Similarly,

(7.8.7) 
$$\beta_n(x) = \sum_{j=n+1}^{\infty} c_j I(x - w_j)$$

defines a monotonically increasing real-valued function on [a, b], with

(7.8.8) 
$$\alpha_n(x) + \beta_n(x) = \alpha(x)$$

for each  $x \in [a, b]$ . Observe that

(7.8.9) 
$$\overline{\int}_{a}^{b} f \, d\beta_n \le M \left(\beta_n(b) - \beta_n(a)\right) = M \sum_{j=n+1}^{\infty} c_j$$

and

(7.8.10) 
$$\int_{-a}^{b} f \, d\beta_n \ge m \left(\beta_n(b) - \beta_n(a)\right) = m \sum_{j=n+1}^{\infty} c_j$$

where the first steps are as in Subsection 6.6.2, and the second steps follow from the definition (7.8.7) of  $\beta_n$ .

Of course, the right sides of (7.8.9) and (7.8.10) tend to 0 as  $n \to \infty$ , because  $\sum_{j=1}^{\infty} c_j$  converges, by hypothesis. To get the conclusions mentioned earlier, one can use (7.8.8) and the remarks in Subsection 6.13.2.

# 7.9 Limits of integrals

Let a and b be real numbers with a < b, and let  $\{f_j\}_{j=1}^{\infty}$  be a sequence of bounded real-valued functions on [a, b] that converges uniformly to a real-valued function f on [a, b]. Note that

(7.9.1) 
$$f$$
 is bounded on  $[a, b]$ 

as well, as mentioned in Subsection 4.9.3. Let  $\alpha$  be a monotonically increasing real-valued function on [a, b], as before.

Of course,  $\{f_j - f\}_{j=1}^{\infty}$  converges to 0 uniformly on [a, b]. It is easy to see that

(7.9.2) 
$$\lim_{j \to \infty} \int_{a}^{b} (f_j - f) \, d\alpha = 0$$

and

(7.9.3) 
$$\lim_{j \to \infty} \underbrace{\int}_{a}^{b} (f_j - f) \, d\alpha = 0,$$

using the remarks in Susection 6.6.2.

It is not too difficult to show that

(7.9.4) 
$$\lim_{j \to \infty} \overline{\int}_{a}^{b} f_{j} \, d\alpha = \overline{\int}_{a}^{b} f \, d\alpha$$

and

(7.9.5) 
$$\lim_{j \to \infty} \underline{\int}_{a}^{b} f_{j} \, d\alpha = \underline{\int}_{a}^{b} f \, d\alpha,$$

using the remarks in Section 6.12. This also uses (7.9.2), (7.9.3), the facts that

(7.9.6) 
$$f_j = f + (f_j - f)$$

and

(7.9.7) 
$$f = f_j + (f - f_j)$$

for each j. If

(7.9.8)  $f_j$  is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b]

for each j, then it follows that

(7.9.9) f is Riemann–Stieltjes integrable with respect to  $\alpha$  on [a, b],

with

(7.9.10) 
$$\lim_{j \to \infty} \int_a^b f_j \, d\alpha = \int_a^b f \, d\alpha,$$

as in Theorem 7.16 on p151 of [192].

# Bibliography

- V. Abramov, Extension of the Bertrand-De Morgan test and its application, American Mathematical Monthly 127 (2020), 444–448.
- [2] R. Agarwal, C. Flaut, and D. O'Regan, An Introduction to Real Analysis, CRC Press, 2018.
- [3] R. Agnew and R. Boas, Jr., Mathematical notes: An integral test for convergence, American Mathematical Monthly 56 (1949), 677–678.
- [4] S. Ali, The mth ratio test: new convergence tests for series, American Mathematical Monthly 115 (2008), 514–524.
- [5] T. Apostol, *Mathematical Analysis*, 2nd edition, Addison-Wesley, 1974.
- [6] M. Ašić and D. Adamović, Limit points of sequences in metric spaces, American Mathematical Monthly 77 (1970), 613–616.
- [7] R. Ayoub, Euler and the zeta function, American Mathematical Monthly 81 (1974), 1067–1086. See also [110].
- [8] R. Ayoub, What is a Napierian logarithm?, American Mathematical Monthly 100 (1993), 351–364.
- Y. Azouzi, Cluster points of slow sequences, American Mathematical Monthly 125 (2018), 457–461.
- [10] R. Bagley and J. McKnight, Differentiation of infinite series of functions, American Mathematical Monthly 66 (1959), 890–892.
- [11] J. Ballantine, Questions, discussions, and notes: Elementary development of certain infinite series, American Mathematical Monthly 44 (1937), 470– 472.
- [12] E. Barbeau, Mathematical education: A course on the development of analysis, American Mathematical Monthly 83 (1976), 42–43.
- [13] E. Barbeau, Euler subdues a very obstreperous series, American Mathematical Monthly 86 (1979), 356–372.

- [14] R. Beals, Analysis: An Introduction, Cambridge University Press, 2004.
- [15] R. Beals, The scope of the power series method, Mathematics Magazine 86 (2013), 56–62.
- [16] R. Beigel, Rearranging terms in alternating series, Mathematics Magazine 54 (1981), 244–246.
- [17] C. Belna, M. Evans, and P. Humke, Symmetric and strong differentiation, American Mathematical Monthly 86 (1979), 121–123.
- [18] L. Bers, Classroom notes: On avoiding the mean value theorem, American Mathematical Monthly 74 (1967), 583.
- [19] A. Besenyei, Lebesgue's road to antiderivatives, Mathematics Magazine 86 (2013), 255–260.
- [20] Á. Besenyei, Peano's unnoticed proof of Borel's theorem, American Mathematical Monthly 121 (2014), 69–72.
- [21] R. Boas, Jr., Classroom notes: Lhospital's rule without mean-value theorems, American Mathematical Monthly 76 (1969), 1051–1053.
- [22] R. Boas, Jr., Partial sums of infinite series, and how they grow, American Mathematical Monthly 84 (1977), 237–258.
- [23] R. Boas, *Estimating remainders*, Mathematics Magazine **51** (1978), 83–89.
- [24] R. Boas, Jr., The teaching of mathematics: Counterexamples to L'Hopital's rule, American Mathematical Monthly **93** (1986), 644–645.
- [25] R. Boas, Jr., Lion Hunting & Other Mathematical Pursuits: A Collection of Mathematics, Verse, and Stories, Edited and with an introduction by G. Alexanderson and D. Mugler, Mathematical Association of America, 1995.
- [26] R. Boas, Jr., A Primer of Real Functions, 4th edition, revised and updated by H. Boas, Mathematical Association of America, 1996.
- [27] S. Bochner, Fourier series came first, American Mathematical Monthly 86 (1979), 197–199.
- [28] M. Botsko, An easy generalization of the Riemann integral, American Mathematical Monthly 93 (1986), 728–732.
- [29] M. Botsko, An elementary proof that a bounded a.e. continuous function is Riemann integrable, American Mathematical Monthly 95 (1988), 249– 252.
- [30] M. Botsko, A fundamental theorem of calculus that applies to all Riemann integrable functions, Mathematics Magazine 64 (1991), 347–348.

- [31] M. Botsko, *Exactly which bounded Darboux functions are derivatives?*, American Mathematical Monthly **114** (2007), 242–246.
- [32] D. Bressoud, A Radical Approach to Real Analysis, 2nd edition, Mathematical Association of America, 2007.
- [33] D. Bressoud, A Radical Approach to Lebesgue's Theory of Integration, MAA Textbooks, Cambridge University Press, 2008.
- [34] D. Bressoud, *Historical reflections on teaching the fundamental theorem* of integral calculus, American Mthematical Monthly **118** (2011), 99–115.
- [35] D. Bressoud, Calculus Reordered: A History of the Big Ideas, Princeton University Press, 2019.
- [36] R. Brink, A simplified integral test for the convergence of infinite series, American Mathematical Monthly 38 (1931), 205–208.
- [37] A. Browder, Mathematical Analysis: An Introduction, Springer-Verlag, 1996.
- [38] A. Brown, A proof of the Lebesgue condition for Riemann integrability, American Mathematical Monthly **43** (1936), 396–398.
- [39] A. Bruckner, *Derivatives: why they elude classification*, Mathematics Magazine **49** (1976), 5–11.
- [40] A. Bruckner, Differentiation of Real Functions, 2nd edition, American Mathematical Society, 1994.
- [41] A. Bruckner, J. Mařík, and C. Weil, Some aspects of products of derivatives, American Mathematical Monthly 99 (1992), 134–145.
- [42] G. Bullock, The teaching of mathematics: A geometric interpretation of the Riemann-Stieltjes integral, American Mathematical Monthly 95 (1988), 448-455.
- [43] J. Burkill and H. Burkill, A Second Course in Mathematical Analysis, Cambridge University Press, 1970.
- [44] P. Calabrese, Classroom notes: A note on alternating series, American Mahtematical Monthly 69 (1962), 215–217.
- [45] B. Cerimele, Extensions on a theme concerning conditionaly convergent series, Mathematics Magazine 40 (1967), 120–128.
- [46] H. Chand, On some generalizations of Cauchy's condensation and integral tests, American Mathematical Monthly 46 (1939), 338–341.
- [47] L. Cohen, Classroom notes: On being mean to the mean value theorem, American Mathematical Monthly 74 (1967), 581–582.

- [48] G. Convertito and D. Cruz-Uribe, OFS, The Stieltjes Integral, CRC Press, 2023.
- [49] J. Conway, A First Course in Analysis, Cambridge University Press, 2018.
- [50] Y. Coudène, A strange inequality concerning alternating series, American Mathematical Monthly 125 (2018), 554–557.
- [51] C. Cowen, K. Davidson, and R. Kaufman, *Rearranging the alternating harmonic series*, American Mathematical Monthly 87 (1980), 817–819.
- [52] D. Cruz-Uribe, SFO, The relation between the root and ratio tests, Mathematics Magazine 70 (1997), 214–215; News and letters, 310–311.
- [53] J. Dettman, The solution of a second order linear differential equation near a regular singular point, American Mathematical Monthly 71 (1964), 378–385.
- [54] J. Dettman, Power series solutions of ordinary differential equations, American Mathematical Monthly 74 (1967), 428–430.
- [55] A. Devinatz, Advanced Calculus, Holt, Rinehart and Winston, 1968.
- [56] T. Drammatico, A new proof of the equality of mixed second partial derivatives, American Mathematical Monthly 122 (2015), 602–603.
- [57] R. Dressler and K. Stromberg, *The Tonelli integral*, American Mathematical Monthly 81 (1974), 67–68.
- [58] R. Duffin, A generalization of the ratio test for series, American Mathematical Monthly 55 (1948), 153–155.
- [59] W. Dunham, *Touring the calculus gallery*, American Mathematical Monthly **112** (2005), 1–19.
- [60] W. Dunham, *The Calculus Gallery: Masterpieces from Newton to Lebesgue*, Princeton University Press, 2018.
- [61] E. Easdale, J. Fleming, and B. Suceavă, Convergence of series with terms defined by a recurrence relation, American Mathematical Monthly 124 (2017), 360–364.
- [62] M. Evans and P. Humke, Revisiting a century-old characterization of Baire class one, Darboux functions, American Mathematical Monthly 116 (2009), 451-455.
- [63] W. Felscher, Bolzano, Cauchy, epsilon, delta, American Mathematical Monthly 107 (2000), 844–862.
- [64] L. Flatto, Advanced Calculus, Williams & Wilkins, 1976.

- [65] T. Forgács, J. Luong, and J. Williamson, A note on infinite series with recursively defined terms, American Mathematical Monthly 126 (2019), 269–274.
- [66] M. Fort, Jr., Classroom notes: The maximum value of a continuous function, American Mathematical Monthly 58 (1951), 32–33.
- [67] M. Fort, Jr., Mathematical notes: A theorem concerning functions discontinuous on a dense set, American Mathematical Monthly 58 (1951), 408–410.
- [68] M. Fort, Jr., Classroom notes: A fundamental theorem of calculus, American Mathematical Monthly 63 (1956), 334–335.
- [69] P. Franklin, A Treatise on Advanced Calculus, Dover, 1964.
- [70] J. Franks, Cantor's other proofs that R is uncountable, Mathematics Magazine 83 (2010), 283–289.
- [71] F. Freniche, On Riemann's rearrangement theorem for the alternating harmonic series, American Mathematical Monthly 117 (2010), 442–448.
- [72] A. Friedman, Advanced Calculus, Holt, Rinehart and Winston, 1971.
- [73] H. Gaskill and P. Narayanaswami, Foundations of Analysis: The Theory of Limits, Harper & Row, 1989.
- [74] B. Gelbaum and J. Olmstead, *Counterexamples in Analysis*, 2nd edition, Dover, 2003.
- [75] M. Gilula, A class of simple rearrangements of the alternating harmonic series, American Mathematical Monthly 125 (2018), 245–256.
- [76] M. Goar, Olivier and Abel on series convergence: an episode from early 19th century analysis, Mathematics Magazine 72 (1999), 347–355.
- [77] K. Gödel, What is Cantor's continuum problem?, American Mathematical Monthly 54 (1947), 515–525.
- [78] C. Goffman, Definition of the Lebesgue integral, American Mathematical Monthly 60 (1953), 251–252.
- [79] C. Goffman, Classroom notes: Arc length, American Mathematical Monthly 71 (1964), 303–304.
- [80] C. Goffman, Introduction to Real Analysis, Harper & Row, 1966.
- [81] C. Goffman, A bounded derivative which is not Riemann integrable, American Mathematical Monthly 84 (1977), 205–206.
- [82] R. Goldberg, Methods of Real Analysis, 2nd edition, Wiley, 1976.

- [83] E. González-Velasco, Connections in mathematical analysis: the case of Fourier series, American Mathematical Monthly 99 (1992), 427–441.
- [84] E. González-Velasco, James Gregory's calculus in the Geometriæ pars universalis, American Mathematical Monthly 114 (2007), 565–576.
- [85] E. González-Velasco, Journey Through Mathematics: Creative Episodes in its History, Springer, 2011.
- [86] R. Gordon, A bounded derivative that is not Riemann integrable, Mathematics Magazine 89 (2016), 364–370.
- [87] F. Gouvêa, Was Cantor surprised?, American Mathematical Monthly 118 (2011), 198–209.
- [88] F. Gouvêa, A Short Book on Long Sums Infinite Series for Calculus Students, Springer, 2023.
- [89] J. Grabiner, The Origins of Cauchy's Rigorous Calculus, MIT Press, 1981.
- [90] J. Grabiner, Who gave you the epsilon? Cauchy and the origins of rigorous calculus, American Mathematical Monthly 90 (1983), 185–194.
- [91] J. Grabiner, The changing concept of change: the derivative from Fermat to Weierstrass, Mathematics Magazine 56 (1983), 195–206.
- [92] J. Grabiner, Was Newton's calculus a dead end? The continental influence of Maclaurin's treatise of fluxions, American Mathematical Monthly 104 (1997), 393–410.
- [93] J. Grabiner, A Historian Looks Back: The Calculus as Algebra and Selected Writings, Mathematical Association of America, 2010.
- [94] J. Gray, Miscellanea: Clear theory, American Mathematical Monthly 92 (1985), 158–159.
- [95] J. Gray, The Real and the Complex: A History of Analysis in the 19th Century, Springer, 2015.
- [96] J. Green, The theory of functions of a real variable, Mathematics Magazine 24 (1951), 209–217.
- [97] R. Grinberg, The Real Analysis Lifesaver: All the Tools that You Need to Understand Proofs, Princeton University Press, 2017.
- [98] R. Gunning, An Introduction to Analysis, Princeton University Press, 2018.
- [99] L. Gurin, A problem, American Mathematical Monthly 103 (1996), 683– 686.

- [100] C. Hammond, The case for Raabe's test, Mathematics Magazine 93 (2020), 36–46.
- [101] C. Hammond and E. Omey, A second look at the second ratio test, American Mathematical Monthly 128 (2021), 579–596.
- [102] R. Hansen, Series representation of elements in separable Banach spaces, Mathematics Magazine 45 (1972), 190–194.
- [103] D. Harris and B. Thorp, Uniform convergence and rearrangement, American Mathematical Monthly 76 (1969), 801–802.
- [104] H. Herrlich, Axiom of Choice, Lecture Notes in Mathematics 1876, Springer-Verlag, 2006.
- [105] I. Herron, The radius of convergence of power series solutions to linear differential equations, American Mathematical Monthly 96 (1989), 824– 827.
- [106] T. Hildebrandt, Definitions of Stieltjes integrals of the Riemann type, American Mathematical Monthly 45 (1938), 265–278.
- [107] T. Hildebrandt, Remarks on the Abel-Dini theorem, American Mathematical Monthly 49 (1942), 441–445.
- [108] E. Hille, Mathematics and mathematicians from Abel to Zermelo, Mathematics Magazine 26 (1953), 127–146.
- [109] K. Hoffman, Analysis in Euclidean Space, Prentice-Hall, 1975.
- [110] A. Howson, Addendum to: "Euler and the zeta function" by Raymond Ayoub" [7], American Mathematical Monthly 82 (1975), 737.
- [111] E. Huynh, A second Raabe's test and other series tests, American Mathematical Monthly 129 (2022), 865–875.
- [112] P. Johnson and R. Redheffer, Scrambled series, American Mathematical Monthly 73 (1966), 822–828.
- [113] R. Johnsonbaugh, Another proof of an estimate for e, American Mathematical Monthly 81 (1974), 1011–1012.
- [114] R. Johnsonbaugh, Summing an alternating series, American Mathematical Monthly 86 (1979), 637–648.
- [115] R. Johnsonbaugh, Mathematical notes: The alternating series defined by an increasing function, American Mathematical Monthly 87 (1980), 36– 38.
- [116] F. Jones, Lebesgue Integration on Euclidean Space, Jones and Bartlett, 1993.

- [117] R. Jones, Achievement sets of sequences, American Mathematical Monthly 118 (2011), 508–512.
- [118] H. Junghenn, A Course in Real Analysis, CRC Press, 2015.
- [119] W. Kaczor and M. Nowak, Problems in Mathematical Analysis I: Real Numbers, Sequences and Series, translated and revised from the 1996 Polish original by the authors, American Mathematical Society, 2000.
- [120] W. Kaczor and M. Nowak, Problems in Mathematical Analysis II: Continuity and Differentiation, translated from the 1998 Polish original, revised, and augmented by the authors, American Mathematical Society, 2001.
- [121] W. Kaczor and M. Nowak, Problems in Mathematical Analysis III: Integration, American Mathematical Society, 2003.
- [122] I. Kaplansky, Set Theory and Metric Spaces, 2nd edition, Chelsea, 1977.
- [123] H. Kaptanoğlu, In praise of  $y = x^{\alpha} \sin(\frac{1}{x})$ , American Mathematical Monthly **108** (2001), 144–150.
- [124] G. Klambauer, Mathematical Analysis, Dekker, 1975.
- [125] M. Kline, Euler and infinite series, Mathematics Magazine 56 (1983), 307–314.
- [126] A. Knapp, *Basic Real Analysis*, Birkhäuser, 2005.
- [127] J. Koliha, Mean, meaner, and the meanest mean value theorem, American Mathematical Monthly 116 (2009), 356–361.
- [128] D. Kopel and M. Schramm, A new extension of the derivative, American Mathematical Monthly 97 (1990), 230–233.
- [129] T. Körner, Differentiable functions on the rationals, Bulletin of the London Mathematical Society 23 (1991), 557–562.
- [130] T. Körner, Butterfly hunting and the symmetry of mixed partial derivativs, American Mathematical Monthly **105** (1998), 756–758.
- [131] T. Körner, A Companion to Analysis: A Second First and First Second Course in Analysis, American Mathematical Society, 2004.
- [132] S. Krantz, You don't need to be a weatherman to know which way the wind blows, American Mathematical Monthly 106 (1999), 915–918.
- [133] S. Krantz, A Guide to Real Variables, Mathematical Association of America, 2009.
- [134] S. Krantz, Foundations of Analysis, CRC Press, 2015.

- [135] S. Krantz, Elementary Introduction to the Lebesgue Integral, CRC Press, 2018.
- [136] S. Krantz, Real Analysis and Foundations, 5th edition, CRC Press, 2022.
- [137] S. Krantz, The Elements of Advanced Mathematics, 5th edition, CRC Press, 2022.
- [138] S. Krantz and J. McNeal, Creating more convergent series, American Mathematical Monthly 111 (2004), 32–38.
- [139] A. Krupowies and F. Prus-Wiśniowski, The character of convergence of the Cauchy product, American Mathematical Monthly 130 (2023), 671– 677.
- [140] M. Laczkovich and V. Sós, Real Analysis Foundations and Functions of One Variable, translated from the third (2005) Hungarian edition by the authors, Springer, 2015.
- [141] S. Lang, Undergraduate Analysis, 2nd edition, Springer-Verlag, 1997.
- [142] L. Lange, On two famous inequlities, Mathematics Magazine 32 (1959), 157–160.
- [143] L. Lange and D. Thoro, Classroom notes: The density of Pythagorean rationals, American Mathematical Monthly 71 (1964), 664–665.
- [144] R. Langer, Fourier's Series: The Genesis and Evolution of a Theory, The first Herbert Ellsworth Slaught Memorial Paper, American Mathematical Monthly 54 (1947), no. 7, part II.
- [145] W. Langlois and L. Holder, The relation of  $f'_+(a)$  to f'(a+), Mathematics Magazine **39** (1966), 112–120.
- [146] J. LaVita, Classroom notes: A necessary and sufficient condition for Riemann integration, American Mathematical Monthly 71 (1964), 193–196.
- [147] P. Lax, Rethinking the Lebesgue integral, American Mathematical Monthly 116 (2009), 863–881.
- [148] T. Lehrer, The Derivative Song, American Mathematical Monthly 81 (1974), 490.
- [149] T. Lehrer, There's a Delta for Every Epsilon (Calypso), American Mathematical Monthly 81 (1974), 612.
- [150] T. Lehrer, The Professor's Song, American Mathematical Monthly 81 (1974), 745.
- [151] R. Leipnik, Note on alternating Tannery series, American Mathematical Monthly 65 (1958), 197–198.

140

- [152] N. Lennes, Questions, discussions, and notes: The ratio test for convergence of series, American Mathematical Monthly 46 (1939), 434–436.
- [153] P. Leonetti, Convergence rates of subseries, American Mathematical Monthly 126 (2019), 163–167.
- [154] P. Leonetti, A characterization of Cesàro convergence, American Mathematical Monthly 128 (2021), 559–562.
- [155] L. Levine, On a necessary and sufficient condition for Riemann integrability, American Mathematical Monthly 84 (1977), 205.
- [156] S. Malik and S. Arora, *Mathematical Analysis*, 2nd edition, Wiley, 1992.
- [157] J. Mathews, A note on well-chained spaces, American Mathematical Monthly 75 (1968), 273–275.
- [158] R. Metzler, On Riemann integrability, American Mathematical Monthly 78 (1971), 1129–1131.
- [159] M. Meyerson, Every power series is a Taylor series, American Mathematical Monthly 88 (1981), 51–52.
- [160] M. Meyerson, The  $x^x$  spindle, Mathematics Magazine **69** (1996), 198–206.
- [161] M. Mezzino and M. Pinsky, Leibniz's formula, Cauchy majorants, and linear differential equations, Mathematics Magazine 71 (1998), 360–368.
- [162] H. Miller, A ratio-type convergence test, American Mathematical Monthly 77 (1970), 285–287.
- [163] N. Miller, Infinite series formed by associating the terms of a series of functions with the points of a sequence, American Mathematical Monthly 37 (1930), 224–227.
- [164] P. Mohazzabi and T. Fournelle, Evaluation of ill-behaved power series, American Mathematical Monthly 111 (2004), 308–321.
- [165] G. Moore, Zermelo's Axiom of Choice: Its Origins, Development, and Influence, Springer-Verlag, 1982.
- [166] F. Morgan, *Real Analysis*, American Mathematical Society, 2005.
- [167] F. Morgan, Real Analysis and Applications: Including Fourier Series and the Calculus of Variations, American Mathematical Society, 2005.
- [168] S. Nadler, Jr., A Proof of Darboux's theorem, American Mathematical Monthly 117 (2010), 174–175.
- [169] Z. Nitecki, Cantorvals and subsum sets of null sequences, American Mathematical Monthly 122 (2015), 862–870.

- [170] M. Norris, Integrability of continuous functions, American Mathematical Monthly 59 (1952), 244–245.
- [171] M. Norris, Some necessary conditions for convergence of infinite series and improper integrals, American Mathematical Monthly 60 (1953), 96–97.
- [172] J. Nurcombe, A sequence of convergence tests, American Mathematical Monthly 86 (1979), 679–681.
- [173] L. Olsen, A new proof of Darboux's theorem, American Mathematical Monthly 111 (2004), 713–715.
- [174] S. Ponnusamy, Foundations of Mathematical Analysis, Birkhäuser / Springer, 2012.
- [175] G. Porter, Classroom notes: An alternative to the integral test for infinite series, American Mathematical Monthly 79 (1972), 634–635.
- [176] M. Protter, Basic Elements of Real Analysis, Springer-Verlag, 1998.
- [177] M. Protter and C. Morrey, Jr., A First Course in Analysis, 2nd edition, Springer-Verlag, 1991.
- [178] F. Prus-Wiśniowski, A refinement of Raabe's test, American Mathematical Monthly 115 (2008), 249–252.
- [179] C. Pugh, Real Mathematical Analysis, 2nd edition, Springer, 2015.
- [180] H. Rademacher, On the condition of Riemann integrability, American Mathematical Monthly 61 (1954), 1–8.
- [181] C. Rajagopal, Remarks on some generalizations of Cauchy's condensation and integral tests, American Mathematical Monthly 48 (1941), 180–185.
- [182] C. Rajagopal, The Abel–Dini and allied theorems, American Mathematical Monthly 51 (1944), 566–570.
- [183] R. Redheffer, Some thoughts about limits, Mathematics Magazine 62 (1989), 176–184.
- [184] M. Reed, Fundamental Ideas of Analysis, Wiley, 1998.
- [185] M. Rosenlicht, Introduction to Analysis, Dover, 1986.
- [186] Daniel Rosenthal, David Dosenthal, and Peter Rosenthal, A Readable Introduction to Real Mathematics, 2nd edition, Springer, 2018.
- [187] P. Rosenthal, The remarkable theorem of Lévy and Steinitz, American Mathematical Monthly 94 (1987), 342–351.
- [188] C. Rosentrater, Varieties of Integration, Mathematical Association of America, 2015.

- [189] K. Ross, Another approach to Riemann-Stieltjes integrals, American Mathematical Monthly 87 (1980), 660–662.
- [190] K. Ross, Elementary Analysis: The Theory of Calculus, 2nd edition, in collaboration with J. López, Springer, 2013.
- [191] W. Rudin, Real and Complex Analysis, 2nd edition, McGraw-Hill, 1974. A third edition edition was published in 1987.
- [192] W. Rudin, Principles of Mathematical Analysis, 3rd edition, McGraw-Hill, 1976.
- [193] H. Samelson, More on Kummer's test, American Mathematical Monthly 102 (1995), 817–818.
- [194] U. Satyanarayana, A note on Riemann-Stieltjes integrals, American Mathematical Monthly 87 (1980), 477–478.
- [195] P. Schaefer, The density of certain classes of rationals, American Mathematical Monthly 72 (1965), 894–895.
- [196] P. Schaefer, *Limit points of bounded sequences*, American Mathematical Monthly **75** (1968), 51.
- [197] P. Schaefer, Sum-preserving rearrangements of infinite series, American Mathematical Monthly 88 (1981), 33–40.
- [198] D. Schattschneider, A multiplicative metric, Mathematics Magazine 49 (1976), 203–2-5.
- [199] M. Schramm, J. Troutman, and D. Waterman, Segmentally alternating series, American Mathematical Monthly 121 (2014), 717–722.
- [200] R. Shakarchi, Problems and Solutions for Undergraduate Analysis ([141]), Springer-Verlag, 1998.
- [201] O. Shisha, An approach to Darboux-Stieltjes integration, American Mathematical Monthly 72 (1965), 890–892.
- [202] O. Stanaitis, Integral tests for infinite series, American Mathematical Monthly 78 (1971), 164–170.
- [203] J. Stefánsson, Forward shifts and backward shifts in a rearrangement of a conditionally convergent series, American Mathematical Monthly 111 (2004), 913–914.
- [204] R. Strichartz, The Way of Analysis, Revised edition, Jones and Bartlett, 2000.
- [205] K. Stromberg, An Introduction to Classical Real Analysis, AMS Chelsea, 1981.

- [206] A. Taylor, Derivatives in the calculus, American Mathematical Monthly 49 (1942), 631–642.
- [207] A. Taylor, L'Hospital's rule, American Mathematical Monthly 59 (1952), 20–24.
- [208] A. Taylor and W. Mann, Advanced Calculus, 3rd edition, Wiley, 1983.
- [209] H. ter Horst, Riemann-Stieltjes and Lebesgue-Stieltjes integrability, American Mathematical Monthly 91 (1984), 551–559.
- [210] T. Tokieda, A mean value theorem, American Mathematical Monthly 106 (1999), 673–674.
- [211] J. Tong, Kummer's test gives characterizations for convergence or divergence of all positive series, American Mathematical Monthly 101 (1994), 450–452.
- [212] M. Villarino, The error in an alternating series, American Mathematical Monthly 125 (2018), 360–364.
- [213] D. Widder, Advanced Calculus, 3rd edition, Dover, 1989.
- [214] B. Youse, Introduction to Real Analysis, Allyn and Bacon, 1972.

## Index

(a, b), 6 [a, b], 7

absolute convergence, 51 absolute value, 1, 38 alternating series test, 56 archimedean property, 4

B(x,r), 5 Baire category theorem, 90 Baire class one, 98 bases, 29 bounded functions, 80 bounded sequences, 40 bounded sets, 12 Brouwer's fixed-point theorem, 90

**C**, 38 C(M, N), 80 $C_b(M, N), 80$ Cantor set, 36 Cartesian products, 20 Cauchy products, 66 Cauchy sequences, 45 Cauchy's condensation test, 54 cells, 20 chain rule, 93 closed intervals, 7 closed sets, 7 closures, 9 compact sets, 10 comparison test, 51 complements, 7 complete metric spaces, 47 completions, 89 complex conjugate, 38 compositions of mappings, 72

connected sets, 35 continuity at a point, 69 continuous mappings, 70 contraction mapping theorem, 90 contractions, 90 convergent sequences, 39 convergent series, 49 convex sets, 88 countably infinite sets, 25

Darboux sums, 104 Darboux sums, 104 Darboux–Stieltjes sums, 104 dense sets, 28 derivative, 91 from the left, 95 from the right, 95 diagonalization argument, 27 diameter, 20 differentiable functions, 91 discrete metric, 2

Euclidean metric, 1 Euclidean metrics, 2 extended real numbers, 57 extreme value theorem, 74

fundamental theorem of calculus, 123

generalized mean value theorem, 97 geometric series, 52 greatest lower bounds, 3, 57

imaginary part, 38 infimum, 3, 57 intermediate value theorem, 76 intervals, 6, 7 inverse images, 71

## INDEX

isometries, 89

least upper bound property, 3 least upper bounds, 3, 57 Lebesgue integrals, 89 limit points, 6 limit inferior, 60 limit of a function, 82 limit point property, 11 limit superior, 60 limits of sequences, 40 Lindelöf's theorem, 31 Lipschitz mappings, 99 lower bounds, 3, 57 lower limits, 60 lower sums, 104

mappings, 69 metric spaces, 1 metrics, 1 modulus, 38 monotone sequences, 40, 41 monotonically increasing functions, 84

one-sided limits, 84 open balls, 5 open coverings, 10 open intervals, 6 open sets, 5

partitions of intervals, 103 common refinements, 108 refinements, 106 path connected sets, 88 perfect sets, 28 pointwise convergence, 77 power series, 52 product rule, 92

 $\mathbf{Q}, 3, 27$ quotient rule, 93

## **R**, 1

 $\mathbf{R}^{n}$ , 2 radius of convergence, 53 ratio test, 64 real part, 38

rearrangements, 65 relatively open sets, 15 Riemann integrability, 109 Riemann integrals, 106, 109 Riemann sums, 104 Riemann-Stieltjes integrability, 110 Riemann-Stieltjes integrals, 107, 110 Riemann-Stieltjes sums, 105 root test, 62 separable metric spaces, 28 separated sets, 34 sequential compactness, 44 standard metric on  $\mathbf{C}$ , 39 subsequences, 44 supremum, 3, 57 supremum metric, 80 totally bounded sets, 23 triangle inequality, 1 uncountable sets, 27

uniform continuity, 74 uniform convergence, 77 unit step function, 86 upper bounds, 3, 57 upper limits, 60 upper sums, 104

Weierstrass' criterion, 78 Weierstrass' theorem, 90

 $\mathbf{Z}, 25 \\
 \mathbf{Z}_{+}, 10$ 

## 146